COACTIONS OF HOPF C*-BIMODULES

THOMAS TIMMERMANN

Communicated by Şerban Strătilă

ABSTRACT. Coactions of Hopf C^* -bimodules simultaneously generalize coactions of Hopf C^* -algebras and actions of groupoids. Following an approach of Baaj and Skandalis, we construct reduced crossed products and establish a duality for fine coactions. Examples of coactions arise from Fell bundles on groupoids and actions of a groupoid on bundles of C^* -algebras. Continuous Fell bundles on an étale groupoid correspond to coactions of the reduced groupoid algebra, and actions of a groupoid on a continuous bundle of C^* -algebras correspond to coactions of the function algebra.

KEYWORDS: Quantum groupoid, groupoid, duality, coaction, crossed product.

MSC (2000): Primary 46L55; Secondary 22A22, 20G42.

1. INTRODUCTION AND PRELIMINARIES

Actions of quantum groupoids that simultaneously generalize actions of quantum groups and actions of groupoids have been studied in various settings, including that of weak Hopf algebras or finite quantum groupoids [24], [25], Hopf algebroids or algebraic quantum groupoids [7], [13], and Hopf–von Neumann bimodules or measured quantum groupoids [10], [11], [30]. In this article, we introduce and investigate coactions of Hopf *C**-bimodules or reduced locally compact quantum groupoids within the framework developed in [28], [27].

In the first part of this article, we construct reduced crossed products and dual coactions, and show that the bidual of a fine coaction is Morita equivalent to the initial coaction. These constructions apply to pairs of Hopf C^* -bimodules that appear as the left and the right leg of a (weak) C^* -pseudo-Kac system, which consists of a C^* -pseudo-multiplicative unitary [27] and an additional symmetry. We associate such a C^* -pseudo-Kac system to every groupoid and to every compact C^* -quantum groupoid and expect that the same can be done for every reduced locally compact quantum groupoid once this concept has been defined properly. The constructions in this part generalize corresponding constructions of Baaj and Skandalis [3] for coactions of Hopf C^* -algebras.

Coactions of the Hopf C^* -bimodules associated to a locally compact Hausdorff groupoid — the function algebra on one side and the reduced groupoid algebra on the other — are studied in detail in the second part of this article. We show that actions of the groupoid on continuous bundles of C^* -algebras correspond to coactions of the first Hopf C^* -bimodule, and that continuous Fell bundles on *G* naturally yield coactions of the second Hopf C^* -bimodule. Generalizing results of Quigg [22] and Baaj and Skandalis [2] from groups to groupoids, we show that if the groupoid is étale, every coaction of the reduced groupoid algebra arises from a Fell bundle.

This article is organized as follows. The first part is concerned with coactions of Hopf C^* -bimodules and associated reduced crossed products.

Section 2 summarizes the relative tensor product of C^* -modules and the fiber product of C^* -algebras over C^* -bases [28] which are fundamental to everything that follows, and introduces coactions of Hopf C^* -bimodules.

Section 3 is concerned with C^* -pseudo-Kac systems. Every C^* -pseudo-Kac system gives rise to two Hopf C^* -bimodules, called the legs of the system, which are dual to each other in a suitable sense. Coactions of these legs on C^* -algebras, associated reduced crossed products, dual coactions and a duality theorem concerning iterated crossed products are discussed in Section 4.

Section 5 gives the construction of the C^* -pseudo-Kac system of a locally compact Hausdorff groupoid *G*. The associated Hopf C^* -bimodules are the function algebra on one side and the reduced groupoid C^* -algebra of *G* on the other side. The second part of the article relates coactions of these Hopf C^* -bimodules to well-known notions.

Section 6 shows that actions of a groupoid *G* on continuous bundles of C^* -algebras correspond to certain fine coactions of the function algebra of *G*.

Section 7 contains preliminaries on Fell bundles, their morphisms and multipliers.

Section 8 shows that continuous Fell bundles on G give rise to coactions of the reduced groupoid C^* -algebra of G, and Section 9 gives a reverse construction that associates to every sufficiently nice coaction of the groupoid algebra a Fell bundle provided that the groupoid G is étale.

We use the following notation. Given a subset *Y* of a normed space *X*, we denote by $[Y] \subset X$ the closed linear span of *Y*. All sesquilinear maps like inner products of Hilbert spaces are assumed to be conjugate-linear in the first component and linear in the second one. Given a Hilbert space *H*, we use the ketbra notation and define for each $\xi \in H$ operators $|\xi\rangle : \mathbb{C} \to H, \lambda \mapsto \lambda \xi$, and $\langle \xi | = |\xi\rangle^* : H \to \mathbb{C}, \xi' \mapsto \langle \xi | \xi' \rangle$. Given a C*-algebra *A* and a subspace $B \subset A$, we denote by $A \cap B'$ the relative commutant $\{a \in A : [a, B] = 0\}$.

We shall make extensive use of (right) Hilbert C^* -modules; see [16]. In particular, we use the internal tensor product and the KSGNS-construction. Let *E* be a Hilbert C^* -module over a C^* -algebra *A*, let *F* be a Hilbert C^* -module over

a C*-algebra B, and let $\phi : A \to \mathcal{L}(F)$ be a completely positive map. We denote by $E \otimes_{\phi} F$ the Hilbert *C*^{*}-module over *B* which is the closed linear span of elements $\eta \otimes_{\phi} \xi$, where $\eta \in E$ and $\xi \in F$ are arbitrary, and $\langle \eta \otimes_{\phi} \xi | \eta' \otimes_{\phi} \xi' \rangle =$ $\langle \xi | \phi(\langle \eta | \eta' \rangle) \xi' \rangle$ and $(\eta \otimes_{\phi} \xi) b = \eta \otimes_{\phi} \xi b$ for all $\eta, \eta' \in E, \xi, \xi' \in F$, and $b \in B$. If ϕ is a *-homomorphism, this is the usual internal tensor product; if F = B, this is the KSGNS-construction. If $S \in \mathcal{L}(E)$ and $T \in \mathcal{L}(F) \cap \phi(A)'$, then there exists a unique operator $S \otimes_{\phi} T \in \mathcal{L}(E \otimes_{\phi} E)$ such that $(S \otimes_{\phi} T)(\eta \otimes_{\phi} \xi) = S\eta \otimes_{\phi} T\xi$ for all $\eta \in E$, $\xi \in F$; see Proposition 1.34 in [9]. We sloppily write " \otimes_A " or " \odot " instead of " \odot_{ϕ} " if no confusion may arise. We also define a flipped product $F_{\phi} \otimes E$ as follows. We equip the algebraic tensor product $F \odot E$ with the structure maps $\langle \xi \odot \eta | \xi' \odot \eta' \rangle := \langle \xi | \phi(\langle \eta | \eta' \rangle) \xi' \rangle$, $(\xi \odot \eta) b := \xi b \odot \eta$, form the separated completion, and obtain a Hilbert C^* -module $F_{\phi} \otimes E$ over B which is the closed linear span of elements $\xi_{\phi} \otimes \eta$, where $\eta \in E$ and $\xi \in F$ are arbitrary, and $\langle \xi_{\phi} \otimes \eta | \xi'_{\phi} \otimes \eta' \rangle = \langle \xi | \phi(\langle \eta | \eta' \rangle) \xi' \rangle$ and $(\xi_{\phi} \otimes \eta) b = \xi b_{\phi} \otimes \eta$ for all $\eta, \eta' \in E$, $\xi,\xi' \in F$, and $b \in B$. Again, we sloppily write " $_A \otimes$ " or " \otimes " instead of " $_{\phi} \otimes$ " if no confusion may arise. Evidently, there exists a unitary $\Sigma : F \otimes E \xrightarrow{\cong} E \otimes F$, $\eta \otimes \xi \mapsto \xi \otimes \eta.$

2. HOPF C*-BIMODULES AND COACTIONS

A groupoid differs from a group in the non-triviality of its unit space. In almost every approach to quantum groupoids, the unit space is replaced by a nontrivial algebra, and a relative tensor product of modules and a fiber product of algebras over that algebra become fundamentally important. We shall use the corresponding constructions for C^* -algebras introduced in [28] and briefly summarize the main definitions and results below. For additional details and motivation, see [28], [27].

2.1. THE RELATIVE TENSOR PRODUCT. A C^* -base is a triple $(\mathfrak{K}, \mathfrak{B}, \mathfrak{B}^{\dagger})$ consisting of a Hilbert space \mathfrak{K} and two commuting nondegenerate C^* -algebras $\mathfrak{B}, \mathfrak{B}^{\dagger} \subseteq \mathcal{L}(\mathfrak{K})$. It should be thought of as a C^* -algebraic counterpart to pairs consisting of a von Neumann algebra and its commutant. Let $\mathfrak{b} = (\mathfrak{K}, \mathfrak{B}, \mathfrak{B}^{\dagger})$ be a C^* -base. Its *opposite* is the C^* -base $\mathfrak{b}^{\dagger} := (\mathfrak{K}, \mathfrak{B}^{\dagger}, \mathfrak{B})$.

A C^* -b-module is a pair $H_{\alpha} = (H, \alpha)$, where H is a Hilbert space and $\alpha \subseteq \mathcal{L}(\mathfrak{K}, H)$ is a closed subspace satisfying $[\alpha \mathfrak{K}] = H$, $[\alpha \mathfrak{B}] = \alpha$, and $[\alpha^* \alpha] = \mathfrak{B} \subseteq \mathcal{L}(\mathfrak{K})$. If H_{α} is a C^* -b-module, then α is a Hilbert C^* -module over B with inner product $(\xi, \xi') \mapsto \xi^* \xi'$ and there exist isomorphisms

$$(2.1) \qquad \qquad \alpha \otimes \mathfrak{K} \to H, \quad \xi \otimes \zeta \mapsto \xi \zeta; \quad \mathfrak{K} \otimes \alpha \to H, \quad \zeta \otimes \xi \mapsto \xi \zeta;$$

and a nondegenerate representation

$$\rho_{\alpha}: \mathfrak{B}^{\dagger} \to \mathcal{L}(H), \quad \rho_{\alpha}(b^{\dagger})(\xi\zeta) = \xi b^{\dagger}\zeta \quad \text{for all } b^{\dagger} \in \mathfrak{B}^{\dagger}, \xi \in \alpha, \zeta \in \mathfrak{K}.$$

A *semi-morphism* between C^* -b-modules H_{α} and K_{β} is an operator $T \in \mathcal{L}(H, K)$ satisfying $T\alpha \subseteq \beta$. If additionally $T^*\beta \subseteq \alpha$, we call T a *morphism*. We denote the set of all (semi-)morphisms by $\mathcal{L}_{(s)}(H_{\alpha}, K_{\beta})$. If $T \in \mathcal{L}_s(H_{\alpha}, K_{\beta})$, then $T\rho_{\alpha}(b^{\dagger}) = \rho_{\beta}(b^{\dagger})T$ for all $b^{\dagger} \in \mathfrak{B}^{\dagger}$, and if additionally $T \in \mathcal{L}(H_{\alpha}, K_{\beta})$, then left multiplication by T defines an operator in $\mathcal{L}(\alpha, \beta)$ which we again denote by T.

We shall use the following notion of C^* -bi- and C^* -*n*-modules. Let $\mathfrak{b}_1, \ldots, \mathfrak{b}_n$ be C^* -bases, where $\mathfrak{b}_i = (\mathfrak{K}_i, \mathfrak{B}_i, \mathfrak{B}_i^{\dagger})$ for each *i*. A C^* - $(\mathfrak{b}_1, \ldots, \mathfrak{b}_n)$ -module is a tuple $(H, \alpha_1, \ldots, \alpha_n)$, where *H* is a Hilbert space and (H, α_i) is a C^* - \mathfrak{b}_i -module for each *i* such that $[\rho_{\alpha_i}(\mathfrak{B}_i^{\dagger})\alpha_j] = \alpha_j$ whenever $i \neq j$. In the case n = 2, we abbreviate ${}_{\alpha}H_{\beta} := (H, \alpha, \beta)$. If $(H, \alpha_1, \ldots, \alpha_n)$ is a C^* - $(\mathfrak{b}_1, \ldots, \mathfrak{b}_n)$ -module, then $[\rho_{\alpha_i}(\mathfrak{B}_i^{\dagger}), \rho_{\alpha_j}(\mathfrak{B}_j^{\dagger})] = 0$ whenever $i \neq j$. The set of (*semi-)morphisms* between C^* - $(\mathfrak{b}_1, \ldots, \mathfrak{b}_n)$ -modules $\mathcal{H} = (H, \alpha_1, \ldots, \alpha_n)$ and $\mathcal{K} = (K, \beta_1, \ldots, \beta_n)$ is $\mathcal{L}_{(s)}(\mathcal{H}, \mathcal{K}) :=$ $\bigcap_{i=1}^n \mathcal{L}_{(s)}(H_{\alpha_i}, K_{\beta_i}) \subseteq \mathcal{L}(H, K)$.

Let $\mathfrak{b} = (\mathfrak{K}, \mathfrak{B}, \mathfrak{B}^{\dagger})$ be a C^* -base, H_{β} a C^* - \mathfrak{b} -module, and K_{γ} a C^* - \mathfrak{b}^{\dagger} -module. The *relative tensor product* of H_{β} and K_{γ} is the Hilbert space

$$H_{\beta \otimes_{\mathfrak{h}} \gamma} K := \beta \otimes \mathfrak{K} \otimes \gamma.$$

It is spanned by elements $\xi \otimes \zeta \otimes \eta$, where $\xi \in \beta$, $\zeta \in \mathfrak{K}$, $\eta \in \gamma$, and the inner product is given by $\langle \xi \otimes \zeta \otimes \eta | \xi' \otimes \zeta' \otimes \eta' \rangle = \langle \zeta | \xi^* \xi' \eta^* \eta' \zeta' \rangle = \langle \zeta | \eta^* \eta' \xi^* \xi' \zeta' \rangle$ for all $\xi, \xi' \in \beta, \zeta, \zeta' \in \mathfrak{K}, \eta, \eta' \in \gamma$. Obviously, there exists a unitary *flip*

$$\Sigma: H_{\beta \underset{\mathfrak{b}}{\otimes} \gamma} K \to K_{\gamma \underset{\mathfrak{b}^{\dagger}}{\otimes} \beta} H, \quad \xi \odot \zeta \odot \eta \mapsto \eta \odot \zeta \odot \xi.$$

Using the unitaries in (2.1) on H_{β} and K_{γ} , respectively, we shall make the following identifications without further notice:

$$H_{\rho_{\beta}} \otimes \gamma \cong H_{\beta} \underset{\mathfrak{b}}{\otimes}_{\gamma} K \cong \beta \otimes_{\rho_{\gamma}} K, \quad \xi \zeta \otimes \eta \equiv \xi \otimes \zeta \otimes \eta \equiv \xi \otimes \eta \zeta.$$

For all $S \in \rho_{\beta}(\mathfrak{B}^{\dagger})'$ and $T \in \rho_{\gamma}(\mathfrak{B})'$, we have operators

$$S \otimes \mathrm{id} \in \mathcal{L}(H_{\rho_{\beta}} \otimes \gamma) = \mathcal{L}(H_{\beta} \underset{\mathfrak{b}}{\otimes} \gamma K), \quad \mathrm{id} \otimes T \in \mathcal{L}(\beta \otimes_{\rho_{\gamma}} K) = \mathcal{L}(H_{\beta} \underset{\mathfrak{b}}{\otimes} \gamma K).$$

If $S \in \mathcal{L}_s(H_\beta)$ or $T \in \mathcal{L}_s(K_\gamma)$, then $(S \otimes id)(\xi \otimes \eta \zeta) = S\xi \otimes \eta \zeta$ or $(id \otimes T)(\xi \zeta \otimes \eta) = \xi \zeta \otimes T\eta$, respectively, for all $\xi \in \beta, \zeta \in \mathfrak{K}, \eta \in \gamma$, so that we can define

$$\begin{split} S &\underset{\mathfrak{b}}{\otimes} T := (S \otimes \mathrm{id})(\mathrm{id} \otimes T) = (\mathrm{id} \otimes T)(S \otimes \mathrm{id}) \in \mathcal{L}(H_{\beta} \underset{\mathfrak{b}}{\otimes}_{\gamma} K) \\ &\text{for all } (S,T) \in (\mathcal{L}_{s}(H_{\beta}) \times \rho_{\gamma}(\mathfrak{B})') \cup (\rho_{\beta}(\mathfrak{B}^{\dagger})' \times \mathcal{L}_{s}(K_{\gamma})). \end{split}$$

For each $\xi \in \beta$ and $\eta \in \gamma$, there exist bounded linear operators

$$|\xi\rangle_1: K \to H_{\beta \underset{\mathfrak{b}}{\otimes} \gamma} K, \ \omega \mapsto \xi \otimes \omega, \quad |\eta\rangle_2: H \to H_{\beta \underset{\mathfrak{b}}{\otimes} \gamma} K, \ \omega \mapsto \omega \otimes \eta,$$

whose adjoints $\langle \xi |_1 := |\xi \rangle_1^*$ and $\langle \eta |_2 := |\eta \rangle_2^*$ are given by

$$\langle \xi |_1 : \xi' \otimes \omega \mapsto \rho_{\gamma}(\xi^* \xi') \omega, \quad \langle \eta |_2 : \omega \otimes \eta' \mapsto \rho_{\beta}(\eta^* \eta') \omega,$$

We write $|\beta\rangle_1 := \{|\xi\rangle_1 : \xi \in \beta\} \subseteq \mathcal{L}(K, H_{\beta \underset{\mathfrak{b}}{\otimes} \gamma}K)$ and similarly define $\langle \beta|_1, |\gamma\rangle_2$, and $\langle \gamma|_2$.

Let $\mathcal{H} = (\mathcal{H}, \alpha_1, ..., \alpha_m, \beta)$ be a C^* - $(\mathfrak{a}_1, ..., \mathfrak{a}_m, \mathfrak{b})$ -module and $\mathcal{K} = (\mathcal{K}, \gamma, \delta_1, ..., \delta_n)$ a C^* - $(\mathfrak{b}^{\dagger}, \mathfrak{c}_1, ..., \mathfrak{c}_n)$ -module, where $\mathfrak{a}_i = (\mathfrak{H}_i, \mathfrak{A}_i, \mathfrak{A}_i^{\dagger})$ and $\mathfrak{c}_j = (\mathfrak{L}_j, \mathfrak{C}_j, \mathfrak{C}_j^{\dagger})$ are C^* -bases for all i, j. We define

$$\alpha_i \triangleleft \gamma := [|\gamma\rangle_2 \alpha_i] \subseteq \mathcal{L}(\mathfrak{H}_i, H_{\beta \underset{\mathfrak{h}}{\otimes} \gamma} K), \quad \beta \triangleright \delta_j := [|\beta\rangle_1 \delta_j] \subseteq \mathcal{L}(\mathfrak{L}_j, H_{\beta \underset{\mathfrak{h}}{\otimes} \gamma} K)$$

for all *i*, *j*. Then $(H_{\beta \otimes_{\mathfrak{b}} \gamma} K, \alpha_1 \triangleleft \gamma, \ldots, \alpha_m \triangleleft \gamma, \beta \triangleright \delta_1, \ldots, \beta \triangleright \delta_n)$ is a C^* - $(\mathfrak{a}_1, \ldots, \mathfrak{a}_m, \mathfrak{c}_1, \ldots, \mathfrak{c}_n)$ -module, called the *relative tensor product* of \mathcal{H} and \mathcal{K} and denoted by $\mathcal{H} \bigotimes_{\mathfrak{b}} \mathcal{K}$. For all *i*, *j* and $a^{\dagger} \in \mathfrak{A}_i^{\dagger}, c^{\dagger} \in \mathfrak{C}_j^{\dagger}$,

$$\rho_{(\alpha_i \triangleleft \gamma)}(a^{\dagger}) = \rho_{\alpha_i}(a^{\dagger}) \underset{\mathfrak{b}}{\otimes} \mathrm{id}, \quad \rho_{(\beta \triangleright \delta_j)}(c^{\dagger}) = \mathrm{id} \underset{\mathfrak{b}}{\otimes} \rho_{\delta_j}(c^{\dagger}).$$

The relative tensor product is functorial in the following sense. Let $\widetilde{\mathcal{H}} = (\widetilde{H}, \widetilde{\alpha}_1, \dots, \widetilde{\alpha}_m, \widetilde{\beta})$ be a C^* - $(\mathfrak{a}_1, \dots, \mathfrak{a}_m, \mathfrak{b})$ -module, let $\widetilde{\mathcal{K}} = (\widetilde{K}, \widetilde{\gamma}, \widetilde{\delta}_1, \dots, \widetilde{\delta}_n)$ be a C^* - $(\mathfrak{b}^{\dagger}, \mathfrak{c}_1, \dots, \mathfrak{c}_n)$ -module, and let $S \in \mathcal{L}_{(s)}(\mathcal{H}, \widetilde{\mathcal{H}}), T \in \mathcal{L}_{(s)}(\mathcal{K}, \widetilde{\mathcal{K}})$. Then there exists a unique operator $S \bigotimes_{\mathfrak{b}} T \in \mathcal{L}_{(s)}(\mathcal{H} \bigotimes_{\mathfrak{b}} \mathcal{K}, \widetilde{\mathcal{H}} \bigotimes_{\mathfrak{b}} \widetilde{\mathcal{K}})$ satisfying $(S \bigotimes_{\mathfrak{b}} T)(\xi \otimes \zeta \otimes \eta) = S\xi \otimes \zeta \otimes T\eta$ for all $\xi \in \beta, \zeta \in \mathfrak{K}, \eta \in \gamma$.

Finally, the relative tensor product is associative in the following sense. Let $\mathfrak{d}, \mathfrak{e}_1, \ldots, \mathfrak{e}_l$ be C^* -bases, $\widehat{\mathcal{K}} = (K, \gamma, \delta_1, \ldots, \delta_n, \varepsilon)$ a C^* - $(\mathfrak{b}^{\dagger}, \mathfrak{c}_1, \ldots, \mathfrak{c}_n, \mathfrak{d})$ -module and $\mathcal{L} = (L, \phi, \psi_1, \ldots, \psi_l)$ a C^* - $(\mathfrak{d}^{\dagger}, \mathfrak{e}_1, \ldots, \mathfrak{e}_l)$ -module. Then there exists a canonical isomorphism

$$(2.2) \qquad a_{\mathcal{H},\mathcal{K},\mathcal{L}}: (H_{\beta \underset{\mathfrak{b}}{\otimes} \gamma} K)_{\beta \triangleright \varepsilon} \underset{\mathfrak{d}}{\otimes} \phi L \to \beta \otimes_{\rho_{\gamma}} K_{\rho_{\varepsilon}} \otimes \phi \to H_{\beta} \underset{\mathfrak{b}}{\otimes}_{\gamma \triangleleft \phi} (K_{\varepsilon} \underset{\mathfrak{d}}{\otimes} \mu L)$$

which is an isomorphism of the C^* - $(\mathfrak{a}_1, \ldots, \mathfrak{a}_m, \mathfrak{c}_1, \ldots, \mathfrak{c}_n, \mathfrak{e}_1, \ldots, \mathfrak{e}_1)$ -modules $(\mathcal{H} \bigotimes_{\mathfrak{b}} \widehat{\mathcal{K}}) \bigotimes_{\mathfrak{d}} \mathcal{L}$ and $\mathcal{H} \bigotimes_{\mathfrak{b}} (\widehat{\mathcal{K}} \bigotimes_{\mathfrak{d}} \mathcal{L})$. From now on, we identify the Hilbert spaces in (2.2) and denote them by $\mathcal{H}_{\beta} \bigotimes_{\gamma} K_{\varepsilon} \bigotimes_{\phi} \mathcal{L}$.

2.2. THE FIBER PRODUCT OF C^* -ALGEBRAS. Let $\mathfrak{b}_1, \ldots, \mathfrak{b}_n$ be C^* -bases, where $\mathfrak{b}_i = (\mathfrak{K}_i, \mathfrak{B}_i, \mathfrak{B}_i^{\dagger})$ for each *i*. A (*nondegenerate*) C^* - $(\mathfrak{b}_1, \ldots, \mathfrak{b}_n)$ -*algebra* consists of a C^* - $(\mathfrak{b}_1, \ldots, \mathfrak{b}_n)$ -module $(H, \alpha_1, \ldots, \alpha_n)$ and a (nondegenerate) C^* -algebra $A \subseteq \mathcal{L}(H)$ such that $\rho_{\alpha_i}(\mathfrak{B}_i^{\dagger})A$ is contained in *A* for each *i*. We shall only be interested in the cases n = 1, 2, where we abbreviate $A_H^{\alpha} := (H_{\alpha}, A), A_H^{\alpha, \beta} := (\alpha H_{\beta}, A)$. Given a C^* - $(\mathfrak{b}_1, \ldots, \mathfrak{b}_n)$ -algebra $\mathcal{A} = ((H, \alpha_1, \ldots, \alpha_n), A)$, we identify M(A) with a C^* -subalgebra of $\mathcal{L}([AH]) \subseteq \mathcal{L}(H)$ and obtain C^* - $(\mathfrak{b}_1, \ldots, \mathfrak{b}_n)$ -algebra $M(\mathcal{A}) = ((H, \alpha_1, \ldots, \alpha_n), M(A))$.

We need several notions of a morphism. Let $\mathcal{A} = (\mathcal{H}, A)$ and $\mathcal{C} = (\mathcal{K}, C)$ be C^* - $(\mathfrak{b}_1, \ldots, \mathfrak{b}_n)$ -algebras, where $\mathcal{H} = (H, \alpha_1, \ldots, \alpha_n)$ and $\mathcal{K} = (K, \gamma_1, \ldots, \gamma_n)$. A *-homomorphism $\pi : A \to C$ is called a *jointly (semi-)normal morphism* or briefly

(*semi-*)morphism from \mathcal{A} to \mathcal{C} if $[\mathcal{L}^{\pi}_{(s)}(\mathcal{H},\mathcal{K})\alpha_i] = \gamma_i$ for each *i*, where

$$\mathcal{L}^{\pi}_{(s)}(\mathcal{H},\mathcal{K}) = \{T \in \mathcal{L}_{(s)}(\mathcal{H},\mathcal{K}) : Ta = \pi(a)T \text{ for all } a \in A\}.$$

One easily verifies that every (semi-)morphism π between C^* -b-algebras A_H^{α} and C_K^{γ} satisfies $\pi(\rho_{\alpha}(b^{\dagger})) = \rho_{\gamma}(b^{\dagger})$ for all $b^{\dagger} \in \mathfrak{B}^{\dagger}$.

We construct a fiber product of C^* -algebras over C^* -bases as follows. Given Hilbert spaces H, K, a closed subspace $E \subseteq \mathcal{L}(H, K)$, and a C^* -algebra $A \subseteq \mathcal{L}(H)$, we define a C^* -algebra

$$\operatorname{Ind}_{E}(A) := \{T \in \mathcal{L}(K) : TE \subseteq [EA] \text{ and } T^{*}E \subseteq [EA]\} \subseteq \mathcal{L}(K).$$

Let \mathfrak{b} be a C^* -base, A_H^{β} a C^* - \mathfrak{b} -algebra, and B_K^{γ} a C^* - \mathfrak{b}^+ -algebra. The *fiber product* of A_H^{β} and B_K^{γ} is the C^* -algebra

$$A_{\beta \overset{*}{\mathfrak{b}}\gamma}B := \operatorname{Ind}_{|\beta\rangle_1}(B) \cap \operatorname{Ind}_{|\gamma\rangle_2}(A) \subseteq \mathcal{L}(H_{\beta \overset{\otimes}{\mathfrak{b}}\gamma}K).$$

To define coactions, we also need to consider the C*-algebra

$$A_{\beta} * {}^{\gamma}_{\mathfrak{h}} B := \operatorname{Ind}_{[|\beta\rangle_1 B]}(B) \cap \operatorname{Ind}_{|\gamma\rangle_2}(A) \subseteq \mathcal{L}(H_{\beta} \otimes_{\mathfrak{h}} K),$$

which evidently contains $A_{\beta} *_{\beta} B$. If *A* and *B* are unital, so is $A_{\beta} *_{b} B$, but otherwise, $A_{\beta} *_{b} B$ and $A_{\beta} *_{b} B$ may be degenerate. Clearly, conjugation by the flip $\Sigma : H_{\beta} \otimes_{p} \gamma K \to K_{\gamma} \otimes_{b} H$ yields an isomorphism

$$\mathrm{Ad}_{\Sigma}: A_{\beta} *_{\mathfrak{h}} B \to B_{\gamma} *_{\mathfrak{h}^{\dagger}} A$$

If $\mathfrak{a}, \mathfrak{c}$ are C^* -bases, $A_H^{\alpha,\beta}$ is a C^* - $(\mathfrak{a}, \mathfrak{b})$ -algebra and $B_K^{\gamma,\delta}$ a C^* - $(\mathfrak{b}^{\dagger}, \mathfrak{c})$ -algebra, then

$$A_{H}^{\alpha,\beta} \underset{\mathfrak{b}}{*} B_{K}^{\gamma,\delta} := (_{\alpha}H_{\beta} \underset{\mathfrak{b}}{\otimes} _{\gamma}K_{\delta}, A_{\beta} \underset{\mathfrak{b}}{*} _{\gamma}B)$$

is a C^* -($\mathfrak{a}, \mathfrak{c}$)-algebra, called the *fiber product* of $A_H^{\alpha,\beta}$ and $B_K^{\gamma,\delta}$; see Proposition 3.15 in [28]. Likewise, $(_{\alpha}H_{\beta} \underset{\mathfrak{b}}{\otimes} {}_{\gamma}K_{\delta}, A_{\beta} \underset{\mathfrak{b}}{*}^{\gamma}B)$ is a C^* -($\mathfrak{a}, \mathfrak{c}$)-algebra.

The fiber product need not be associative, but in this article, it will only appear as the target of a comultiplication or coaction whose coassociativity will compensate the non-associativity of the fiber product.

More importantly, the fiber product is functorial in the following sense. Let ϕ be a (semi-)morphism of C^* - $(\mathfrak{a}, \mathfrak{b})$ -algebras $\mathcal{A} = A_H^{\alpha,\beta}$ and $\mathcal{C} = C_1^{\kappa,\lambda}$, and ψ a (semi-)morphism of C^* - $(\mathfrak{b}^{\dagger}, \mathfrak{c})$ -algebras $\mathcal{B} = B_K^{\gamma,\delta}$ and $\mathcal{D} = D_M^{\mu,\nu}$. Then there exists a unique (semi-)morphism of C^* - $(\mathfrak{a}, \mathfrak{c})$ -algebras $\phi * \psi$ from $({}_{\alpha}H_{\beta} \otimes_{\mathfrak{b}} {}_{\gamma}K_{\delta}, A_{\beta} * {}_{\mathfrak{b}}^*B)$ to $({}_{\kappa}L_{\lambda} \otimes_{\mathfrak{b}} {}_{\mu}M_{\nu}, C_{\lambda} * {}_{\mathfrak{b}}^{\mu}D)$ such that

$$(\phi * \psi)(x)R = Rx$$
 for all $x \in A_{\beta} *_{\mathfrak{b}} B$ and $R \in I_M J_H + J_1 I_K$,

where $I_X = \mathcal{L}^{\phi}(H, L) \underset{\mathfrak{b}}{\otimes} \operatorname{id}_X$, $J_Y = \operatorname{id}_Y \underset{\mathfrak{b}}{\otimes} \mathcal{L}^{\psi}(K, M)$ for $X \in \{K, M\}$, $Y \in \{H, L\}$, and $\phi * \psi$ restricts to a (semi-)morphism from $A_H^{\alpha,\beta} \underset{\mathfrak{b}}{*} B_K^{\gamma,\delta}$ to $C_1^{\kappa,\lambda} \underset{\mathfrak{b}}{*} D_M^{\mu,\nu}$; see Theorem 3.19 in [28]. The proof uses the following result, which essentially is Lemma 3.18 in [28].

LEMMA 2.1. Let \mathfrak{c} be a \mathbb{C}^* -base, π a semi-morphism of \mathbb{C}^* - \mathfrak{b} -algebras A_H^{β} , \mathbb{C}_l^{λ} , and ${}_{\gamma}K_{\delta}$ a \mathbb{C}^* - $(\mathfrak{b}^{\dagger}, \mathfrak{c})$ -module. Let $I := \mathcal{L}_s^{\pi}(H_{\beta}, L_{\lambda}) \underset{\mathfrak{b}}{\otimes} \operatorname{id} \subseteq \mathcal{L}(H_{\beta} \underset{\mathfrak{b}}{\otimes} {}_{\gamma}K, L_{\lambda} \underset{\mathfrak{b}}{\otimes} {}_{\gamma}K)$ and

$$X := (I^*I)' \subseteq \mathcal{L}(H_{\beta \underset{\mathfrak{b}}{\otimes} \gamma} K), \quad Y := (II^*)' \subseteq \mathcal{L}(L_{\lambda \underset{\mathfrak{b}}{\otimes} \gamma} K).$$

(i) $\mathcal{X} := (H_{\beta \otimes_{\mathbf{h}} \gamma} K_{\delta}, X)$ and $\mathcal{Y} := (L_{\lambda \otimes_{\mathbf{h}} \gamma} K_{\delta}, Y)$ are C^* -c-algebras.

(ii) There exists a semi-morphism $\operatorname{Ind}_{|\gamma\rangle_2}(\pi) : \mathcal{X} \to \mathcal{Y}$ such that $(\operatorname{Ind}_{|\gamma\rangle_2}(\pi))(x)z = zx$ for all $x \in X$ and $z \in I$.

(iii) If B_K^{γ} is a $C^* \cdot \mathfrak{b}^\dagger$ -algebra, then $A_{\beta \mathfrak{b}^* \gamma} B \subseteq A_{\beta \mathfrak{b}^* \gamma} B \subseteq X$, $(\operatorname{Ind}_{|\gamma\rangle_2}(\pi))(A_{\beta \mathfrak{b}^* \gamma} B) \subseteq C_{\lambda \mathfrak{b}^* \gamma} B$ and $(\operatorname{Ind}_{|\gamma\rangle_2}(\pi))(A_{\beta \mathfrak{b}^* \gamma} B) \subseteq C_{\lambda \mathfrak{b}^* \gamma} B$. (iv) $[|\gamma\rangle_2 A\langle\gamma|_2] \subseteq X$ and $(\operatorname{Ind}_{|\gamma\rangle_2}(\pi))([|\gamma\rangle_2 A\langle\gamma|_2]) = [|\gamma\rangle_2 \pi(A)\langle\gamma|_2]$.

2.3. HOPF C^* -BIMODULES AND COACTIONS. The notion of a Hopf C^* -bimodule was introduced in [27].

DEFINITION 2.2. Let $\mathfrak{b} = (\mathfrak{K}, \mathfrak{B}, \mathfrak{B}^{\dagger})$ be a *C**-base. A *Hopf C**-*bimodule over* \mathfrak{b} is a *C**-($\mathfrak{b}^{\dagger}, \mathfrak{b}$)-algebra $A_{H}^{\beta,\alpha}$ with a morphism Δ from $A_{H}^{\beta,\alpha}$ to $A_{H}^{\beta,\alpha} * A_{H}^{\beta,\alpha}$ satisfying $(\delta * \mathrm{id}) \circ \delta = (\mathrm{id} * \Delta) \circ \delta$ as maps from *A* to $\mathcal{L}(H_{\alpha} \otimes_{\beta} H_{\alpha} \otimes_{\beta} H)$.

Let (\mathcal{A}, Δ) be a Hopf *C*^{*}-bimodule, where $\mathcal{A} = A_H^{\beta, \alpha}$.

A *coaction* of (\mathcal{A}, Δ) consists of a C^* -b-algebra C_K^{γ} and a semi-morphism δ from (K_{γ}, C) to $(K_{\gamma} \bigotimes_{\mathfrak{b}} \beta H_{\alpha}, C_{\gamma} \bigotimes_{\mathfrak{b}}^{\beta} A)$ such that $(\delta * \mathrm{id}) \circ \delta = (\mathrm{id} * \Delta) \circ \delta$ as maps from C to $\mathcal{L}(K_{\gamma} \bigotimes_{\mathfrak{b}} \beta H_{\alpha} \bigotimes_{\mathfrak{b}} H)$. We call such a coaction (C_K^{γ}, δ)

(i) *left-full* if $[\delta(C)|\gamma\rangle_1 A] = [|\gamma\rangle_1 A]$, and *right-full* if $[\delta(C)|\beta\rangle_2] = [|\beta\rangle_2 C]$;

(ii) *fine* if δ is injective, a morphism, and right-full, and if $[\rho_{\gamma}(\mathfrak{B}^{\dagger})C] = C$;

(iii) very fine if it is fine and if $\delta^{-1} : \delta(C) \to C$ is a morphism of C^* - \mathfrak{b} -algebras from $(K_{\gamma} \bigotimes_{\mathfrak{b}} H_{\alpha}, \delta(C))$ to (K_{γ}, C) .

A morphism between coactions (C_K^{γ}, δ_C) and $(D_1^{\varepsilon}, \delta_D)$ is a semi-morphism ρ from C_K^{γ} to $M(D)_1^{\varepsilon}$ satisfying $[\rho(C)D] = D$ and $\delta_D(d) \cdot (\rho * id)(\delta_C(c)) = \delta_D(d\rho(c))$ for all $d \in D, c \in C$. We denote the category of all coactions of (\mathcal{A}, Δ) by **Coact** $_{(\mathcal{A}, \Delta)}$.

Examples of Hopf C^* -bimodules and coactions will be discussed in detail in Sections 5, 6, and 8.

3. WEAK C*-PSEUDO-KAC SYSTEMS

To form a reduced crossed product for a coaction of a Hopf *C*^{*}-bimodule (\mathcal{A}, Δ) and to equip this reduced crossed product with a dual coaction, one needs a second Hopf *C*^{*}-bimodule $(\widehat{\mathcal{A}}, \widehat{\Delta})$ that is dual to (\mathcal{A}, Δ) in a suitable sense. We shall see that a good notion of duality is that (\mathcal{A}, Δ) and $(\widehat{\mathcal{A}}, \widehat{\Delta})$ are the legs of a weak *C*^{*}-pseudo-Kac system, which is a generalization of the balanced multiplicative unitaries and Kac systems introduced by Baaj and Skandalis [1], [3].

3.1. C^* -PSEUDO-MULTIPLICATIVE UNITARIES. A weak C^* -pseudo-Kac system is a well-behaved C^* -pseudo-multiplicative unitary V together with a symmetry Usatisfying a number of axioms. Before we state these axioms, we recall the notion of a C^* -pseudo-multiplicative unitary and the construction of the associated Hopf C^* -bimodules from [27].

Let \mathfrak{b} be a C^* -base. A C^* -pseudo-multiplicative unitary over \mathfrak{b} consists of a C^* - $(\mathfrak{b}^+, \mathfrak{b}, \mathfrak{b}^+)$ -module $(H, \widehat{\beta}, \alpha, \beta)$ and a unitary $V : H_{\widehat{\beta} \underset{\mathfrak{b}}{\otimes} \alpha} H \to H_{\alpha} \underset{\mathfrak{b}}{\otimes} \beta H$ such that

 $(3.1) \quad V(\alpha \triangleleft \alpha) = \alpha \triangleright \alpha, \quad V(\widehat{\beta} \triangleright \beta) = \widehat{\beta} \triangleleft \beta, \quad V(\widehat{\beta} \triangleright \widehat{\beta}) = \alpha \triangleright \widehat{\beta}, \quad V(\beta \triangleleft \alpha) = \beta \triangleleft \beta,$

in $\mathcal{L}(\mathfrak{K}, H_{\mathfrak{a}} \underset{\mathfrak{b}}{\otimes}_{\mathfrak{b}} H)$ and $V_{12}V_{13}V_{23} = V_{23}V_{12}$ in the sense that the following diagram

$$(3.2) \qquad \begin{array}{c} H_{\widehat{\beta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} \alpha} H_{\widehat{\beta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} \alpha}{\otimes} H \xrightarrow{V_{12}} H_{\alpha} \underset{\mathfrak{b}}{\otimes} \beta} H_{\widehat{\beta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} \alpha}{\otimes} H \xrightarrow{V_{23}} H_{\alpha} \underset{\mathfrak{b}}{\otimes} \beta} H_{\alpha} \underset{\mathfrak{b}}{\otimes} \beta}{\otimes} H, \\ \downarrow^{V_{23}} & & V_{12} \\ H_{\widehat{\beta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} \alpha \triangleright \alpha} (H_{\alpha} \underset{\mathfrak{b}}{\otimes} \beta} H) \xrightarrow{V_{13}} (H_{\widehat{\beta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} \alpha} H)_{\alpha \triangleleft \alpha} \underset{\mathfrak{b}}{\otimes} \beta} H \end{array}$$

commutes, where V_{ij} is the leg notation for the operator that acts like *V* on the *i*th and *j*th factor in the relative tensor product; see [27].

Let *V* be a C^* -pseudo-multiplicative unitary as above, let

$$\widehat{A} = \widehat{A}_V = [\langle \beta |_2 V | \alpha \rangle_2] \subseteq \mathcal{L}(H), \quad \widehat{\Delta} = \widehat{\Delta}_V : \widehat{A} \to \mathcal{L}(H_{\widehat{\beta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} \alpha H), \ \widehat{a} \mapsto V^*(1 \underset{\mathfrak{b}}{\otimes} \widehat{a})V,$$
$$A = A_V = [\langle \alpha |_1 V | \widehat{\beta} \rangle_1] \subseteq \mathcal{L}(H), \quad \Delta = \Delta_V : A \to \mathcal{L}(H_\alpha \underset{\mathfrak{b}}{\otimes} \beta H), \ a \mapsto V(a \underset{\mathfrak{b}^{\dagger}}{\otimes} 1)V^*,$$

and let $\widehat{\mathcal{A}} = \widehat{A}_{H}^{\alpha,\widehat{\beta}}$ and $\mathcal{A} = A_{H}^{\beta,\alpha}$. We call *V* well-behaved if $(\widehat{\mathcal{A}},\widehat{\Delta})$ and (\mathcal{A},Δ) are Hopf *C*^{*}-bimodules. This happens for example if *V* is *regular* in the sense that $[\langle \alpha |_1 V | \alpha \rangle_2] = [\alpha \alpha^*] \subseteq \mathcal{L}(H)$; see Theorem 4.5 in [27].

The *opposite* of *V* is the *C**-pseudo-multiplicative unitary

$$V^{\mathrm{op}} := \Sigma V^* \Sigma : H_{\beta \underset{\mathfrak{b}^{\dagger}}{\otimes} \alpha} H \xrightarrow{\Sigma} H_{\alpha \underset{\mathfrak{b}}{\otimes} \beta} H \xrightarrow{V^*} H_{\widehat{\beta} \underset{\mathfrak{b}^{\dagger}}{\otimes} \alpha} H \xrightarrow{\Sigma} H_{\alpha \underset{\mathfrak{b}}{\otimes} \widehat{\beta}} H.$$

If *V* is well-behaved or regular, then the same is true for *V*^{op}, and then

(3.3)
$$\widehat{A}_{V^{\text{op}}} = A_V, \quad \widehat{\Delta}_{V^{\text{op}}} = \operatorname{Ad}_{\Sigma} \circ \Delta_V, \quad A_{V^{\text{op}}} = \widehat{A}_V, \quad \Delta_{V^{\text{op}}} = \operatorname{Ad}_{\Sigma} \circ \widehat{\Delta}_V.$$

Let $(H, \hat{\alpha}, \hat{\beta}, \alpha, \beta)$ be a C^* - $(\mathfrak{b}, \mathfrak{b}^{\dagger}, \mathfrak{b}, \mathfrak{b}^{\dagger})$ -module and $U \in \mathcal{L}(_{\hat{\alpha}}H_{\hat{\beta}}, _{\alpha}H_{\beta})$ a symmetry, that is, $U = U^* = U^{-1}$. Then $U\hat{\alpha} = \alpha$, $U\hat{\beta} = \beta$, and the diagram



commutes, where each arrow can be read in both directions and the diagonal maps are $U \underset{\mathfrak{b}^{(\dagger)}}{\otimes} U$. We use the leg notation and write U_1 for $U \underset{\mathfrak{b}^{(\dagger)}}{\otimes} 1$ and U_2 for $1 \otimes U$.

1 ⊗ U b^(†)

For each
$$T \in \mathcal{L}(H_{\widehat{\beta}} \underset{\mathfrak{b}^{+}}{\otimes} H, H_{\alpha} \underset{\mathfrak{b}^{+}}{\otimes} H)$$
, let
 $\check{T} := \Sigma(1 \underset{\mathfrak{b}}{\otimes} U)T(1 \underset{\mathfrak{b}^{+}}{\otimes} U)\Sigma : H_{\widehat{\alpha}} \underset{\mathfrak{b}^{+}}{\otimes} H \to H_{\widehat{\beta}} \underset{\mathfrak{b}^{+}}{\otimes} H,$
 $\widehat{T} := \Sigma(U \underset{\mathfrak{b}}{\otimes} 1)T(U \underset{\mathfrak{b}^{+}}{\otimes} 1)\Sigma : H_{\alpha} \underset{\mathfrak{b}^{+}}{\otimes} H \to H_{\beta} \underset{\mathfrak{b}^{+}}{\otimes} H.$

Switching from $(\mathfrak{b}, H, \hat{\alpha}, \hat{\beta}, \alpha, \beta)$ to $(\mathfrak{b}^{\dagger}, H, \beta, \hat{\alpha}, \hat{\beta}, \alpha)$ or to $(\mathfrak{b}^{\dagger}, H, \hat{\beta}, \alpha, \beta, \hat{\alpha})$, respectively, we can iterate the assignments $T \mapsto \check{T}$ and $T \mapsto \hat{T}$, and obtain

(3.4)
$$\check{\tilde{T}} = \widehat{T}, \quad \check{\tilde{T}} = (U \bigotimes_{\mathfrak{b}} U)T(U \bigotimes_{\mathfrak{b}^{\dagger}} U) = \widehat{\tilde{T}}, \quad \check{T} = \widehat{\tilde{T}}.$$

DEFINITION 3.1. A balanced C^{*}-pseudo-multiplicative unitary (V, U) on a C^{*}- $(\mathfrak{b}, \mathfrak{b}^{\dagger}, \mathfrak{b}, \mathfrak{b}^{\dagger})$ -module $(H, \hat{\alpha}, \hat{\beta}, \alpha, \beta)$ consists of a symmetry $U \in \mathcal{L}(_{\hat{\alpha}}H_{\hat{\beta},\alpha}H_{\beta})$ and a C^{*}-pseudo-multiplicative unitary $V : H_{\hat{\beta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} H \to H_{\alpha} \underset{\mathfrak{b}}{\otimes} H$ such that \check{V} and \hat{V} are C^{*}-pseudo-multiplicative unitaries again.

Note that in this definition, (\check{V}, U) is a C^* -pseudo-multiplicative unitary if and only if (\widehat{V}, U) is one because $\check{V} = (U \underset{\mathfrak{b}^{\dagger}}{\otimes} U)\widehat{V}(U \underset{\mathfrak{b}}{\otimes} U)$.

Let (V, U) be a balanced C^* -pseudo-multiplicative unitary as above.

REMARK 3.2. (i) One easily verifies that (\check{V}, U) , (\widehat{V}, U) , (V^{op}, U) are balanced *C**-pseudo-multiplicative unitaries again. We call them the *predual*, *dual*, and *opposite* of (V, U), respectively.

(ii) The relations (3.1) for the unitaries \check{V} , \hat{V} read as follows:

$$\widehat{\beta} \lhd \widehat{\beta} \xrightarrow{\check{V}} \widehat{\beta} \rhd \widehat{\beta}, \quad \widehat{\alpha} \rhd \alpha \xrightarrow{\check{V}} \widehat{\alpha} \lhd \alpha, \quad \widehat{\alpha} \rhd \widehat{\alpha} \xrightarrow{\check{V}} \widehat{\beta} \rhd \widehat{\alpha}, \quad \alpha \lhd \widehat{\beta} \xrightarrow{\check{V}} \alpha \lhd \alpha, \beta \lhd \beta \xrightarrow{\check{V}} \beta \rhd \beta, \quad \alpha \rhd \widehat{\alpha} \xrightarrow{\check{V}} \alpha \lhd \widehat{\alpha}, \quad \alpha \rhd \alpha \xrightarrow{\check{V}} \beta \rhd \alpha, \quad \widehat{\alpha} \lhd \beta \xrightarrow{\check{V}} \widehat{\alpha} \lhd \widehat{\alpha},$$

where $X \xrightarrow{W} Y$ means WX = Y. They furthermore imply

$$\begin{split} &\widehat{\beta} \triangleright \widehat{\alpha} \xrightarrow{V} \alpha \triangleright \widehat{\alpha}, \quad \widehat{\alpha} \triangleright \beta \xrightarrow{\check{V}} \widehat{\beta} \triangleright \beta, \quad \alpha \triangleright \widehat{\beta} \xrightarrow{\widehat{V}} \beta \triangleright \widehat{\beta}, \\ &\widehat{\alpha} \triangleleft \alpha \xrightarrow{V} \widehat{\alpha} \triangleleft \beta, \quad \beta \triangleleft \widehat{\beta} \xrightarrow{\check{V}} \beta \triangleleft \alpha, \quad \widehat{\beta} \triangleleft \beta \xrightarrow{\check{V}} \widehat{\beta} \triangleleft \widehat{\alpha}. \end{split}$$

(iii) The spaces \widehat{A} and A are contained in $\mathcal{L}(H_{\widehat{\alpha}})$ since $[\widehat{A}\widehat{\alpha}] = [\langle \beta|_2 V | \alpha \rangle_2 \widehat{\alpha}] = [\langle \beta|_2 | \beta \rangle_2 \widehat{\alpha}] = [\langle \alpha|_1 V | \widehat{\beta} \rangle_1 \widehat{\alpha}] = [\langle \alpha|_1 V | \widehat{\beta} \rangle_1 \widehat{\alpha}] = \widehat{\alpha}$.

LEMMA 3.3. $V_{13}V_{23}\check{V}_{12}=\check{V}_{12}V_{13}$ and $\widehat{V}_{23}V_{12}V_{13}=V_{13}\widehat{V}_{23}$, that is, the diagrams

$$(3.5) \qquad (H_{\widehat{\alpha}} \bigotimes_{\mathfrak{b}} \mathcal{H})_{\widehat{\beta} \triangleleft \widehat{\beta}} \bigotimes_{\mathfrak{b}^{\dagger}} \mathcal{H} \xrightarrow{V_{13}} (H_{\widehat{\alpha}} \bigotimes_{\mathfrak{b}} \mathcal{H})_{\alpha \triangleleft \widehat{\beta}} \bigotimes_{\mathfrak{b}} \mathcal{H} \xrightarrow{V_{12}} (H_{\widehat{\beta}} \bigotimes_{\mathfrak{b}^{\dagger}} \mathcal{H})_{\alpha \triangleleft \alpha} \bigotimes_{\mathfrak{b}} \mathcal{H},$$
$$\overset{\check{V}_{12}}{\underset{H_{\widehat{\beta}} \bigotimes_{\mathfrak{b}^{\dagger}} \mathcal{H}}{\overset{V_{13}}{\underset{\mathfrak{b}^{\dagger}}{\overset{V_{23}}{\longrightarrow}}} H \xrightarrow{V_{23}} \mathcal{H}_{\widehat{\beta}} \bigotimes_{\mathfrak{b}^{\dagger}} \mathcal{A} (H_{\alpha} \bigotimes_{\mathfrak{b}} \mathcal{H})$$

$$(3.6) \qquad H_{\widehat{\beta}} \underset{\mathfrak{b}^{\mathfrak{a}}}{\otimes} _{\alpha \rhd \alpha} (H_{\alpha} \underset{\mathfrak{b}}{\otimes} _{\beta} H) \xrightarrow{\widehat{V}_{23}} H_{\widehat{\beta}} \underset{\mathfrak{b}^{\mathfrak{a}}}{\otimes} _{\beta \rhd \alpha} (H_{\beta} \underset{\mathfrak{b}^{\mathfrak{a}}}{\otimes} _{\widehat{\mathfrak{a}}} H) \xrightarrow{V_{13}} H_{\alpha} \underset{\mathfrak{b}}{\otimes} _{\beta \rhd \beta} (H_{\beta} \underset{\mathfrak{b}^{\mathfrak{a}}}{\otimes} _{\widehat{\mathfrak{a}}} H) \xrightarrow{\widehat{V}_{23}} H_{\widehat{\beta}} \underset{\mathfrak{b}^{\mathfrak{a}}}{\otimes} _{\beta} H \xrightarrow{V_{12}} H_{\alpha} \underset{\mathfrak{b}}{\otimes} _{\beta} H_{\alpha} \underset{\mathfrak{b}}{\otimes} _{\beta} H$$

commute.

Proof. Let $W := \Sigma V \Sigma$. We insert the relation $\check{V} = U_1 W U_1$ into the equation $\check{V}_{12}\check{V}_{13}\check{V}_{23} = \check{V}_{23}\check{V}_{12}$ and obtain $U_1W_{12}U_1 \cdot U_1W_{13}U_1 \cdot \check{V}_{23} = \check{V}_{23} \cdot U_1W_{12}U_1$ and hence $W_{12}W_{13}\check{V}_{23} = \check{V}_{23}W_{12}$. Renumbering the legs of the operators according to the permutation $(1, 2, 3) \mapsto (2, 3, 1)$, we find $V_{13}V_{23}\check{V}_{12} = \check{V}_{12}V_{13}$. A similar calculation shows that $\hat{V}_{23}V_{12}V_{13} = V_{13}\hat{V}_{23}$.

PROPOSITION 3.4. We have:

$$\widehat{A}_{\breve{V}} = UA_V U, \quad \widehat{\Delta}_{\breve{V}} = \mathrm{Ad}_{(U \underset{\mathfrak{h}}{\otimes} U)} \circ \Delta_V \circ \mathrm{Ad}_U, \quad A_{\breve{V}} = \widehat{A}_V, \quad \Delta_{\breve{V}} = \widehat{\Delta}_V,$$

and

$$A_{\widehat{V}} = U\widehat{A}_{V}U, \quad \Delta_{\widehat{V}} = \operatorname{Ad}_{(U \otimes U)} \circ \widehat{\Delta}_{V} \circ \operatorname{Ad}_{U}, \quad \widehat{A}_{\widehat{V}} = A_{V}, \quad \widehat{\Delta}_{\widehat{V}} = \Delta_{V}.$$

Proof. By definition,

$$A_{\breve{V}} = [\langle \widehat{\beta}|_1 \Sigma U_2 V U_2 \Sigma | \widehat{\alpha} \rangle_1] = [\langle U \widehat{\beta}|_2 V | U \widehat{\alpha} \rangle_2] = [\langle \beta|_2 V | \alpha \rangle_2] = \widehat{A}_V.$$

Let $\hat{a} = \langle \xi'|_2 V | \xi \rangle_2 \in \widehat{A}_V$, where $\xi' \in \beta$, $\xi \in \alpha$. Then $\Delta_{\check{V}}(\hat{a}) = \check{V}(\hat{a} \underset{\mathfrak{b}}{\otimes} 1)\check{V}^* = V^*(1 \underset{\mathfrak{b}}{\otimes} \hat{a})V = \widehat{\Delta}_V(\hat{a})$ because the diagram

$$\begin{array}{c} H_{\widehat{\beta} \underset{\mathfrak{b}^{\dagger}}{\mathfrak{b}^{\dagger}} \alpha} H \xrightarrow{V^{\ast}} H_{\widehat{\alpha} \underset{\mathfrak{b}}{\mathfrak{b}} \widehat{\beta}} H \xrightarrow{\widehat{\alpha} \underset{\mathfrak{b}}{\mathfrak{b}} \widehat{\beta}} H \xrightarrow{\widehat{\alpha} \underset{\mathfrak{b}}{\mathfrak{b}} \widehat{\beta}} H \xrightarrow{\widehat{\beta} \underset{\mathfrak{b}^{\dagger}}{\mathfrak{b}^{\dagger}} \alpha} H_{\widehat{\beta} \underset{\mathfrak{b}^{\dagger}}{\mathfrak{b}^{\dagger}} \alpha} H \xrightarrow{\widehat{\beta} \underset{\mathfrak{b}}{\mathfrak{b}^{\dagger}} \alpha} H \xrightarrow{\widehat{\beta} \underset{\mathfrak{b}}{\mathfrak{b}} \alpha} H \xrightarrow{\widehat{\beta} \underset{\mathfrak{b}}{\mathfrak{b}} \alpha} H \xrightarrow{\widehat{\beta} \underset{\mathfrak{b}}{\mathfrak{b}} \alpha} H \xrightarrow{\widehat{\beta} \underset{\mathfrak{b}}{\mathfrak{b}} \alpha} H$$

commutes. Since elements of the form like \hat{a} are dense in \hat{A}_V , we can conclude $\Delta_{\check{V}} = \hat{\Delta}_{\check{V}}$. The proof of the remaining assertions is similar.

COROLLARY 3.5. If V is well-behaved, then also \check{V} and \widehat{V} are well-behaved.

3.2. WEAK C^* -PSEUDO-KAC SYSTEMS. Let (V, U) as above.

LEMMA 3.6. For each $\hat{a} \in \hat{A}$ and $a \in A$, we have equivalences

$$(1 \underset{\mathfrak{b}^{\dagger}}{\otimes} \widehat{a})\widehat{V} = \widehat{V}(1 \underset{\mathfrak{b}}{\otimes} \widehat{a}) \quad \Leftrightarrow \quad (U\widehat{a}U \underset{\mathfrak{b}}{\otimes} 1)V = V(U\widehat{a}U \underset{\mathfrak{b}^{\dagger}}{\otimes} 1) \quad \Leftrightarrow \quad [U\widehat{a}U, \widehat{A}] = 0,$$

$$(a \underset{\mathfrak{b}^{\dagger}}{\otimes} 1)\check{V} = \check{V}(a \underset{\mathfrak{b}}{\otimes} 1) \quad \Leftrightarrow \quad (1 \underset{\mathfrak{b}}{\otimes} UaU)V = V(1 \underset{\mathfrak{b}^{\dagger}}{\otimes} UaU) \quad \Leftrightarrow \quad [UaU, A] = 0.$$

These equivalent conditions hold for all $\hat{a} \in \hat{A}$ and $a \in A$ if and only if $V_{23}\hat{V}_{12} = \hat{V}_{12}V_{23}$ and $\check{V}_{23}V_{12} = V_{12}\check{V}_{23}$ in the sense that the following diagrams commute:

Proof. This is straightforward, for example, $V_{23}\widehat{V}_{12} = \widehat{V}_{12}V_{23}$ holds if and only if $\langle \xi'|_3 V_{23} \widehat{V}_{12} | \xi \rangle_3 = \langle \xi'|_3 \widehat{V}_{12} V_{23} | \xi \rangle_3$ for all $\xi \in \alpha, \xi' \in \beta$ and hence if and only if $(1 \underset{\mathfrak{b}^{\dagger}}{\otimes} \widehat{a}) \widehat{V} = \widehat{V}(1 \underset{\mathfrak{b}}{\otimes} \widehat{a})$ for all $\widehat{a} \in \widehat{A}$.

DEFINITION 3.7. We call (V, U) a *weak* C^* -*pseudo-Kac system* if V is wellbehaved and if the equivalent conditions in Lemma 3.6 hold, and a C^* -*pseudo-Kacsystem* if V, \check{V}, \hat{V} are regular and additionally $(\Sigma(1 \bigotimes_{\mathfrak{b}} U)V)^3 = \mathrm{id}$, where $\Sigma(1 \bigotimes_{\mathfrak{b}} U)V : H_{\widehat{\beta} \underset{\star}{\otimes} \mathfrak{a}} H \to H_{\widehat{\beta} \underset{\star}{\otimes} \mathfrak{a}} H$.

REMARK 3.8. In leg notation, the equation $(\Sigma(1 \bigotimes_{\mathfrak{b}} U)V)^3 = 1$ can be rewritten as $(\Sigma U_2 V)^3 = 1$. Conjugating by Σ or V, we see that this condition is equivalent to the relation $(U_2 V \Sigma)^3 = 1$ and to the relation $(V \Sigma U_2)^3 = 1$.

LEMMA 3.9.
$$(\Sigma U_2 V)^3 = 1$$
 if and only if $\widehat{V}V\check{V} = U_1\Sigma$.
Proof. $U_1U_2(\Sigma U_2 V)^3U_2\Sigma = \Sigma U_1VU_1\Sigma \cdot V \cdot \Sigma U_2VU_2\Sigma = \widehat{V} \cdot V \cdot \check{V}$.

PROPOSITION 3.10. Every C*-pseudo-Kac system is a weak C*-pseudo-Kac system.

Proof. Let (V, U) be a C^* -pseudo-Kac system. Then $V, \check{V}, \widehat{V}$ are regular and therefore well-behaved. Using diagrams (3.2) and (3.5), we find

$$V_{12}\check{V}_{12}\Sigma_{12}V_{23} = V_{12}\check{V}_{12}V_{13}\Sigma_{12} = V_{12}V_{13}V_{23}\check{V}_{12}\Sigma_{12} = V_{23}V_{12}\check{V}_{12}\Sigma_{12}.$$

By Lemma 3.9, $V_{12}\check{V}_{12}\Sigma_{12} = \hat{V}_{12}^*U_1$ and hence $\hat{V}_{12}^*U_1V_{23} = V_{23}\hat{V}_{12}^*U_1$. Since \hat{V}_{12} is unitary and $U_1V_{23} = V_{23}U_1$, we can conclude $\hat{V}_{12}V_{23} = V_{23}\hat{V}_{12}$. A similar argument shows that $\check{V}_{23}V_{12} = V_{12}\check{V}_{23}$.

The following result is crucial for the duality presented in the next section.

PROPOSITION 3.11. Let (V, U) be a C^* -pseudo-Kac system. Then $[A\widehat{A}] = [\widehat{\alpha}\widehat{\alpha}^*]$.

Proof. The relation $[\hat{\alpha}^* \hat{A}] = \hat{\alpha}^*$ (Remark 3.2(iii)), regularity of *V*, and the relations $V^* = \Sigma U_2 V \Sigma U_2 V \Sigma U_2$ and $[V | \alpha \rangle_2 \hat{A}] = [|\beta \rangle_2 \hat{A}]$ (see Remark 4.7 in [27]) imply

$$\begin{split} [\widehat{\alpha}\widehat{\alpha}^*] &= [U\alpha\alpha^*U\widehat{A}] = [U\langle\alpha|_2 V^*|\alpha\rangle_1 U\widehat{A}] = [U\langle\alpha|_2 \Sigma U_2 V \Sigma U_2 V \Sigma U_2|\alpha\rangle_1 U\widehat{A}] \\ &= [\langle\alpha|_1 V \Sigma U_2 V|\alpha\rangle_2 \widehat{A}] = [\langle\alpha|_1 V \Sigma U_2|\beta\rangle_2 \widehat{A}] = [\langle\alpha|_1 V|\widehat{\beta}\rangle_1 \widehat{A}] = [A\widehat{A}]. \quad \blacksquare \end{split}$$

LEMMA 3.12. Let (V, U) be a (weak) C^{*}-pseudo-Kac system. Then also (\check{V}, U) , (\widehat{V}, U) , and (V^{op}, U) are (weak) C^{*}-pseudo-Kac systems.

Proof. If (V, U) is a weak C^* -pseudo-Kac system, then the tuples above are balanced C^* -pseudo-multiplicative unitaries by Remark 3.2(i), and the remaining necessary conditions follow easily from Proposition 3.4 and equation (3.3).

If (V, U) is a C^* -pseudo-Kac system, then equation (3.4), the relation $(V^{\circ p}) = U_1 V^* U_1 = (\hat{V})^{\circ p}$, and the fact that $V^{\circ p}$ is regular, imply that the tuples above satisfy the regularity condition in Definition 3.7. To check that they also satisfy the second condition, we use Remark 3.8 and calculate $(\Sigma U_2 \hat{V})^3 = (V \Sigma U_2)^3 = 1$, $(\check{V} \Sigma U_2)^3 = (\Sigma U_2 V)^3 = 1$, $(U_2 V^{\circ p} \Sigma)^3 = (U_2 \Sigma V^*)^3 = ((V \Sigma U_2)^3)^* = 1$.

3.3. THE *C**-PSEUDO-KAC SYSTEM OF A COMPACT *C**-QUANTUM GROUPOID. In [26], we introduced compact *C**-quantum groupoids and associated to each such object a regular *C**-pseudo-multiplicative unitary *V*. We now recall this construction and define a symmetry *U* such that (V, U) is a *C**-pseudo-Kac system.

A compact C^* -quantum graph consists of a unital C^* -algebra B with a faithful KMS-state μ , a unital C^* -algebra A with unital embeddings $r : B \to A$ and $s : B^{\text{op}} \to A$ such that $[r(B), s(B^{\text{op}})] = 0$, and faithful conditional expectations $\phi : A \to r(B) \cong B$ and $\psi : A \to s(B^{\text{op}}) \cong B^{\text{op}}$ such that the compositions $v := \mu \circ \phi$ and $v^{-1} := \mu^{\text{op}} \circ \psi$ are KMS-states related by some positive invertible element $\delta \in A \cap r(B)' \cap s(B^{\text{op}})'$ via the formula $v^{-1}(a) = v(\delta^{1/2}a\delta^{1/2})$, valid for all $a \in A$. An *involution* for such a compact C^* -quantum graph is a *-antiisomorphism $R : A \to A$ such that $R \circ R = \text{id}_A$, $R(r(b)) = s(b^{\text{op}})$ and $\phi(R(a)) = \psi(a)^{\text{op}}$ for all $b \in B$, $a \in A$.

Let $(B, \mu, A, r, s, \phi, \psi)$ be a compact C^* -quantum graph with involution R. We denote by $(H_{\mu}, \zeta_{\mu}, J_{\mu})$ and $(H_{\nu}, \zeta_{\nu}, J_{\nu})$ the GNS-spaces, canonical cyclic vectors, and modular conjugations for the KMS-states μ and ν , respectively, and let $\zeta_{\nu^{-1}} = \delta^{1/2} \zeta_{\nu}$. As usual, we have representations $B^{\text{op}} \to \mathcal{L}(H_{\mu}), b^{\text{op}} \mapsto J_{\mu} b^* J_{\mu}$, and $A^{\text{op}} \to \mathcal{L}(H_{\nu}), a^{\text{op}} \mapsto J_{\nu} a^* J_{\nu}$. Using the isometries

$$\zeta_{m{\phi}}: H_{\mu} o H_{
u}, \ b\zeta_{\mu} \mapsto r(b)\zeta_{
u}, \quad \zeta_{\psi}: H_{\mu^{\mathrm{op}}} o H_{
u}, \ b^{\mathrm{op}}\zeta_{\mu^{\mathrm{op}}} \mapsto s(b^{\mathrm{op}})\zeta_{
u^{-1}},$$

we define subspaces $\hat{\alpha}, \hat{\beta}, \alpha, \beta \subseteq \mathcal{L}(H_{\mu}, H_{\nu})$ by $\hat{\alpha} := [A\zeta_{\phi}], \hat{\beta} := [A\zeta_{\psi}], \beta := [A\zeta_{\psi}], \beta := [A^{\text{op}}\zeta_{\phi}], \beta := [A^{\text{op}}\zeta_{\psi}]$. Let $H = H_{\nu}$ and $\mathfrak{b} = (\mathfrak{K}, \mathfrak{B}, \mathfrak{B}^{\dagger})$, where $\mathfrak{K} = H_{\mu}, \mathfrak{B} = B \subseteq \mathcal{L}(H_{\mu}), \mathfrak{B}^{\dagger} = B^{\text{op}} \subseteq \mathcal{L}(H_{\mu})$. Then $(H, \hat{\alpha}, \hat{\beta}, \alpha, \beta)$ is a C^* - $(\mathfrak{b}, \mathfrak{b}^{\dagger}, \mathfrak{b}, \mathfrak{b}^{\dagger})$ -module and $\mathcal{A} := A_{H}^{\beta, \alpha}$ a C^* - $(\mathfrak{b}^{\dagger}, \mathfrak{b})$ -algebra [26].

A compact C^* -quantum groupoid consists of a compact C^* -quantum graph with involution as above and a morphism $\mathcal{A} \to \mathcal{A} \underset{\mathfrak{b}}{*} \mathcal{A}$ of C^* - $(\mathfrak{b}^{\dagger}, \mathfrak{b})$ -algebras satisfying the following conditions:

(i) $(\Delta * id) \circ \Delta = (id * \Delta) \circ \Delta$ as maps from A to $\mathcal{L}(H_{\alpha} \bigotimes_{\beta} H_{\alpha} \bigotimes_{\beta} H);$

(ii) $\langle \zeta_{\phi}|_{2}\Delta(a)|\zeta_{\phi}\rangle_{2} = \rho_{\beta}(\phi(a))$ and $\langle \zeta_{\psi}|_{1}\Delta(a)|\zeta_{\psi}\rangle_{1} = \rho_{\alpha}(\psi(a))$ for all $a \in A$; (iii) $[\Delta(A)|\alpha\rangle_{1}] = [|\alpha\rangle_{1}A] = [\Delta(A)|\zeta_{\psi}\rangle_{1}A]$ and similarly $[\Delta(A)|\beta\rangle_{2}] = [|\beta\rangle_{2}A]$ $= [\Delta(A)|\zeta_{\phi}\rangle_{2}A];$

(iv)
$$R(\langle \tilde{\zeta}_{\psi}|_1 \Delta(a)(d^{\mathrm{op}} \underset{h}{\otimes} 1)|\zeta_{\psi}\rangle_1) = \langle \zeta_{\psi}|_1(a^{\mathrm{op}} \underset{h}{\otimes} 1)\Delta(d)|\zeta_{\psi}\rangle_1$$
 for all $a, d \in A$.

Given a compact C^* -quantum groupoid as above, there exists a regular C^* -pseudo-multiplicative unitary $V : H_{\widehat{\beta}_{\mathfrak{b}^{\dagger}}^{\otimes \alpha}} H \to H_{\mathfrak{a} \otimes_{\mathfrak{b}}^{\otimes \beta}} H$ such that $V|a\zeta_{\psi}\rangle_1 = \Delta(a)|\zeta_{\psi}\rangle_1$ for all $a \in A$; see Theorem 5.4 in [26]. Denote by $J = J_{\nu}$ the modular conjugation for ν , by $I : H \to H$ the antiunitary given by $Ia\zeta_{\nu^{-1}} = R(a)^*\zeta_{\nu}$ for all $a \in A$, and let $U = IJ \in \mathcal{L}(H)$.

PROPOSITION 3.13. (V, U) is a C^{*}-pseudo-Kac system.

Proof. First, $U^2 = IJIJ = IJJI = II = id_H$ because IJ = JI, and $U\zeta_{\phi} = \zeta_{\psi}$, $U\zeta_{\nu} = \zeta_{\nu^{-1}}$, $U\hat{\alpha} = I\beta = \alpha$, $U\hat{\beta} = I\alpha = \beta$ by Lemma 2.7 and Proposition 3.8 in

[26]. The relation $(J_{\alpha} \bigotimes_{J_{\mu}} I)V(J_{\alpha} \bigotimes_{J_{\mu}} I) = V^*$ (see Theorem 5.6 in [26]) implies

$$\begin{split} \check{V} &= \Sigma(1 \underset{\mathfrak{b}}{\otimes} JI) V(1 \underset{\mathfrak{b}^{\dagger}}{\otimes} JI) \Sigma = (J_{\alpha} \underset{J_{\mu}}{\otimes} \widetilde{J}) \Sigma(J_{\alpha} \underset{J_{\mu}}{\otimes} \beta I) V(J_{\alpha} \underset{J_{\mu}}{\otimes} \beta I) \Sigma(J_{\widehat{\alpha}} \underset{J_{\mu}}{\otimes} \widetilde{J}) \\ &= (J_{\alpha} \underset{J_{\mu}}{\otimes} \widetilde{J}) \Sigma V^* \Sigma(J_{\widehat{\alpha}} \underset{J_{\mu}}{\otimes} \widetilde{J}) = (J_{\alpha} \underset{J_{\mu}}{\otimes} \widetilde{J}) V^{\operatorname{op}}(J_{\widehat{\alpha}} \underset{J_{\mu}}{\otimes} \widetilde{J}). \end{split}$$

But V^{op} is a regular C^* -pseudo-multiplicative unitary, so \check{V} is regular as well. In particular, (V, U) is a balanced C^* -pseudo-multiplicative unitary. We shall show that $\widehat{V}V = U_1 \Sigma \check{V}^*$, and then the claim follows from Lemma 3.9. Let $a, b \in A$ and $\omega = \widehat{V}V(a\zeta_{\psi} \otimes Ub\zeta_{\nu^{-1}})$. By Proposition 3.4, $\Delta(a) = \widehat{V}^*(1 \bigotimes a)\widehat{V}$ and hence

$$\begin{split} \omega &= \widehat{V}\Delta(a)(\zeta_{\psi} \otimes Ub\zeta_{\nu^{-1}}) = (1 \underset{\mathfrak{b}^{\dagger}}{\otimes} a)\widehat{V}(\zeta_{\psi} \otimes Ub\zeta_{\nu^{-1}}) \\ &= \Sigma(U \underset{\mathfrak{b}}{\otimes} 1)(UaU \underset{\mathfrak{b}}{\otimes} 1)V(b\zeta_{\nu^{-1}} \otimes \zeta_{\psi}) = \Sigma(U \underset{\mathfrak{b}}{\otimes} 1)(UaU \underset{\mathfrak{b}}{\otimes} 1)\Delta(b)(\zeta_{\psi} \otimes \zeta_{\nu^{-1}}). \end{split}$$

Since $UaU = JIaIJ = R(a)^{\text{op}}$ and $[UaU \underset{\mathfrak{b}}{\otimes} 1, \Delta(b)] \in [A^{\text{op}} \underset{\mathfrak{b}}{\otimes} 1, A_{\alpha} \underset{\mathfrak{b}}{*}_{\beta} A] = 0$,

$$\omega = \Sigma(U \underset{\mathfrak{b}}{\otimes} 1)\Delta(b)(UaU\zeta_{\psi} \otimes \zeta_{\nu^{-1}}) = (U \underset{\mathfrak{b}^{\dagger}}{\otimes} 1)\Sigma(U \underset{\mathfrak{b}}{\otimes} U)\Delta(b)(U \underset{\mathfrak{b}}{\otimes} U)(aU\zeta_{\psi} \otimes \zeta_{\nu}).$$

By Proposition 3.4, $\check{V}^*(1 \underset{\mathfrak{b}^{\dagger}}{\otimes} UbU)\check{V} = (U \underset{\mathfrak{b}}{\otimes} U)\Delta(b)(U \underset{\mathfrak{b}}{\otimes} U)$ and hence

$$\begin{split} \omega &= (U \underset{\mathfrak{b}^{\dagger}}{\otimes} 1) \Sigma \check{V}^* (1 \underset{\mathfrak{b}^{\dagger}}{\otimes} UbU) \check{V} (aU\zeta_{\psi} \otimes \zeta_{\nu}) \\ &= (U \underset{\mathfrak{b}^{\dagger}}{\otimes} 1) \Sigma \check{V}^* (1 \underset{\mathfrak{b}^{\dagger}}{\otimes} UbU) \Sigma (1 \underset{\mathfrak{b}}{\otimes} U) V (\zeta_{\nu} \otimes UaU\zeta_{\psi}). \end{split}$$

Finally, by Proposition 5.5 in [26], $V(\zeta_{\nu} \otimes UaU\zeta_{\psi}) = \zeta_{\nu} \otimes UaU\zeta_{\phi}$, whence

$$\omega = (U \underset{\mathfrak{b}^{\dagger}}{\otimes} 1) \Sigma \check{V}^* (1 \underset{\mathfrak{b}^{\dagger}}{\otimes} UbU) (aU\zeta_{\phi} \otimes \zeta_{\nu}) = (U \underset{\mathfrak{b}^{\dagger}}{\otimes} 1) \Sigma \check{V}^* (a\zeta_{\psi} \otimes Ub\zeta_{\nu^{-1}}). \quad \blacksquare$$

4. REDUCED CROSSED PRODUCTS AND DUALITY

Let (V, U) be a weak C^* -pseudo-Kac system and let (\mathcal{A}, Δ) , $(\widehat{\mathcal{A}}, \widehat{\Delta})$ be the Hopf C^* -bimodules associated to V as in the preceding section. Generalizing the corresponding constructions and results for coactions of Hopf C^* -algebras [3], we now associate to every coaction of one of these Hopf C^* -bimodules a reduced crossed product that carries a dual coaction of the other Hopf C^* -bimodule, and prove a duality theorem concerning the iteration of this construction.

4.1. REDUCED CROSSED PRODUCTS FOR COACTIONS OF (\mathcal{A}, Δ) . Let δ be a coaction of the Hopf C^* -bimodule (\mathcal{A}, Δ) on a C^* -b-algebra $\mathcal{C} = C_K^{\gamma}$ and let (the notation $C \rtimes_r \widehat{A}$ is consistent with [3] but not with [10], where $C \rtimes_r A$ is used instead):

$$C \rtimes_{\mathbf{r}} \widehat{A} := [\delta(C)(1 \underset{\mathfrak{b}}{\otimes} \widehat{A})] \subseteq \mathcal{L}(K_{\gamma} \underset{\mathfrak{b}}{\otimes}_{\beta} H), \quad \mathcal{C} \rtimes_{\mathbf{r}} \widehat{\mathcal{A}} := (K_{\gamma} \underset{\mathfrak{b}}{\otimes}_{\beta} H_{\widehat{\beta}}, C \rtimes_{\mathbf{r}} \widehat{A}).$$

PROPOSITION 4.1. (i) $[\delta(C)(\gamma \triangleright \widehat{\beta})] \subseteq \gamma \triangleright \widehat{\beta}$ with equality if δ is left-full. (ii) $C \rtimes_{\mathbf{r}} \widehat{A}$ is a C^* -algebra and $\mathcal{C} \rtimes_{\mathbf{r}} \widehat{\mathcal{A}}$ is a C^* - \mathfrak{b}^+ -algebra.

(iii) There exist nondegenerate *-homomorphisms $C \to M(C \rtimes_r \widehat{A})$ and $\widehat{A} \to M(C \rtimes_r \widehat{A})$ \widehat{A}), given by $c \mapsto \delta(c)$ and $\widehat{a} \mapsto 1 \bigotimes \widehat{a}$, respectively.

Proof. (i) The relation $\hat{\beta} = [A\hat{\beta}]$ (see Proposition 3.2 (ii) in [27]) implies that

 $[\delta(C)|\gamma\rangle_1\widehat{\beta}] = [\delta(C)|\gamma\rangle_1A\widehat{\beta}] \subseteq [|\gamma\rangle_1A\widehat{\beta}] = [|\gamma\rangle_1\widehat{\beta}].$ (ii) We first show that $[(1 \bigotimes_{\mathfrak{b}} \widehat{A})\delta(C)] \subseteq [\delta(C)(1 \bigotimes_{\mathfrak{b}} \widehat{A})].$ Let $\delta^{(2)} := (\mathrm{id} * \Delta) \circ$ $\delta = (\delta * \mathrm{id}) \circ \delta : C \to \mathcal{L}(K_{\gamma \bigotimes_{\beta}} H_{\gamma \bigotimes_{\beta}} H).$ By definition of \widehat{A} and Δ ,

$$\begin{split} [(1 \underset{\mathfrak{b}}{\otimes} \widehat{A})\delta(C)] &= [\langle \beta|_{3}(1 \underset{\mathfrak{b}}{\otimes} V)|\alpha\rangle_{3}\delta(C)] = [\langle \beta|_{3}(1 \underset{\mathfrak{b}}{\otimes} V)(\delta(C) \underset{\mathfrak{b}^{\dagger}}{\otimes} 1)|\alpha\rangle_{3}] \\ &= [\langle \beta|_{3}\delta^{(2)}(C)(1 \underset{\mathfrak{b}}{\otimes} V)|\alpha\rangle_{3}] \subseteq [\delta(C)\langle \beta|_{3}(1 \underset{\mathfrak{b}}{\otimes} V)|\alpha\rangle_{3}] = [\delta(C)(1 \underset{\mathfrak{b}}{\otimes} \widehat{A})]. \end{split}$$

Consequently, $C \rtimes_{\mathbf{r}} \widehat{A}$ is a C^* -algebra. By Proposition 3.2(i) in [27], $[\widehat{A}\rho_{\widehat{\beta}}(\mathfrak{B})] = \widehat{A}$, and hence $[(C \rtimes_{\mathbf{r}} \widehat{A})\rho_{(\gamma \triangleright \widehat{\beta})}(\mathfrak{B})] = [\delta(C)(1 \bigotimes_{\mathfrak{h}} \widehat{A}\rho_{\widehat{\beta}}(\mathfrak{B}))] = [\delta(C)(1 \bigotimes_{\mathfrak{h}} \widehat{A})] = C \rtimes_{\mathbf{r}} \widehat{A}.$ (iii) Immediate.

THEOREM 4.2. There exists a unique coaction $\hat{\delta}$ of $(\hat{A}, \hat{\Delta})$ on $\mathcal{C} \rtimes_{\mathbf{r}} \hat{\mathcal{A}}$ such that $\hat{\delta}(\delta(c)(1 \bigotimes_{\mathfrak{b}} \hat{a})) = (\delta(c) \bigotimes_{\mathfrak{b}^{\dagger}} 1)(1 \bigotimes_{\mathfrak{b}} \hat{\Delta}(\hat{a}))$ for all $c \in C$, $\hat{a} \in \hat{A}$. If $\hat{\Delta}$ is a fine coaction, then $\hat{\delta}$ is a very fine coaction. If δ is left-full, then $\hat{\delta}$ is left-full.

Proof. Define $\hat{\delta} : C \rtimes_{\mathbf{r}} \hat{A} \to \mathcal{L}(K_{\gamma \bigotimes_{\mathbf{b}} \beta} H_{\hat{\beta} \bigotimes_{\mathbf{b}^{\dagger}} H})$ by $x \mapsto (1 \bigotimes_{\mathbf{b}} \check{V})(x \bigotimes_{\mathbf{b}} 1)(1 \bigotimes_{\mathbf{b}} \check{V})(x \bigotimes_{\mathbf{b}} I)(1 \bigotimes_{\mathbf{b}} \check{V})(x \bigotimes_{\mathbf{b}} I)(x \bigotimes_{\mathbf{b}} I$ \check{V}^*). Then $\hat{\delta}$ is injective and satisfies $\hat{\delta}(\delta(c)(1 \otimes \widehat{a})) = (\delta(c) \otimes 1)(1 \otimes \widehat{\Delta}(\widehat{a}))$ for all $c \in C$, $\hat{a} \in \widehat{A}$ because $\check{V}(\widehat{a} \bigotimes_{\mathfrak{b}} 1)\check{V}^* = \widehat{\Delta}(\widehat{a})$ by Proposition 3.4 and $(1 \bigotimes_{\mathfrak{b}} \check{V})\delta(c)(1 \bigotimes_{\mathfrak{b}} \check{V})\delta(c)$ \check{V}^*) = $\delta(c)$ as a consequence of the relation $\check{V}(a \otimes 1)\check{V}^* = a \otimes 1$. We show that $\widehat{\delta}$ is a coaction of $(\widehat{\mathcal{A}}, \widehat{\Delta})$. First, $[\widehat{\delta}(C \rtimes_r \widehat{A}) | \alpha \rangle_3] \subseteq [|\alpha \rangle_3(C \rtimes_r \widehat{A})]$ because

(4.1)
$$[(\delta(C) \underset{\mathfrak{b}^{\dagger}}{\otimes} 1)(1 \underset{\mathfrak{b}}{\otimes} \widehat{\Delta}(\widehat{A}))|\alpha\rangle_{3}] \subseteq [(\delta(C) \underset{\mathfrak{b}^{\dagger}}{\otimes} 1)|\alpha\rangle_{3}(1 \underset{\mathfrak{b}}{\otimes} \widehat{A})] = [|\alpha\rangle_{3}\delta(C)(1 \underset{\mathfrak{b}}{\otimes} \widehat{A})].$$

Next, $[\widehat{\delta}(C \rtimes_{\mathbf{r}} \widehat{A})|\gamma \triangleright \widehat{\beta}\rangle_1 \widehat{A}] \subseteq [|\gamma \triangleright \widehat{\beta}\rangle_1 \widehat{A}]$ because by Proposition 4.1(i),

$$[(1 \bigotimes_{\mathfrak{b}} \widehat{\Delta}(\widehat{A}))(\delta(C) \bigotimes_{\mathfrak{b}^{\dagger}} 1)|\gamma \triangleright \widehat{\beta}\rangle_{1}\widehat{A}] \subseteq [(1 \bigotimes_{\mathfrak{b}} \widehat{\Delta}(\widehat{A}))|\gamma \triangleright \widehat{\beta}\rangle_{1}\widehat{A}]$$

(4.2)
$$= [|\gamma\rangle_{1}\widehat{\Delta}(\widehat{A})|\widehat{\beta}\rangle_{1}\widehat{A}] \subseteq [|\gamma\rangle_{1}|\widehat{\beta}\rangle_{1}\widehat{A}].$$

Furthermore, $\hat{\delta}(x)(1 \bigotimes_{\mathbf{b}} \check{V})|\xi\rangle_3 = (1 \bigotimes_{\mathbf{b}} \check{V})|\xi\rangle_3 x$ for each $x \in C \rtimes_{\mathbf{r}} \widehat{A}, \xi \in \widehat{\beta}$, and by Remark 3.2(ii), $[(1 \otimes \check{V})|\widehat{\beta}\rangle_3(\gamma \triangleright \widehat{\beta})] = \gamma \triangleright \widehat{\beta} \triangleright \widehat{\beta}$ and $[\langle \widehat{\beta}|_3(1 \otimes \check{V})^*(\gamma \triangleright \widehat{\beta} \triangleright \widehat{\beta})] = \langle \widehat{\beta} \lor \widehat{\beta} \lor \widehat{\beta} \rangle$ $\gamma \triangleright \widehat{\beta}$. The maps $(\widehat{\delta} * \mathrm{id}) \circ \widehat{\delta}$ and $(\mathrm{id} * \widehat{\Delta}) \circ \widehat{\delta}$ from $C \rtimes_{\mathrm{r}} \widehat{A}$ to $\mathcal{L}(K_{\gamma \bigotimes_{\mathbf{b}} \beta} H_{\widehat{\beta} \bigotimes_{\mathbf{b}} \alpha} H_{\widehat{\beta} \bigotimes_{\mathbf{b}} \alpha} H_{\widehat{\beta} \bigotimes_{\mathbf{b}} \alpha} H)$ are given by $\delta(c)(1 \bigotimes_{b} \widehat{a}) \mapsto (\delta(c) \bigotimes_{b^{\dagger}} 1 \bigotimes_{b^{\dagger}} 1)(1 \bigotimes_{b} \widehat{\Delta}^{(2)}(\widehat{a}))$ for all $c \in C, \widehat{a} \in \widehat{A}$, where $\widehat{\Delta}^{(2)} := (\widehat{\Delta} * \mathrm{id}) \circ \widehat{\Delta} = (\mathrm{id} * \widehat{\Delta}) \circ \widehat{\Delta}$. Thus, $(C \rtimes_{\mathbf{r}} \widehat{A}, \widehat{\delta})$ is a coaction of $(\widehat{A}, \widehat{\Delta})$. If the coactions $\widehat{\Delta}$ is fine, then the inclusion (4.1) is an equality and in any case $[\langle \widehat{\beta}|_{3}(1 \bigotimes_{b} \check{V})^{*}(\gamma \triangleright \widehat{\beta} \triangleright \widehat{\beta})] = \gamma \triangleright \widehat{\beta}$, whence $\widehat{\delta}$ will be very fine. If δ is left-full, then the inclusion (4.2) is an equality by Proposition 4.1(i) and hence $\widehat{\delta}$ is left full.

the inclusion (4.2) is an equality by Proposition 4.1(i) and hence $\hat{\delta}$ is left-full.

DEFINITION 4.3. We call $\mathcal{C} \rtimes_{\mathbf{r}} \widehat{\mathcal{A}}$ the *reduced crossed product* and $(\mathcal{C} \rtimes_{\mathbf{r}} \widehat{\mathcal{A}}, \widehat{\delta})$ the *reduced dual coaction* of (\mathcal{C}, δ) .

The construction of reduced dual coactions is functorial in the following sense:

PROPOSITION 4.4. Let ρ be a morphism between coactions $(\mathcal{C}, \delta_{\mathcal{C}})$ and $(\mathcal{D}, \delta_{\mathcal{D}})$ of (\mathcal{A}, Δ) . Then there exists a unique morphism $\rho \rtimes_{\mathbf{r}} \operatorname{id} from (\mathcal{C} \rtimes_{\mathbf{r}} \widehat{\mathcal{A}}, \widehat{\delta}_{\mathcal{C}})$ to $(\mathcal{D} \rtimes_{\mathbf{r}} \widehat{\mathcal{A}}, \widehat{\delta}_{\mathcal{D}})$ such that $(\rho \rtimes_{\mathbf{r}} \operatorname{id})((1 \otimes \widehat{a})\delta_{\mathcal{C}}(c)) \cdot \delta_{\mathcal{D}}(d)(1 \otimes \widehat{a}') = (1 \otimes \widehat{a})\delta_{\mathcal{D}}(\rho(c)d)(1 \otimes \widehat{a}')$ for all $c \in C, d \in D, \widehat{a}, \widehat{a}' \in \widehat{\mathcal{A}}$.

Proof. The semi-morphism $\operatorname{Ind}_{|\beta\rangle_2}(\rho)$ of Lemma 2.1 evidently restricts to a semi-morphism $\rho \rtimes_r \operatorname{id}$ from $\mathcal{C} \rtimes_r \widehat{\mathcal{A}}$ to $M(\mathcal{D} \rtimes_r \widehat{\mathcal{A}})$ which satisfies the formula given above, and this formula implies that $\rho \rtimes_r \operatorname{id}$ is a morphism of coactions as claimed.

COROLLARY 4.5. There exists a functor $- \rtimes_{\mathbf{r}} \widehat{\mathcal{A}} : \mathbf{Coact}_{(\mathcal{A}, \Delta)} \to \mathbf{Coact}_{(\widehat{\mathcal{A}}, \widehat{\Delta})}$ given by $(\mathcal{C}, \delta) \mapsto (\mathcal{C} \rtimes_{\mathbf{r}} \widehat{\mathcal{A}}, \widehat{\delta})$ and $\rho \mapsto \rho \rtimes_{\mathbf{r}} \mathrm{id}$.

4.2. REDUCED CROSSED PRODUCTS FOR COACTIONS OF $(\widehat{\mathcal{A}}, \widehat{\Delta})$. The construction in the preceding paragraph carries over to coactions of the Hopf *C*^{*}-bimodule $(\widehat{\mathcal{A}}, \widehat{\Delta})$ as follows. Let δ be a coaction of $(\widehat{\mathcal{A}}, \widehat{\Delta})$ on a *C*^{*}-b[†]-algebra $\mathcal{C} = C_K^{\gamma}$ and let

$$C \rtimes_{\mathbf{r}} A := [\delta(C)(1 \underset{\mathfrak{b}^{\dagger}}{\otimes} UAU)] \subseteq \mathcal{L}(K_{\gamma \underset{\mathfrak{b}^{\dagger}}{\otimes} \mathfrak{a}} H), \quad C \rtimes_{\mathbf{r}} \mathcal{A} = (K_{\gamma \underset{\mathfrak{b}^{\dagger}}{\otimes} \mathfrak{a}} H_{\widehat{\mathfrak{a}}}, C \rtimes_{\mathbf{r}} A).$$

Using straightforward modifications of the preceding proofs, one shows:

- **PROPOSITION 4.6.** (i) $[\delta(C)(\gamma \triangleright \widehat{\alpha})] \subseteq \gamma \triangleright \widehat{\alpha}$ with equality if δ is fine.
- (ii) $C \rtimes_{\mathbf{r}} A$ is a C^* -algebra and $C \rtimes_{\mathbf{r}} A$ is a C^* -b-algebra.

(iii) There exist nondegenerate *-homomorphisms $C \to M(C \rtimes_{\mathbf{r}} A)$ and $A \to M(C \rtimes_{\mathbf{r}} A)$, given by $c \mapsto \delta(c)$ and $a \mapsto \underset{\mathfrak{b}^{\dagger}}{1 \otimes a}$, respectively.

THEOREM 4.7. There exists a unique coaction $(\mathcal{C} \rtimes_{\mathbf{r}} \mathcal{A}, \widehat{\delta})$ of (\mathcal{A}, Δ) such that $\widehat{\delta}(\delta(c)(1 \underset{\mathfrak{b}^{\dagger}}{\otimes} UaU)) = (\delta(c) \underset{\mathfrak{b}}{\otimes} 1)(1 \underset{\mathfrak{b}^{\dagger}}{\otimes} \operatorname{Ad}_{(U \underset{\mathfrak{b}}{\otimes} 1)} \Delta(a))$ for all $c \in C$, $a \in A$. If Δ is a fine coaction, then $\widehat{\delta}$ is a very fine coaction. If δ is left-full, then $\widehat{\delta}$ is left-full.

DEFINITION 4.8. Let (\mathcal{C}, δ) be a coaction of $(\widehat{\mathcal{A}}, \widehat{\Delta})$. Then we call $\mathcal{C} \rtimes_{\mathbf{r}} \mathcal{A}$ the *reduced crossed product* and $(\mathcal{C} \rtimes_{\mathbf{r}} \mathcal{A}, \widehat{\delta})$ the *reduced dual coaction* of (\mathcal{C}, δ) .

PROPOSITION 4.9. Let ρ be a morphism between coactions $(\mathcal{C}, \delta_{\mathcal{C}})$ and $(\mathcal{D}, \delta_{\mathcal{D}})$ of $(\widehat{\mathcal{A}}, \widehat{\Delta})$. Then there exists a unique morphism $\rho \rtimes_{\mathbf{r}} \operatorname{id} from (\mathcal{C} \rtimes_{\mathbf{r}} \mathcal{A}, \widehat{\delta}_{\mathcal{C}})$ to $(\mathcal{D} \rtimes_{\mathbf{r}} \mathcal{A}, \widehat{\delta}_{\mathcal{D}})$ such that $(\rho \rtimes_{\mathbf{r}} \operatorname{id})((1 \bigotimes_{\mathfrak{b}^{\dagger}} UaU)\delta_{\mathcal{C}}(c)) \cdot \delta_{\mathcal{D}}(d)(1 \bigotimes_{\mathfrak{b}^{\dagger}} Ua'U) = (1 \bigotimes_{\mathfrak{b}^{\dagger}} UaU)\delta_{\mathcal{D}}(\rho(c)d)(1 \bigotimes_{\mathfrak{b}^{\dagger}} Ua'U)$ for all $c \in C$, $d \in D$, $a, a' \in A$.

COROLLARY 4.10. There exists a functor $-\rtimes_{\mathbf{r}} \mathcal{A} : \mathbf{Coact}_{(\widehat{\mathcal{A}},\widehat{\Delta})} \to \mathbf{Coact}_{(\mathcal{A},\Delta)}$ given by $(\mathcal{C}, \delta) \mapsto (\mathcal{C} \rtimes_{\mathbf{r}} \mathcal{A}, \widehat{\delta})$ and $\rho \mapsto \rho \rtimes_{\mathbf{r}} \mathrm{id}$.

4.3. THE DUALITY THEOREM. The preceding constructions yield for each coaction $(\mathcal{C}, \delta_{\mathcal{C}})$ of (\mathcal{A}, Δ) and each coaction $(\mathcal{D}, \delta_{\mathcal{D}})$ of $(\widehat{\mathcal{A}}, \widehat{\Delta})$ a bidual $(\mathcal{C} \rtimes_r \widehat{\mathcal{A}} \rtimes_r \mathcal{A}, \widehat{\widehat{\delta}_{\mathcal{C}}})$ and $(\mathcal{D} \rtimes_r \mathcal{A} \rtimes_r \widehat{\mathcal{A}}, \widehat{\widehat{\delta}_{\mathcal{D}}})$, respectively. The following generalization of the Baaj–Skandalis duality theorem [3] identifies these biduals in the case where (V, U) is a C^* -pseudo-Kac system and the initial coactions are fine. Morally, it says that up to Morita equivalence, the functors $- \rtimes_r \widehat{\mathcal{A}}$ and $- \rtimes_r \mathcal{A}$ implement an equivalence of the categories $\mathbf{Coact}_{(\mathcal{A}, \mathcal{A})}^f$ and $\mathbf{Coact}_{(\widehat{\mathcal{A}}, \widehat{\mathcal{A}})}^f$.

THEOREM 4.11. Assume that (V, U) is a C^{*}-pseudo-Kac system.

(i) Let (\mathcal{C}, δ) be a (very) fine coaction of (\mathcal{A}, Δ) , where $\mathcal{C} = C_K^{\gamma}$. Then there exists an isomorphism $\Phi : \mathbb{C} \rtimes_r \widehat{\mathcal{A}} \rtimes_r \mathcal{A} \to [|\beta\rangle_2 \mathbb{C}\langle\beta|_2] \subseteq \mathcal{L}(K_{\gamma \bigotimes_{\mathfrak{b}} \beta} H)$ such that Φ^{-1} is an (iso)morphism from $(K_{\gamma \bigotimes_{\mathfrak{b}} \beta} H_{\widehat{\mathfrak{a}}}, [|\beta\rangle_2 \mathbb{C}\langle\beta|_2])$ to $\mathbb{C} \rtimes_r \widehat{\mathcal{A}} \rtimes_r \mathcal{A}$ and $\widehat{\delta} \circ \Phi^{-1} = (\Phi^{-1} * \mathrm{id}) \circ \mathrm{Ad}_{(1 \otimes \Sigma \widehat{\mathcal{V}})} \circ \mathrm{Ind}_{|\beta\rangle_2}(\delta)$.

(ii) Let (\mathcal{D}, δ) be a (very) fine coaction of $(\widehat{\mathcal{A}}, \widehat{\Delta})$, where $\mathcal{D} = D_{l}^{\varepsilon}$. Then there exists an isomorphism $\Phi : D \rtimes_{\mathbf{r}} A \rtimes_{\mathbf{r}} \widehat{A} \cong [|\alpha\rangle_{2} D\langle\alpha|_{2}] \subseteq \mathcal{L}(L_{\varepsilon \otimes_{\alpha}} H)$ such that Φ^{-1} is an (iso)morphism from $(L_{\varepsilon \otimes_{\alpha}} H_{\widehat{\beta}}, [|\alpha\rangle_{2} D\langle\alpha|_{2}])$ to $\mathcal{D} \rtimes_{\mathbf{r}} \mathcal{A} \rtimes_{\mathbf{r}} \widehat{\mathcal{A}}$ and $\widehat{\widehat{\delta}} \circ \Phi^{-1} = (\Phi^{-1} * \mathrm{id}) \circ \mathrm{Ad}_{(1 \otimes_{\alpha} \Sigma V)} \circ \mathrm{Ind}_{|\alpha\rangle_{2}}(\delta)$.

Proof. We only prove (i); then (ii) follows after replacing (V, U) by (\check{V}, U) . By Proposition 3.4 and Proposition 3.11, applied to the *C**-pseudo-Kac system (\check{V}, U) , we have $[\widehat{A} \operatorname{Ad}_U(A)] = [A_{\check{V}} \widehat{A}_{\check{V}}] = [\beta \beta^*]$, and since δ is fine,

$$[|\beta\rangle_2 C\langle\beta|_2] = [\delta(C)(1 \underset{\mathfrak{b}}{\otimes} \beta\beta^*)] = [\delta(C)(1 \underset{\mathfrak{b}}{\otimes} \widehat{A} \operatorname{Ad}_U(A))].$$

One easily verifies that the *-homomorphism $\operatorname{Ind}_{|\beta\rangle_2}(\delta)$ (see Lemma 2.1) yields an (iso)morphism of *C**-b-algebras

$$\operatorname{Ind}_{|\beta\rangle_{2}}(\delta):(K_{\gamma}\underset{\mathfrak{b}}{\otimes}_{\beta}H_{\widehat{\alpha}},[|\beta\rangle_{2}C\langle\beta|_{2}])\to(K_{\gamma}\underset{\mathfrak{b}}{\otimes}_{\beta}H_{\alpha}\underset{\mathfrak{b}}{\otimes}_{\beta}H_{\widehat{\alpha}},[|\beta\rangle_{2}\delta(C)\langle\beta|_{2}]).$$

Denote by Ψ the composition of this (iso)morphism with $\operatorname{Ad}_{(1 \otimes V^*)}_{\mathfrak{b}}$ and let $\delta^{(2)} = (\delta * \operatorname{id}) \circ \delta = (\operatorname{id} * \Delta) \circ \delta$. Let $x = \delta(c)(1 \otimes \widehat{a} U a U) \in [|\beta\rangle_2 C \langle \beta|_2]$, where $c \in C, \widehat{a} \in C$.

 $\widehat{A}, a \in A. \text{ By Lemma 3.6,}$ $\Psi(x) = \operatorname{Ad}_{(1 \otimes V^*)}(\delta^{(2)}(c)(1 \otimes 1 \otimes \widehat{a}UaU)) = (\delta(c) \otimes 1)(1 \otimes \widehat{\Delta}(\widehat{a}))(1 \otimes 1 \otimes UaU).$

Consequently, $\Psi([|\beta\rangle_2 C\langle\beta|_2]) = C \rtimes_r \widehat{A} \rtimes_r A$. Next, the relations $\mathcal{C} \rtimes_r \widehat{A} \rtimes_r \mathcal{A} = (K_{\gamma} \bigotimes_{\mathfrak{b}} H_{\widehat{\beta}} \bigotimes_{\mathfrak{b}^{\dagger}} H_{\widehat{\alpha}}, C \rtimes_r \widehat{A} \rtimes_r A)$ and $(1 \bigotimes_{\mathfrak{b}} V^*)(\gamma \triangleright \alpha \triangleright \widehat{\alpha}) = \gamma \triangleright \widehat{\beta} \triangleright \widehat{\alpha}$ imply that Ψ is a

morphism of C^* -b-algebras as claimed. Using the definition of $\hat{\delta}$, Proposition 3.4, and Lemma 3.6, we find

$$\begin{split} \widehat{\delta}(\Psi(x)) &= (\delta(c) \underset{\mathfrak{b}^{\dagger}}{\otimes} 1 \underset{\mathfrak{b}}{\otimes} 1)(1 \underset{\mathfrak{b}}{\otimes} \widehat{\Delta}(\widehat{a}) \underset{\mathfrak{b}}{\otimes} 1)(1 \underset{\mathfrak{b}^{\dagger}}{\otimes} 1 \underset{\mathfrak{b}^{\dagger}}{\otimes} \operatorname{Ad}_{(U \underset{\mathfrak{b}^{\dagger}}{\otimes} 1)}(\Delta(a))) \\ &= \operatorname{Ad}_{(1 \underset{\mathfrak{b}}{\otimes} V^* \underset{\mathfrak{b}}{\otimes} 1)}((\delta^{(2)}(c) \underset{\mathfrak{b}}{\otimes} 1)(1 \underset{\mathfrak{b}}{\otimes} 1 \underset{\mathfrak{b}^{\bullet}}{\otimes} \widehat{a} \underset{\mathfrak{b}^{\bullet}}{\otimes} 1)(1 \underset{\mathfrak{b}^{\bullet}}{\otimes} 1 \underset{\mathfrak{b}^{\bullet}}{\otimes} \operatorname{Ad}_{(U \underset{\mathfrak{b}}{\otimes} 1)}(\Delta(a)))) \\ &= (\Psi * \operatorname{id})((\delta(c) \underset{\mathfrak{b}}{\otimes} 1)(1 \underset{\mathfrak{b}^{\bullet}}{\otimes} \widehat{a} \underset{\mathfrak{b}^{\bullet}}{\otimes} 1)(1 \underset{\mathfrak{b}^{\bullet}}{\otimes} \operatorname{Ad}_{(U \underset{\mathfrak{b}}{\otimes} 1)}(\Delta(a)))) \\ &= (\Psi * \operatorname{id})((1 \underset{\mathfrak{b}}{\otimes} \Sigma \widehat{V})\delta^{(2)}(c)(1 \underset{\mathfrak{b}^{\bullet}}{\otimes} 1 \underset{\mathfrak{b}^{\bullet}}{\otimes} 2 \operatorname{Ad}_{U}(1 \underset{\mathfrak{b}^{\bullet}}{\otimes} \widehat{V}^* \Sigma)) \\ &= (\Psi * \operatorname{id})((1 \underset{\mathfrak{b}}{\otimes} \Sigma \widehat{V})(\operatorname{Ind}_{|\beta\rangle_2}(\delta)(x))(1 \underset{\mathfrak{b}^{\bullet}}{\otimes} \widehat{V}^* \Sigma)). \quad \blacksquare \end{split}$$

5. THE C*-PSEUDO-KAC SYSTEM OF A GROUPOID

For the remainder of this article, we fix a locally compact, Hausdorff, second countable groupoid *G* with a left Haar system λ . In [27], we associated to such a groupoid a regular *C*^{*}-pseudo-multiplicative unitary *V* and identified the underlying *C*^{*}-algebras of the Hopf *C*^{*}-bimodules $(\widehat{A}, \widehat{\Delta})$ and (\mathcal{A}, Δ) of *V* with the function algebra $C_0(G)$ and the reduced groupoid *C*^{*}-algebra $C_r^*(G)$, respectively. We now recall this construction and define a symmetry *U* such that (V, U)becomes a *C*^{*}-pseudo-Kac system. For background on groupoids, see [20], [23].

Denote by λ^{-1} the right Haar system associated to λ and let μ be a measure on the unit space G^0 with full support. We denote the range and the source map of *G* by *r* and *s*, respectively, let $G^u := r^{-1}(u)$ and $G_u := s^{-1}(u)$ for each $u \in G^0$, and define measures ν, ν^{-1} on *G* such that

$$\int_{G} f \, \mathrm{d}\nu = \int_{G^0} \int_{G^u} f(x) \, \mathrm{d}\lambda^u(x) \, \mathrm{d}\mu(u), \quad \int_{G} f \, \mathrm{d}\nu^{-1} = \int_{G^0} \int_{G_u} f(x) \, \mathrm{d}\lambda_u^{-1}(x) \, \mathrm{d}\mu(u)$$

for all $f \in C_c(G)$. We assume that μ is quasi-invariant in the sense that ν and ν^{-1} are equivalent, and denote by $D := \int d\nu / \int d\nu^{-1}$ the Radon–Nikodym derivative. One can choose D such that it is a Borel homomorphism (see page 89 in [20]), and we do so.

We identify functions in $C_{b}(G^{0})$ and $C_{b}(G)$ with multiplication operators on the Hilbert spaces $L^{2}(G^{0}, \mu)$ and $L^{2}(G, \nu)$, respectively, and let $\mathfrak{K} = L^{2}(G^{0}, \mu)$, $\mathfrak{B} = \mathfrak{B}^{\dagger} = C_{0}(G^{0}) \subseteq \mathcal{L}(\mathfrak{K}), \mathfrak{b} = (\mathfrak{K}, \mathfrak{B}, \mathfrak{B}^{\dagger}) = \mathfrak{b}, H = L^{2}(G, \nu).$ Pulling functions on G^0 back to G along r or s, we obtain representations r^* : $C_0(G^0) \to C_b(G) \hookrightarrow \mathcal{L}(H)$ and $s^* : C_0(G^0) \to C_b(G) \hookrightarrow \mathcal{L}(H)$. We define Hilbert $C^*-C_0(G^0)$ -modules $L^2(G, \lambda)$ and $L^2(G, \lambda^{-1})$ as the respective completions of the pre- C^* -module $C_c(G)$, the structure maps being given by

$$\begin{aligned} \langle \xi' | \xi \rangle(u) &= \int_{G^u} \overline{\xi'(x)} \xi(x) \, d\lambda^u(x), \quad \xi f = r^*(f) \xi \quad \text{in the case of } L^2(G, \lambda), \\ \langle \xi' | \xi \rangle(u) &= \int_{G_u} \overline{\xi'(x)} \xi(x) \, d\lambda_u^{-1}(x), \quad \xi f = s^*(f) \xi \quad \text{in the case of } L^2(G, \lambda^{-1}) \end{aligned}$$

respectively, for all $\xi, \xi' \in C_c(G), u \in G^0, f \in C_0(G^0)$. Then there exist isometric embeddings $j : L^2(G, \lambda) \to \mathcal{L}(\mathfrak{K}, H)$ and $\hat{j} : L^2(G, \lambda^{-1}) \to \mathcal{L}(\mathfrak{K}, H)$ such that

$$(j(\xi)\zeta)(x) = \xi(x)\zeta(r(x)), \quad (\hat{j}(\xi)\zeta)(x) = \xi(x)D^{-1/2}(x)\zeta(s(x))$$

for all $\xi \in C_{c}(G)$, $\zeta \in C_{c}(G^{0})$. Let $\alpha = \beta := j(L^{2}(G,\lambda))$ and $\widehat{\alpha} = \widehat{\beta} := \widehat{j}(L^{2}(G,\lambda^{-1}))$. Then $(H,\widehat{\alpha},\widehat{\beta},\alpha,\beta)$ is a C^{*} - $(\mathfrak{b},\mathfrak{b}^{\dagger},\mathfrak{b},\mathfrak{b}^{\dagger})$ -module, $\rho_{\alpha} = \rho_{\beta} = r^{*}$ and $\rho_{\widehat{\alpha}} = \rho_{\widehat{\beta}} = s^{*}$, and $j(\xi)^{*}j(\xi') = \langle \xi | \xi' \rangle$ and $\widehat{j}(\eta)^{*}\widehat{j}(\eta') = \langle \eta | \eta' \rangle$ for all $\xi, \xi' \in L^{2}(G,\lambda), \eta, \eta' \in L^{2}(G,\lambda^{-1})$; see Section 2.3 in [27].

The Hilbert spaces $H_{\hat{\beta}} \underset{\mathfrak{b}^+}{\otimes_{\mathfrak{b}^+}} H$ and $H_{\alpha} \underset{\mathfrak{b}}{\otimes_{\mathfrak{b}^+}} H$ can be described as follows. Define measures $v_{s,r}^2$ on $G_s \times_r G$ and $v_{r,r}^2$ on $G_r \times_r G$ such that

(5.1)
$$\int_{G_r \times_{\mathbf{r}} G} f \int d\nu_{s,r}^2 = \int_{G^0} \int_{G^u} \int_{G^{s(x)}} f(x,y) \, d\lambda^{s(x)}(y) \, d\lambda^u(x) \, d\mu(u),$$
$$\int_{G_r \times_{\mathbf{r}} G} g \int d\nu_{r,r}^2 = \int_{G^0} \int_{G^u} \int_{G^u} g(x,y) \, d\lambda^u(y) \, d\lambda^u(x) \, d\mu(u)$$

for all $f \in C_c(G_s \times_r G)$, $g \in C_c(G_r \times_r G)$. Then there exist unitaries

$$\Phi: H_{\widehat{\beta}\underset{\mathfrak{b}^{\dagger}}{\otimes}_{\mathfrak{a}}} H \to L^{2}(G_{s} \times_{\mathbf{r}} G, v_{s,r}^{2}) \quad \text{and} \quad \Psi: H_{\mathfrak{a}}\underset{\mathfrak{b}}{\otimes}_{\beta} H \to L^{2}(G_{r} \times_{\mathbf{r}} G, v_{r,r}^{2})$$

such that for all $\eta, \xi \in C_c(G), \zeta \in C_c(G^0)$,

$$\begin{split} &\Phi(\widehat{j}(\eta) \oslash \zeta \oslash j(\xi))(x,y) = \eta(x)D^{-1/2}(x)\zeta(s(x))\xi(y), \\ &\Psi(j(\eta) \oslash \zeta \oslash j(\xi))(x,y) = \eta(x)\zeta(r(x))\xi(y). \end{split}$$

From now on, we use these isomorphisms without further notice.

THEOREM 5.1. There exists a C^{*}-pseudo-Kac system (V, U) on $(H, \hat{\alpha}, \hat{\beta}, \alpha, \beta)$ such that for all $\omega \in C_c(G_s \times_r G), (x, y) \in G_r \times_r G, \xi \in C_c(G), z \in G,$

(5.2)
$$(V\omega)(x,y) = \omega(x,x^{-1}y)$$
 and $(U\xi)(x) = \xi(x^{-1})D(x)^{-1/2}$

Proof. By Theorem 2.5 and Example 4.3(ii) in [27], there exists a regular C^* -pseudo-multiplicative unitary V as claimed. The second formula in (5.2) defines a unitary $U \in \mathcal{L}(H)$ by definition of the Radon–Nikodym derivative D =

 $d\nu/d\nu^{-1}$, and $U^2 = id$ because $(U^2\xi)(x) = (U\xi)(x^{-1})D(x)^{-1/2} = \xi(x)D(x)^{1/2}$ $D(x)^{-1/2} = \xi(x)$ for all $\xi \in C_c(G)$ and $x \in G$. The unitary $\widehat{V} = \Sigma U_1 V U_1 \Sigma$ is equal to $V^{op} = \Sigma V^* \Sigma$ because

$$(U_1 V U_1 \omega)(x, y) = (V U_1 \omega)(x^{-1}, y) D(x)^{-1/2} = (U_1 \omega)(x^{-1}, xy) D(x)^{-1/2}$$
$$= \omega(x, xy) D(x^{-1})^{-1/2} D(x)^{-1/2} = \omega(x, xy)$$

for all $\omega \in C_c(G_r \times_r G)$, $(x, y) \in G_s \times_r G$. In particular, \widehat{V} is a regular C^* -pseudomultiplicative unitary. It remains to show that the map $Z := \Sigma U_2 V : H_{\widehat{\beta}_{p+1}^{\otimes \alpha}} H \to$

 $H_{\widehat{\beta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} _{\mathfrak{a}} H$ satisfies $Z^3 = 1$. But for all $\omega \in C_{\mathsf{c}}(G_s \times_{\mathsf{r}} G)$ and $(x, y) \in G_s \times_{\mathsf{r}} G$,

$$\begin{aligned} (Z\omega)(x,y) &= (V\omega)(y,x^{-1})D(x)^{-1/2} = \omega(y,y^{-1}x^{-1})D(x)^{-1/2}, \\ (Z^3\omega)(x,y) &= (Z^2\omega)(y,y^{-1}x^{-1})D(x)^{-1/2} = (Z\omega)(y^{-1}x^{-1},xyy^{-1})(D(x)D(y))^{-1/2} \\ &= \omega(x,x^{-1}xy)(D(x)D(y)D(y^{-1}x^{-1}))^{-1/2} = \omega(x,y). \end{aligned}$$

The Hopf C^* -bimodules $(\widehat{\mathcal{A}}, \widehat{\Delta})$ and (\mathcal{A}, Δ) associated to *V* can be described as follows; see Theorem 3.16 in [27]. Given $g \in C_c(G)$, define $L(g) \in C_r^*(G) \subseteq \mathcal{L}(H)$ by

$$(L(g)f)(x) = \int_{G^{r(x)}} g(z)f(z^{-1}x)D^{-1/2}(z) \,\mathrm{d}\lambda^{r(x)}(z)$$

for all $x \in G$, $f \in C_c(G) \subseteq L^2(G, \nu) = H$. Then

(5.3)
$$\widehat{A} = C_0(G) \subseteq \mathcal{L}(H), \quad (\widehat{\Delta}(f)\omega)(x,y) = f(xy)\omega(x,y),$$

 $A = C_r^*(G), \quad (\Delta(L(g))\omega')(x',y') = \int_{G^{u'}} g(z)D^{-1/2}(z)\omega'(z^{-1}x',z^{-1}y')d\lambda^{u'}(z)$

for all $f \in C_0(G)$, $\omega \in C_c(G_s \times_r G)$, $(x, y) \in G_s \times_r G$ and $g \in C_c(G)$, $\omega' \in C_c(G_r \times_r G)$, $(x', y') \in G_r \times_r G$, where u' = r(x') = r(y'). We shall loosely refer to $C_0(G)$ and $C_r^*(G)$ as Hopf *C**-bimodules, having in mind $(\widehat{A}, \widehat{\Delta})$ and (\mathcal{A}, Δ) , respectively.

6. ACTIONS OF G AND COACTIONS OF $C_0(G)$

Let *G* be a groupoid and consider $C_0(G)$ as a Hopf *C*^{*}-bimodule as in the preceding section. Then coactions of $C_0(G)$ can be related to actions of *G* as follows. Let us say that a tuple (**F**, **G**, η , ε) is *an embedding of a category* **C** *into a category* **D** *as a full and coreflective subcategory* if **F** : **C** \rightarrow **D** is a full and faithful functor and **G** : **D** \rightarrow **C** is a faithful right adjoint to **F**, where η : id_C \rightarrow **GF** is the unit and ε : **FG** \rightarrow id_D is the counit of the adjunction; see also Section IV.3 in [18]. In this section, we construct such an embedding of the category of actions of *G* on

continuous $C_0(G^0)$ -algebras into the category of certain admissible coactions of $C_0(G)$. We keep the notation introduced in the preceding section.

6.1. $C_0(G^0)$ -ALGEBRAS AND C^* -b-ALGEBRAS. We shall embed the category of admissible $C_0(G^0)$ -algebras into the category of admissible C^* -b-algebras as a full and coreflective subcategory.

Recall that a $C_0(X)$ -algebra, where X is some locally compact Hausdorff space, is a C^* -algebra C with a fixed nondegenerate *-homomorphism of $C_0(X)$ into the center of the multiplier algebra M(C) [6], [14]. We denote the fiber of a $C_0(X)$ -algebra C at a point $x \in X$ by C_x and write the quotient map $p_x : C \to C_x$ as $c \mapsto c_x$. Recall that C is a *continuous* $C_0(X)$ -algebra if the map $X \to \mathbb{R}$ given by $x \mapsto ||c_x||$ is continuous for each $c \in C$. A *morphism* of $C_0(X)$ -algebras C, D is a nondegenerate *-homomorphism $\pi : C \to M(D)$ such that $\pi(fc) = f\pi(c)$ for all $f \in C_0(X), c \in C$.

DEFINITION 6.1. We call a $C_0(G^0)$ -algebra C admissible if it is continuous and if $C_u \neq 0$ for each $u \in G^0$, and we call a C^* -b-algebra C_K^{γ} admissible if $[\rho_{\gamma}(C_0(G^0))C] = C$ and $[C\gamma] = \gamma$. A morphism between admissible C^* -b-algebras C_K^{γ} , D_1^{ε} is a semi-morphism π from C_K^{γ} to $M(D)_1^{\varepsilon}$ that is nondegenerate in the sense that $[\pi(C)D] = D$. Denote by $\mathbf{C_0}(\mathbf{G^0})$ -alg^{*a*} the category of all admissible $C_0(G^0)$ -algebras, and by \mathbf{C}^* -b-alge^{*a*} the category of all admissible C^* -b-algebras.

LEMMA 6.2. (i) Let C_K^{γ} be an admissible C*- \mathfrak{b} -algebra. Then C is an admissible $C_0(G^0)$ -algebra with respect to ρ_{γ} .

(ii) Let π be a morphism between admissible C*-b-algebras C_K^{γ} and D_1^{ε} . Then π is a morphism of $C_0(G^0)$ -algebras from (C, ρ_{γ}) to (D, ρ_{ε}) .

Proof. (i) First, note that $\rho_{\gamma}(C_0(G^0)) \subseteq M(C)$ is central because $C \subseteq \mathcal{L}(K_{\gamma}) \subseteq \rho_{\gamma}(C_0(G^0))'$. The map $C \hookrightarrow \mathcal{L}(K_{\gamma}) \cong \mathcal{L}(\gamma)$ is a faithful field of representations in the sense of Theorem 3.3 in [6], and therefore *C* is a continous $C_0(G^0)$ -algebra. We have $C_u \neq 0$ for each $u \in G^0$ because otherwise $C = [CI_u]$, where $I_u = C_0(G^0 \setminus \{u\})$, and then $[\gamma^*\gamma] = [\gamma^*C\gamma] = [\gamma^*I_uC\gamma] = [\gamma^*\gamma I_u] = I_u \neq C_0(G^0)$, contradicting the fact that K_{γ} is a C^* -b-module.

(ii) This is Lemma 3.4 in [28].

We embed $C_0(G^0)$ -**a**lg^{*a*} into C^* -**b**-**a**lg^{*a*} using a KSGNS-construction for the following kind of weights.

DEFINITION 6.3. A $C_0(G^0)$ -weight on a $C_0(G^0)$ -algebra C is a $C_0(G^0)$ -linear, positive map $\phi : C \to C_0(G^0)$. We denote the set of all such weights by $\mathcal{W}(C)$.

Let *C* be an admissible $C_0(G^0)$ -algebra. The results in [4] imply:

LEMMA 6.4.
$$\bigcap_{\phi \in \mathcal{W}(C)} \ker \phi = \{0\} \text{ and } \left[\bigcup_{\phi \in \mathcal{W}(C)} \phi(C) \right] = C_0(G^0).$$

Let $\phi \in \mathcal{W}(C)$. Then ϕ is completely positive by Theorem 3.9 in [21] and bounded by Lemma 5.1 in [16]. Let $E_{\phi} = C \otimes_{\phi} \mathfrak{K}$ (see Section 1) and define $\eta_{\phi} : C \to \mathcal{L}(E_{\phi})$ and $l_{\phi} : C \to \mathcal{L}(\mathfrak{K}, E_{\phi})$ by $\eta_{\phi}(c)(d \otimes_{\phi} \zeta) = cd \otimes_{\phi} \zeta$ and $l_{\phi}(c)\zeta = c \otimes_{\phi} \zeta$ for all $c, d \in C, \zeta \in \mathfrak{K}$. One easily verifies that for all $c, d \in C, f \in C_0(G^0), \zeta \in \mathfrak{K}$,

(6.1)
$$l_{\phi}(c)^* l_{\phi}(d) = \phi(c^*d), \quad l_{\phi}(c)f = l_{\phi}(cf), \\ \eta_{\phi}(c)(d \otimes_{\phi} f\zeta) = cdf \otimes_{\phi} \zeta = \eta_{\phi}(cf)(d \otimes_{\phi} \zeta).$$

The *universal* $C_0(G^0)$ -*representation* $\eta_C : C \to \mathcal{L}(E_C)$ of *C* is the direct sum of the representations $\eta_{\phi} : C \to \mathcal{L}(E_{\phi})$, where $\phi \in \mathcal{W}(C)$. Denote by $l_C \subseteq \mathcal{L}(\mathfrak{K}, E_C)$ the closed linear span of all maps $l_{\phi}(c) : \mathfrak{K} \to E_{\phi} \hookrightarrow E_C$, where $c \in C$, $\phi \in \mathcal{W}(C)$.

LEMMA 6.5. $\eta_C(C)_{E_C}^{l_C}$ is an admissible C^* -b-algebra and η_C is an isomorphism of $C_0(G^0)$ -algebras from C to $(\eta_C(C), \rho_{l_C})$.

Proof. The definition of l_C , the equations (6.1) and Lemma 6.4 imply that $[l_C \mathfrak{K}] = \bigoplus_{\phi} E_{\phi} = E_C$, $[l_C^* l_C] = \left[\bigcup_{\phi} \phi(C)\right] = C_0(G^0)$ and $[l_C C_0(G^0)] = l_C$, whence (E_C, l_C) is a C^* -b-module, and that $[\eta_C(C)\rho_{l_C}(C_0(G^0))] = [\eta_C(CC_0(G^0))] = \eta_C(C)$ and $[\eta_C(C)l_C] = l_C$, whence $\eta_C(C)_{E_C}^{l_C}$ is an admissible C^* -b-algebra. Lemma 6.4 implies that η_C is injective and hence an isomorphism of C onto $\eta_C(C)$, and the last equation in (6.1) implies that $\eta_C(c)\rho_\gamma(f) = \eta_C(cf)$ for all $c \in C$, $f \in C_0(G^0)$.

THEOREM 6.6. There exists an embedding as a full and coreflective subcategory $(\mathbf{F}, \mathbf{G}, \eta, \varepsilon)$ of $\mathbf{C}_0(\mathbf{G}^0)$ -**a**lg^{*a*} into \mathbf{C}^* -b-**a**lg^{*a*} such that the following conditions hold:

(i) **F** is given by $C \mapsto \eta_C(C)_{E_C}^{l_C}$ on objects and by $\mathbf{F}\pi : \eta_C(c) \mapsto \eta_D(\pi(c))$ for each morphism π between objects C, D in $\mathbf{C}_0(\mathbf{G}^0)$ - \mathbf{alg}^a ;

(ii) **G** is given by $C_K^{\gamma} \mapsto (C, \rho_{\gamma})$ on objects and $\pi \mapsto \pi$ on morphisms;

- (iii) η_C is defined as above for each object C in $C_0(G^0)$ -alg^{*a*};
- (iv) $\varepsilon_{\mathcal{C}} = \eta_{\mathbf{GC}}^{-1}$ for each object \mathcal{C} in \mathbf{C}^* - \mathfrak{b} - \mathbf{alg}^a .

Proof. The functor $G : C^*-\mathfrak{b}-alg^a \to C_0(G^0)-alg^a$ is well defined by Lemma 6.2 and evidently faithful.

Let *C* be an admissible $C_0(G^0)$ -algebra, $\mathcal{D} = D_K^{\gamma}$ an admissible C^* -b-algebra, and $\pi : C \to \mathbf{G}\mathcal{D}$ a morphism in $\mathbf{C}_0(\mathbf{G}^0)$ -alg^{*a*}. We claim that $\pi \circ \eta_C^{-1}$ is a morphism from FC to \mathcal{D} in \mathbf{C}^* -b-alg^{*a*}. Let $\xi \in \gamma$. Then the map $\phi : C \to C_0(G^0) \subseteq \mathcal{L}(\mathfrak{K})$ given by $c \mapsto \xi^*\pi(c)\xi$ is a $C_0(G^0)$ -weight, and there exists an isometry $S : E_{\phi} \to K$ such that $S(c \otimes_{\phi} \zeta) = \pi(c)\xi\zeta$ for all $c \in C, \zeta \in \mathfrak{K}$. Denote by $P : E_C \to E_{\phi}$ the natural projection. Then $[SPl_C] = [Sl_{\phi}(C)] = [\pi(C)\xi]$ lies in γ and contains ξ , and $SP\eta_C(c) = S\eta_{\phi}(c) = \pi(c)$ for each $c \in C$. Since $\xi \in \gamma$ was arbitrary, the claim follows. Using Lemma 6.5, we conclude that **F** is well defined and that η is a natural isomorphism from id to **GF**. Indeed, if $\pi : C \to D$ is a morphism in $C_0(G^0)$ -alg^{*a*}, then $F\pi = \eta_D \circ \pi \circ \eta_C^{-1}$ is a morphism from FC to FD by the argument above.

Finally, let \mathcal{D} be an admissible C^* -b-algebra. The argument above, applied to the identity on $G\mathcal{D}$, yields a morphism $\varepsilon_{\mathcal{D}}$ from $FG\mathcal{D}$ to \mathcal{D} in C^* -b-alg^{*a*} such that the composition $G\mathcal{D} \xrightarrow{\eta_{G\mathcal{D}}} GFG\mathcal{D} \xrightarrow{G\varepsilon_{\mathcal{D}}} G\mathcal{D}$ is the identity. Since η is a natural transformation, also ε : $FG \rightarrow$ id is one. For each admissible $C_0(G^0)$ -algebra C, the composition $FC \xrightarrow{F\eta_C} FGFC \xrightarrow{\varepsilon_{FC}} FC$ is the identity by construction. From Theorem 2 of Section IV.1 in [18], we can conclude that F is a left adjoint to G such that η and ε form the unit and counit, respectively, of the adjunction. Since η is a natural isomorphism, F is full and faithful by Theorem 1 of Section IV.3 in [18].

6.2. ACTIONS OF *G* AND COACTIONS OF $C_0(G)$. We next embed the category of admissible actions of *G* as a full and coreflective subcategory into the category of all admissible coactions of $C_0(G)$.

The definition of an action of *G* requires the following preliminaries. Given $C_0(G^0)$ -algebras (C, ρ) and (D, σ) , where *D* is commutative, we denote by $C_\rho \boxtimes_{\sigma} D$ the $C_0(G^0)$ -tensor product [5], and drop the subscript ρ or σ if this map is understood. Given a $C_0(G^0)$ -algebra *C* and a continuous surjection $t : G \to G^0$, we consider $C_0(G)$ as a $C_0(G^0)$ -algebra via $t^* : C_0(G^0) \to M(C_0(G))$ and let $t^*C := C \boxtimes_{t^*} C_0(G)$, which is a $C_0(G)$ -algebra in a natural way. Each morphism π of $C_0(G^0)$ -algebras *C*, *D* induces a morphism of $t^*\pi$ of $C_0(G)$ -algebras from t^*C to t^*D via $c \boxtimes f \mapsto \pi(c) \boxtimes f$. An *action* of *G* on a $C_0(G^0)$ -algebra *C* is an isomorphism $\sigma : s^*C \to r^*C$ of $C_0(G)$ -algebras such that the restrictions of σ to the fibers satisfy $\sigma_x \circ \sigma_y = \sigma_{xy}$ for all $(x, y) \in G_s \times_r G$ [17]. A *morphism* between actions (C, σ^C) and (D, σ^D) of *G* is a morphism of $C_0(G^0)$ -algebras π from *C* to *D* satisfying $\sigma^D \circ s^*\pi = r^*\pi \circ \sigma^C$.

DEFINITION 6.7. We call an action (C, σ) of *G* admissible if the $C_0(G^0)$ -algebra *C* is admissible, and we call a coaction (C_K^{γ}, δ) of $C_0(G)$ admissible if C_K^{γ} is an admissible C^* -b-algebra and $[\delta(C)(1 \otimes C_0(G))] = C \bigotimes_h C_0(G)$ in $\mathcal{L}(K_{\gamma} \bigotimes_h H)$.

REMARK 6.8. If σ is an action of G on a continuous $C_0(G^0)$ -algebra, then the subset $Y := \{u \in G^0 : C_u \neq 0\} \subseteq G^0$ is open, C is an admissible $C_0(Y)$ -algebra, and σ restricts to an action of the subgroupoid $G|_Y := \{x \in G : r(x), s(x) \in Y\} \subseteq G$.

LEMMA 6.9. Let C_K^{γ} and D_1^{ε} be admissible C^* -b-algebras, where D is commutative. Then there exists an isomorphism $C_{\rho_{\gamma}} \boxtimes_{\rho_{\varepsilon}} D \to C_{\gamma} \bigotimes_{\mathfrak{b}} \mathcal{E}D$, $c \boxtimes d \mapsto c \bigotimes_{\mathfrak{b}} d$.

Proof. Use Lemma 2.7 in [5] and apply Proposition 4.1 in [5] to the field of representations $C \hookrightarrow \mathcal{L}(K_{\gamma}) \cong \mathcal{L}(\gamma)$, noting that $\gamma \otimes_{\rho_{\varepsilon}} D \cong [|\gamma\rangle_1 D]$ as a Hilbert C^* -*D*-module via $\xi \otimes d \mapsto |\xi\rangle_1 d$ and that $(C_{\gamma} \bigotimes_{\varepsilon} D)[|\gamma\rangle_1 D] \subseteq [|\gamma\rangle_1 D]$.

We use the isomorphism above without further notice.

PROPOSITION 6.10. (i) Let (C_K^{γ}, δ) be an admissible coaction of $C_0(G)$. There exists a unique action σ_{δ} of G on (C, ρ_{γ}) given by $c \boxtimes f \mapsto \delta(c)(1 \bigotimes f)$.

(ii) Let (C, σ) be an admissible action of G. There exists a unique admissible, injective coaction δ_{σ} of $C_0(G)$ on **F**C given by $\eta_C(c) \mapsto (r^*\eta_C)(\sigma(c \boxtimes 1))$.

Proof. (i) Since $\delta(C)$ and $1 \bigotimes_{\mathfrak{b}} C_0(G)$ commute, there exists a unique *-homomorphism $\tilde{\sigma}$ from the algebraic tensor product $C \odot C_0(G)$ to r^*C such that $\tilde{\sigma}(c \odot f) = \delta(c)(1 \bigotimes f)$ for all $c \in C$, $f \in C_0(G)$. Since δ is a coaction, $\delta(c\rho_{\gamma}(g)) = \delta(c)\rho_{(\gamma \triangleright \hat{\beta})}(g) = \delta(c)(1 \bigotimes s^*(g))$ for all $g \in C_0(G^0)$. From Lemma 2.7 in [5], we can conclude that $\tilde{\sigma}$ factorizes to a *-homomorphism $\sigma = \sigma_{\delta} : s^*C \to r^*C$ satisfying the formula in (i). This σ is surjective because $[\delta(C)(1 \bigotimes C_0(G))] = C \bigotimes C_0(G)$. In particular, σ_x is surjective for each $x \in G$. We claim that $\sigma_x \circ \sigma_y = \sigma_{xy}$ for all $(x, y) \in G_s \times_r G$. Define $r_1 : G_s \times_r G \to G^0$ by $(x, y) \mapsto r(x)$. By Lemma 6.9, we have isomorphisms $C_{\gamma \bigotimes_{\mathfrak{a}} C_0(G)_{\widehat{\mathfrak{b}} \bigotimes_{\mathfrak{b}} C_0(G)} \cong C \boxtimes_{r^*} C_0(G)_{s^*} \boxtimes_{r^*} C_0(G) \cong C \boxtimes_{r^*} C_0(G)_{s^*} \boxtimes_{r^*} C_0(G) \cong C \boxtimes_{r^*}$

(6.2)
$$\sigma_{x} \circ \sigma_{y} \circ p_{s(y)} = \sigma_{x} \circ p_{y} \circ \delta = p_{(x,y)} \circ (\delta * \mathrm{id}) \circ \delta,$$
$$\sigma_{xy} \circ p_{s(y)} = p_{xy} \circ \delta = p_{(x,y)} \circ (\mathrm{id} * \widehat{\Delta}) \circ \delta,$$

and the claim follows. Finally, $\sigma_u = id_{C_u}$ for each $u \in G^0$ because σ_u is surjective and idempotent, and σ_x is injective for each $x \in G$ because $\sigma_{s(x)} = \sigma_{x^{-1}} \circ \sigma_x$ is injective. Therefore, σ is injective.

(ii) Let $D := \eta_C(C)_{l_C \bigotimes_{\mathfrak{b}} \alpha} C_0(G)$. Then $\mathcal{D} := ((E_C)_{l_C \bigotimes_{\mathfrak{b}} \alpha} H_{\widehat{\beta}}, D)$ is an admissible C^* -b-algebra. Define $\delta : C \to D$ by $c \mapsto (r^*\eta_C)(\sigma(c \boxtimes 1))$. Let $c \in C$, $g \in C_0(G^0)$. Then $cg \boxtimes 1 = c \boxtimes s^*(g)$ in $M(C \boxtimes_{s^*} C_0(G))$ and therefore $\delta(cg) = \delta(c)(1 \bigotimes_{\mathfrak{b}} s^*(g)) = \delta(c)\rho_{(l_C \triangleright_{\widehat{\beta}})}(g)$. Consequently, δ is a morphism of $C_0(G^0)$ -algebras from C to $(D, \rho_{l_C \triangleright_{\widehat{\beta}}}) = \mathbf{G}\mathcal{D}$. By definition of \mathbf{F} and ε , the morphism $\delta_{\sigma} := \varepsilon_{\mathcal{D}} \circ \mathbf{F}\delta : \mathbf{F}C \to \mathbf{F}\mathbf{G}\mathcal{D} \to \mathcal{D}$ satisfies $\delta_{\sigma} \circ \eta_C = \delta$, and a similar calculation as in (6.2) shows that $(\delta_{\sigma} * \mathrm{id}) \circ \delta_{\sigma} = (\mathrm{id} * \widehat{\Delta}) \circ \delta_{\sigma}$. Consequently, δ_{σ} is a coaction of $(\widehat{\mathcal{A}}, \widehat{\Delta})$. Since σ is injective, so are δ and δ_{σ} . Finally, δ_{σ} is admissible because $[\delta_{\sigma}(\eta_C(C))(1 \bigotimes_{\mathfrak{b}} C_0(G))] = (r^*\eta_C)(\sigma(s^*C)) = r^*\eta_C(C) = [\eta_C(C) \bigotimes_{\mathfrak{b}} C_0(G)]$.

COROLLARY 6.11. Every admissible coaction of $C_0(G)$ is injective, left-full, and right-full.

Proof. If (C_K^{γ}, δ) is an admissible coaction, then the relations $[C_0(G)\alpha] = \alpha$ and $[C\gamma] = \gamma$ imply $[\delta(C)|\alpha\rangle_2] = [\delta(C)(1 \otimes C_0(G))|\alpha\rangle_2] = [(C \otimes C_0(G))|\alpha\rangle_2] = [|\alpha\rangle_2 C]$ and $[\delta(C)|\gamma\rangle_1 C_0(G)] = [\delta(C)(1 \otimes C_0(G))|\gamma\rangle_1] = [(C \otimes$ $[|\gamma\rangle_1 C_0(G)]$. Finally, δ is injective because σ_δ is injective and $\delta(c) = \sigma_\delta(c \boxtimes 1)$ for all $c \in C$.

PROPOSITION 6.12. Let $(\mathcal{C}, \delta^{\mathcal{C}})$, $(\mathcal{D}, \delta^{\mathcal{D}})$ be admissible coactions with associated actions $\sigma^{\mathcal{C}} = \sigma_{\delta^{\mathcal{C}}}, \sigma^{\mathcal{D}} = \sigma_{\delta^{\mathcal{D}}}$, and let $\pi \in \mathbf{C}^* \cdot \mathfrak{b} - \mathbf{alg}^a(\mathcal{C}, \mathcal{D}) = \mathbf{C_0}(\mathbf{G}^0) - \mathbf{alg}^a(\mathbf{G}\mathcal{C}, \mathbf{G}\mathcal{D})$. Then $(\pi * \mathrm{id}) \circ \delta^{\mathcal{C}} = \delta^{\mathcal{D}} \circ \pi$ if and only if $r^* \pi \circ \sigma^{\mathcal{C}} = \sigma^{\mathcal{D}} \circ s^* \pi$.

Proof. Write $C = C_K^{\gamma}$. The assertion holds because for all $c \in C$ and $f \in C_0(G)$,

$$((\pi * \mathrm{id})(\delta^{\mathcal{C}}(c)))(1 \underset{\mathfrak{b}}{\otimes} f) = (\pi * \mathrm{id})(\delta^{\mathcal{C}}(c)(1 \underset{\mathfrak{b}}{\otimes} f)) = (r^* \pi \circ \sigma^{\mathcal{C}})(c \boxtimes f)$$
$$\delta^{\mathcal{D}}(\pi(c))(1 \underset{\mathfrak{b}}{\otimes} f) = \sigma^{\mathcal{D}}(\pi(c) \boxtimes f) = (\sigma^{\mathcal{D}} \circ s^* \pi)(c \boxtimes f). \quad \blacksquare$$

We denote by \mathbf{G} -act^{*a*} and $\mathbf{Coact}^{a}_{C_{0}(G)}$ the categories of all admissible actions of *G* and all admissible coactions of $C_{0}(G)$, respectively.

THEOREM 6.13. *There exists an embedding as a full and coreflective subcategory* $(\hat{\mathbf{F}}, \hat{\mathbf{G}}, \hat{\eta}, \hat{\epsilon})$ of \mathbf{G} -act^{*a*} into $\mathbf{Coact}^{a}_{Co(G)}$, where

- (i) $\widehat{\mathbf{F}}$ is given by $(C, \sigma) \mapsto (\mathbf{F}C, \delta_{\sigma})$ on objects and $\pi \mapsto \mathbf{F}\pi$ on morphisms;
- (ii) $\widehat{\mathbf{G}}$ is given by $(\mathcal{C}, \delta) \mapsto (\mathbf{G}\mathcal{C}, \sigma_{\delta})$ on objects and $\pi \mapsto \mathbf{G}\pi = \pi$ on morphisms;
- (iii) $\widehat{\eta}_{(C,\sigma)} = \eta_C$ and $\widehat{\varepsilon}_{(C,\delta)} = \varepsilon_C$ for all objects (C,σ) and (C,δ) .

Proof. The assignments $\widehat{\mathbf{G}}$ and $\widehat{\mathbf{F}}$ are well defined on objects and morphisms by Propositions 6.10 and 6.12. For each admissible action (C, σ) , we have $\eta_C \in$ \mathbf{G} -act^{*a*} $((C, \sigma), \widehat{\mathbf{GF}}(C, \sigma))$ because $\sigma_{\delta_{\sigma}}(\eta_C(c) \boxtimes f) = \delta_{\sigma}(\eta_C(c))(1 \bigotimes_{\mathfrak{b}} f) = r^* \eta_C(\sigma(c \boxtimes f))$ f)) for all $c \in C$, $f \in C_0(G)$, and Proposition 6.12 implies that $\varepsilon_C = \eta_{\mathbf{GC}}^{-1} \in$ $\mathbf{Coact}^a_{C_0(G)}(\widehat{\mathbf{FG}}(C, \delta), (C, \delta))$ for each admissible coaction (C, δ) . Now, the assertion follows from Theorem 6.6.

6.3. COMPARISON OF THE ASSOCIATED REDUCED CROSSED PRODUCTS. The reduced crossed product for an action (C, σ) of G is defined as follows [17]. The subspace $C_c(G; C, \sigma) := C_c(G)r^*C \subseteq r^*C$ carries the structure of a *-algebra and the structure of a pre-Hilbert C^* -module over C such that

$$(ab)_{x} = \int_{G^{r(x)}} a_{y} \sigma_{y}(b_{y^{-1}x}) \, d\lambda^{r(x)}(y), \qquad (a^{*})_{x} = \sigma_{x}(a_{x^{-1}}^{*}),$$

$$\langle a|b\rangle_{u} = \int_{G^{u}} \sigma_{y}((a_{y^{-1}})^{*}b_{y^{-1}}) \, d\lambda^{u}(y) = (a^{*}b)_{u}, \qquad (ac)_{x} = a_{x}\sigma_{x}(c_{s(x)})$$

for all $a, b \in C_c(G; C, \sigma)$, $u \in G^0$ and $c \in C$, $x \in G$. Denote the completion of this pre-Hilbert C^* -module by $L^2(G, \lambda^{-1}; C, \sigma)$. Using the relation $\langle a|bd \rangle_u = (abd)_u = \langle b^*a|d \rangle_u$, which holds for all $a, b, d \in C_c(G; C, \sigma)$, $u \in G^0$, and a routine norm estimate, one verifies the existence of a *-homomorphism $\pi : C_c(G; C, \sigma) \rightarrow \mathcal{L}(L^2(G, \lambda^{-1}; C, \sigma))$ such that $\pi(b)d = bd$ for all $b, d \in C_c(G; C, \sigma)$. Then the reduced crossed product of (C, σ) is the C*-algebra $C \rtimes_{\sigma, \mathbf{r}} G := [\pi(C_{\mathbf{c}}(G; C, \sigma))] \subseteq \mathcal{L}(L^2(G, \lambda^{-1}; C, \sigma)).$

PROPOSITION 6.14. Let (C_K^{γ}, δ) be an admissible coaction of $C_0(G)$, consider C as a $C_0(G^0)$ -algebra via ρ_{γ} , and let $\sigma = \sigma_{\delta}$. Then there exists an isomorphism $C \rtimes_{\sigma_{\delta}, \mathbf{r}} G \to C \rtimes_{\mathbf{r}} C_{\mathbf{r}}^*(G)$ given by $\pi(c \boxtimes f) \mapsto \delta(c)(\mathrm{id} \bigotimes UL(f)U)$ for all $c \in C$, $f \in C_c(G)$.

Proof. Let $\delta_U := \operatorname{Ad}_{(\operatorname{id} \otimes U)} \circ \delta : C \to \mathcal{L}(K_{\gamma \otimes_{\widehat{a}}} H)$. We equip $C_c(G; C, \sigma)$ with the structure of a pre-Hilbert C*-module over C such that $(ac)_x = a_x c_{s(x)}$ and $\langle a|b\rangle_u = \int_{G_u} (a_x)^* b_x \, d\lambda_u^{-1}(x)$ for all $a, b \in C_c(G; C, \sigma)$, $c \in C$, $u \in G^0$, and denote

by $L^2(G, \lambda^{-1}; C)$ the completion. One easily checks that there exists a unique unitary $\Phi : L^2(G, \lambda^{-1}; C) \to [|\hat{\alpha}\rangle_2 C] = [\delta_U(C)|\hat{\alpha}\rangle_2\rangle]$ given by $c \boxtimes f \mapsto |\hat{j}(f)\rangle_2 c$, and that for all $c \in C, f \in C_c(G), y \in G$,

$$\Phi^{-1}(\delta_{U}(c)|\hat{j}(f)\rangle_{2})_{y} = \sigma_{y^{-1}}(c_{r(y)})f(y)$$

Hence, there exists a unitary $\Psi : L^2(G, \lambda^{-1}; C, \sigma) \to [\delta_U(C) | \widehat{\alpha} \rangle_2 \rangle]$ given by $c \boxtimes f \mapsto \delta_U(c) | \widehat{j}(f) \rangle_2$. Let $c, d \in C$, $f, g \in C_c(G)$ and $\omega = \Phi^{-1}(\Psi(c \boxtimes f))$. Then $\delta_U(d)(\operatorname{id} \bigotimes L(g))\Psi = \Psi \pi(d \boxtimes g)$ because for all $x \in G$,

$$\begin{split} \Phi^{-1}(\delta_{U}(d)(\mathrm{id} \underset{\mathfrak{b}}{\otimes} L(g))\Phi(\omega))_{x} &= \int_{G_{u}} \sigma_{x^{-1}}(d_{r(x)})g(xy^{-1})\omega_{y} \,\mathrm{d}\lambda_{u}^{-1}(y) \\ &= \int_{G_{u}} \sigma_{x^{-1}}(d_{r(xy^{-1})}g(xy^{-1})\sigma_{xy^{-1}}(c_{r(y)})f(y)) \,\mathrm{d}\lambda_{u}^{-1}(y) \\ &= \Phi^{-1}(\Psi(\pi(d \boxtimes g)(c \boxtimes f)))_{x}. \end{split}$$

Since $d \in C$ and $g \in C_c(G)$ were arbitrary, the assertion follows.

7. FELL BUNDLES ON GROUPOIDS

We now gather preliminaries on Fell bundles that are needed in Sections 8 and 9. We use the notion of a Banach bundle and standard notation; see [8].

7.1. FELL BUNDLES ON GROUPOIDS AND THEIR C*-ALGEBRAS. We first recall the notion of a Fell bundle on *G* and the definition of the associated reduced C*-algebra [15]. Given an upper semicontinuous Banach bundle $p : \mathcal{F} \to G$, denote by \mathcal{F}^0 the restriction of \mathcal{F} to G^0 , by $\mathcal{F}_{sp} \times_{rp} \mathcal{F}$ the restriction of $\mathcal{F} \times \mathcal{F}$ to $G_s \times_r G$, by \mathcal{F}_x for each $x \in G$ the fiber at x, by $\Gamma_c(\mathcal{F})$ the space of continuous sections of \mathcal{F} with compact support, and by $\Gamma_0(\mathcal{F}^0)$ the space of continuous sections of \mathcal{F}^0 that vanish at infinity in norm.

DEFINITION 7.1. A *Fell bundle* on *G* is an upper semicontinuous Banach bundle $p : \mathcal{F} \to G$ with a continuous multiplication $\mathcal{F}_{sp} \times_{rp} \mathcal{F} \to \mathcal{F}$ and a continuous involution $* : \mathcal{F} \to \mathcal{F}$ such that for all $e \in \mathcal{F}$, $(e_1, e_2) \in \mathcal{F}_{sp} \times_{rp} \mathcal{F}$, $(x, y) \in G_s \times_r G$,

(i) $p(e_1e_2) = p(e_1)p(e_2)$ and $p(e^*) = p(e)^{-1}$;

(ii) the map $\mathcal{F}_x \times \mathcal{F}_y \to \mathcal{F}_{xy}, (e'_1, e'_2) \mapsto e'_1 e'_2$, is bilinear and the map $\mathcal{F}_x \to \mathcal{F}_{x^{-1}}, e' \mapsto e'^*$, is conjugate linear;

(iii) $(e_1e_2)e_3 = e_1(e_2e_3)$, $(e_1e_2)^* = e_2^*e_1^*$, and $(e^*)^* = e_2^*e_1^*$

(iv) $||e_1e_2|| \leq ||e_1|| ||e_2||$, $||e^*e|| = ||e||^2$, and $e^*e \geq 0$ in the *C**-algebra $\mathcal{F}_{s(p(e))}$.

We call \mathcal{F} saturated if $[\mathcal{F}_x \mathcal{F}_y] = \mathcal{F}_{xy}$ for all $(x, y) \in G_s \times_r G$, and *admissible* if $\Gamma_0(\mathcal{F}^0)$ is an admissible $C_0(G^0)$ -algebra with respect to the pointwise operations.

Let \mathcal{F} be a Fell bundle on G. The associated reduced C^* -algebra is defined as follows. The space $\Gamma_c(\mathcal{F})$ is a *-algebra with respect to the multiplication and involution given by

(7.1)
$$(cd)(x) = \int_{G^{r(x)}} c(y)d(y^{-1}x) \, \mathrm{d}\lambda^{r(x)}(y) = \int_{G_{s(x)}} c(xz^{-1})d(z) \, \mathrm{d}\lambda^{-1}_{s(x)}(z)$$

and $c^*(x) = c(x^{-1})^*$, respectively, and a pre-Hilbert C^* -module over $\Gamma_0(\mathcal{F}^0)$ with respect to the structure maps

$$\langle c|d\rangle(u) = \int\limits_{G_u} c(x)^* d(x) \, \mathrm{d}\lambda_u^{-1}(x) = (c^*d)(u), \quad (ce)(x) = c(x)e(s(x)),$$

where $c, d \in \Gamma_c(\mathcal{F}), e \in \Gamma_0(\mathcal{F}^0), x \in G$. Denote by $\Gamma^2(\mathcal{F}, \lambda^{-1})$ the completion of this pre-Hilbert *C*^{*}-module. Then there exists a *-homomorphism

$$L_{\mathcal{F}}: \Gamma_{c}(\mathcal{F}) \to \mathcal{L}(\Gamma^{2}(\mathcal{F}, \lambda^{-1})), \quad L_{\mathcal{F}}(a)b = ab \text{ for all } a, b \in \Gamma_{c}(\mathcal{F}),$$

and $C^*_r(\mathcal{F}) := [L_{\mathcal{F}}(\Gamma_c(\mathcal{F}))] \subseteq \mathcal{L}(\Gamma^2(\mathcal{F}, \lambda^{-1}))$ is the *reduced* C^* -algebra of \mathcal{F} . We identify $\Gamma_c(\mathcal{F})$ with $L_{\mathcal{F}}(\Gamma_c(\mathcal{F})) \subseteq C^*_r(\mathcal{F})$ via $L_{\mathcal{F}}$.

We equip $\Gamma_{c}(\mathcal{F})$ with the *inductive limit topology*; thus, a net converges if it converges uniformly and if the supports of its members are contained in some compact set. We shall use the following result; see Proposition 2.3 in [8].

LEMMA 7.2. Let \mathcal{E} be an upper semicontinuous Banach bundle on a locally compact, second countable, Hausdorff space X and let $\Gamma' \subseteq \Gamma_{c}(\mathcal{E})$ be a subspace such that

- (i) Γ' is closed under pointwise multiplication with elements of $C_{c}(X)$;
- (ii) $\{f(x) : f \in \Gamma'\} \subseteq \mathcal{E}_x$ is dense for each $x \in X$. Then Γ' is dense in $\Gamma_c(\mathcal{E})$.

Given $f \in \Gamma_c(\mathcal{F})$ and $g \in \Gamma_0(\mathcal{F}^0)$, define $fg,gf \in \Gamma_c(\mathcal{F})$ by (fg)(x) = f(x)g(s(x)), (gf)(x) = g(r(x))f(x) for all $x \in G$. Using the relation $[\mathcal{F}_x] = [\mathcal{F}_x \mathcal{F}_x^* \mathcal{F}_x]$, where $x \in G$, and Lemma 7.2, we find:

LEMMA 7.3. $\Gamma_{c}(\mathcal{F})\Gamma_{0}(\mathcal{F}^{0})$ and $\Gamma_{0}(\mathcal{F}^{0})\Gamma_{c}(\mathcal{F})$ are linearly dense in $\Gamma_{c}(\mathcal{F})$.

7.2. THE MULTIPLIER BUNDLE OF A FELL BUNDLE. Given a Fell bundle \mathcal{F} on *G*, we define a multiplier bundle $\mathcal{M}(\mathcal{F})$ on *G*, extending the definition in Section VIII.2.14 of [12]. Given a subspace $C \subseteq G$, we denote by $\mathcal{F}|_C$ the restriction of \mathcal{F} to *C*.

DEFINITION 7.4. Let $x \in G$. A *multiplier of* \mathcal{F} *of order* x is a map $T : \mathcal{F}|_{G^{s(x)}} \to \mathcal{F}|_{G^{r(x)}}$ such that $T\mathcal{F}_y \subseteq \mathcal{F}_{xy}$ for all $y \in G^{s(x)}$ and such that there exists a map $T^* : \mathcal{F}|_{G^{r(x)}} \to \mathcal{F}|_{G^{s(x)}}$ such that $e^*Tf = (T^*e)^*f$ for all $e \in \mathcal{F}|_{G^{r(x)}}$, $f \in \mathcal{F}|_{G^{s(x)}}$. We denote by $\mathcal{M}(\mathcal{F})_x$ the set of all multipliers of \mathcal{F} of order x.

As for adjointable operators of Hilbert C^* -modules, one deduces from the definition the following simple properties. Let $x \in G$. Then for each $T \in \mathcal{M}(\mathcal{F})_x$, the map T^* is uniquely determined, $T^* \in \mathcal{M}(\mathcal{F})_{x^{-1}}$, and $T^{**} = T$. Moreover, each $T \in \mathcal{M}(\mathcal{F}_x)$ is fiberwise linear in the sense that $T(\kappa e + f) = \kappa Te + Tf$ for all $\kappa \in \mathbb{C}$, $e, f \in \mathcal{F}_y, y \in G^{s(x)}$. The restrictions $T_{s(x)} : \mathcal{F}_{s(x)} \to \mathcal{F}_x$ and $(T^*)_x : \mathcal{F}_x \to \mathcal{F}_{s(x)}$ are adjoint operators of Hilbert C^* -modules over $\mathcal{F}_{s(x)}$, and since $\mathcal{F}_y = [\mathcal{F}_{r(y)}\mathcal{F}_y]$ for each $y \in G^{s(x)}$, the map $\mathcal{M}(\mathcal{F})_x \to \mathcal{L}(\mathcal{F}_{s(x)}, \mathcal{F}_x), T \mapsto T_{s(x)}$, is a bijection. Clearly, we have a natural embedding $\mathcal{F}_x \hookrightarrow \mathcal{M}(\mathcal{F})_x$, where each $f \in \mathcal{F}$ acts as a multiplier via left multiplication. For each $y \in G^{s(x)}$, we have $\mathcal{M}(\mathcal{F})_x \mathcal{M}(\mathcal{F})_y \subseteq \mathcal{M}(\mathcal{F})_{xy}$, and for each $f \in \mathcal{F}_z, z \in G_{r(x)}$, we let $fT := (T^*f^*)^*$.

DEFINITION 7.5. For each $x \in G$, consider $\mathcal{M}(\mathcal{F})_x$ as a Banach space via the identification with $\mathcal{L}(\mathcal{F}_{s(x)}, \mathcal{F}_x)$. Let $\mathcal{M}(\mathcal{F}) = \coprod_{x \in G} \mathcal{M}(\mathcal{F})_x$ and denote by $\tilde{p} : \mathcal{M}(\mathcal{F}) \to G$ the natural map. The *strict topology* on $\mathcal{M}(\mathcal{F})$ is the weakest topology that makes \tilde{p} and the maps $\mathcal{M}(\mathcal{F}) \to \mathcal{F}$ of the form $c \mapsto c \cdot d(s(\tilde{p}(c)))$ and $c \mapsto d(r(\tilde{p}(c))) \cdot c$ continuous for each $d \in \Gamma_c(\mathcal{F}^0)$. Denote by $\Gamma_c(\mathcal{M}(\mathcal{F}))$ the space of all sections that are strictly continuous, norm-bounded, and compactly supported.

REMARK 7.6. The bundle $\mathcal{M}(\mathcal{F})$ satisfies all axioms of a Fell bundle except for the fact that it is no Banach bundle with respect to the strict topology unless $\mathcal{M}(\mathcal{F}) = \mathcal{F}$. Indeed, for each $u \in G^0$, the subspace topology on $\mathcal{M}(\mathcal{F})_u \cong$ $\mathcal{L}(\mathcal{F}_u) \cong \mathcal{M}(\mathcal{F}_u)$ is the strict topology and coincides with the norm topology only if $\mathcal{M}(\mathcal{F}_u) = \mathcal{F}_u$.

Given $f \in \Gamma_{c}(\mathcal{M}(\mathcal{F}))$ and $g \in \Gamma_{0}(\mathcal{F}^{0})$, define $fg, gf \in \Gamma_{c}(\mathcal{F})$ by (fg)(x) = f(x)g(s(x)), (gf)(x) = g(r(x))f(x) for all $x \in G$ again.

LEMMA 7.7. (i) Let $c \in \Gamma_c(\mathcal{M}(\mathcal{F}))$ and $d \in \Gamma_c(\mathcal{F})$. Then there exists a section $cd \in \Gamma_c(\mathcal{F})$ such that $(cd)(x) = \int_{G^{r(x)}} c(y)d(y^{-1}x) d\lambda^{r(x)}(y)$ for all $x \in G$.

(ii) $\Gamma_{c}(\mathcal{M}(\mathcal{F}))$ carries a structure of a *-algebra such that $c^{*}(x) = c(x^{-1})^{*}$ and $(cd)(x)e = \int_{G^{r(x)}} c(y)d(y^{-1}x)e \, d\lambda^{r(x)}$ for all $c, d \in \Gamma_{c}(\mathcal{M}(\mathcal{F})), x \in G, e \in \mathcal{F}_{s(x)}$.

(iii) There exists a *-homomorphism $L_{\mathcal{M}(\mathcal{F})} : \Gamma_{c}(\mathcal{M}(\mathcal{F})) \to M(C_{r}^{*}(\mathcal{F}))$ such that $L_{\mathcal{M}(\mathcal{F})}(c)L_{\mathcal{F}}(d) = L_{\mathcal{F}}(cd)$ for all $c \in \Gamma_{c}(\mathcal{M}(\mathcal{F})), d \in \Gamma_{c}(\mathcal{F})$.

(iv) $\Gamma_{c}(\mathcal{M}(\mathcal{F}))$ is closed under pointwise multiplication with elements of $C_{c}(G)$.

Proof. (i) Define $cd : G \to \mathcal{F}$ as above, and let $\varepsilon > 0$. Using Lemma 7.3, we find a sequence $(g_n)_n$ in the span of $\Gamma_0(\mathcal{F}^0)\Gamma_c(\mathcal{F})$ that converges to d in the inductive limit topology. Since $\Gamma_c(\mathcal{M}(\mathcal{F}))\Gamma_0(\mathcal{F}^0) \subseteq \Gamma_c(\mathcal{F})$, the map $h_n : x \mapsto \int_{G^{r(x)}} c(y)g_n(y^{-1}x) d\lambda^{r(x)}(y)$ lies in $\Gamma_c(\mathcal{F})$ for each n. Using the fact that c has com-

pact support and bounded norm, one easily concludes that $(h_n)_n$ converges in the inductive limit topology to *cd* which therefore is in $\Gamma_c(\mathcal{F})$.

(ii) Note that (cd)(x) is well defined because the map $y \mapsto d(y^{-1}x)e$ is in $\Gamma_{c}(\mathcal{F})$ and thus (i) applies. Now, the assertion follows from standard arguments.

(iii) One easily sees that there exists a representation $L_{\mathcal{M}(\mathcal{F})} : \Gamma_{c}(\mathcal{M}(\mathcal{F})) \rightarrow \mathcal{L}(\Gamma^{2}(\mathcal{F}))$ such that $L_{\mathcal{M}(\mathcal{F})}(c)d = cd$ for all $c \in \Gamma_{c}(\mathcal{M}(\mathcal{F})), d \in \Gamma_{c}(\mathcal{F})$, and that $L_{\mathcal{M}(\mathcal{F})}(c)L_{\mathcal{F}}(d)e = cde = L_{\mathcal{F}}(cd)e$ for all $c \in \Gamma_{c}(\mathcal{M}(\mathcal{F})), d, e \in \Gamma_{c}(\mathcal{F})$.

(iv) This follows immediately from the fact that $\Gamma_c(\mathcal{F})$ is closed under pointwise multiplication by elements of $C_c(G)$.

7.3. MORPHISMS BETWEEN FELL BUNDLES. Let \mathcal{F} and \mathcal{G} be Fell bundles on G.

DEFINITION 7.8. A (*fibrewise nondegenerate*) *morphism* from \mathcal{F} to \mathcal{G} is a continuous map $T : \mathcal{F} \to \mathcal{M}(\mathcal{G})$ that satisfies the following conditions:

(i) for each $x \in G$, the map *T* restricts to a linear map $T_x : \mathcal{F}_x \to \mathcal{M}(\mathcal{G})_x$;

(ii) $T(e_1)T(e_2) = T(e_1e_2)$ and $T(e)^* = T(e^*)$ for all $(e_1, e_2) \in \mathcal{F}_{sp} \times_{rp} \mathcal{F}, e \in \mathcal{F}$; (iii) $\mathcal{G}_x = [T(\mathcal{F}_x)\mathcal{G}_{s(x)}]$ for each $x \in G^0$.

Let *T* be a morphism from \mathcal{F} to \mathcal{G} . Then $T_u : \mathcal{F}_u \to \mathcal{M}(\mathcal{G})_u$ is a nondegenerate *-homomorphism for each $u \in G^0$; in particular, $||T_u|| \leq 1$. One easily concludes that $||T_x|| \leq 1$ for each $x \in G$. Hence, the formula $f \mapsto T \circ f$ defines *-homomorphisms $T_* : \Gamma_c(\mathcal{F}) \to \Gamma_c(\mathcal{M}(\mathcal{G}))$ and $T^*_0 : \Gamma_0(\mathcal{F}^0) \to \mathcal{M}(\Gamma_0(\mathcal{G}^0))$.

PROPOSITION 7.9. (i) $T^0_* : \Gamma_0(\mathcal{F}^0) \to M(\Gamma_0(\mathcal{G}^0))$ is nondegenerate. (ii) $T_*(\Gamma_c(\mathcal{F}))\Gamma_c(\mathcal{G}^0)$ is dense in $\Gamma_c(\mathcal{F})$.

(iii) T_* extends to a nondegenerate *-homomorphism $T_* : C^*_r(\mathcal{F}) \to M(C^*_r(\mathcal{G})).$

Proof. Assertions (i) and (ii) follow immediately from Lemmas 7.2 and 7.7. Part (ii) and a straightforward calculation show that there exists a unique unitary $\Psi : \Gamma^2(\mathcal{F}, \lambda^{-1}) \otimes_{T^0_*} \Gamma_0(\mathcal{G}^0) \to \Gamma^2(\mathcal{G}, \lambda^{-1})$ such that $(\Psi(f \otimes g))(x) = T_*(f)g$ for all $f \in \Gamma_c(\mathcal{F}), g \in \Gamma_0(\mathcal{G}^0)$. The map $C^*_r(\mathcal{F}) \to \mathcal{L}(\Gamma^2(\mathcal{G}, \lambda^{-1}))$ given by $f \mapsto \Psi(f \otimes id)\Psi^*$ is the desired extension. Lemma 7.3 and part (ii) imply that $[T_*(\Gamma_c(\mathcal{F}))\Gamma_c(\mathcal{G})] = [T_*(\Gamma_c(\mathcal{F}))\Gamma_0(\mathcal{G}^0)\Gamma_c(\mathcal{G})] = [\Gamma_c(\mathcal{G})\Gamma_c(\mathcal{G})] = C^*_r(\mathcal{G})$.

8. FROM FELL BUNDLES ON G TO COACTIONS OF $C_r^*(G)$

Let *G* be a groupoid, *V* the associated *C*^{*}-pseudo-multiplicative unitary, and $C_r^*(G)$ or, more precisely, (\mathcal{A}, Δ) the associated Hopf *C*^{*}-bimodule as in Section 5. We relate Fell bundles on *G* to coactions of $C_r^*(G)$ as follows. Let \mathcal{F} be an admissible Fell bundle \mathcal{F} on *G*. We shall construct a coaction of $C_r^*(G)$ on $C_r^*(\mathcal{F})$ which is unitarily implemented by a representation of *V*, and identify the reduced crossed product of this coaction with the reduced *C*^{*}-algebra of another Fell bundle. Finally, we show that this construction is functorial.

A representation of the unitary *V* is a C^{*}-($\mathfrak{b}, \mathfrak{b}^{\dagger}$)-module ${}_{\gamma}K_{\widehat{\delta}}$ together with a unitary $X: K_{\widehat{\delta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} H \to K_{\gamma} \underset{\mathfrak{b}}{\otimes} H$ that satisfies $X(\gamma \triangleleft \alpha) = \gamma \triangleright \alpha, X(\widehat{\delta} \triangleright \beta) = \widehat{\delta} \triangleleft \beta, X(\widehat{\delta} \triangleright \widehat{\beta}) = \gamma \triangleright \widehat{\beta}$, and $X_{12}X_{13}V_{23} = V_{23}X_{12}$; see Section 4 in [29]. We construct a coaction out of such a representation as follows.

LEMMA 8.1. Let $({}_{\gamma}K_{\hat{\delta}}, X)$ be a representation of V, let C_{K}^{γ} be a C^* -b-algebra such that $[C, \rho_{\hat{\beta}}(\mathfrak{B})] = 0$, define $\delta : C \to \mathcal{L}(K_{\gamma} \underset{\mathfrak{b}}{\otimes}_{\beta} H)$ by $c \mapsto X(c \underset{\mathfrak{b}^{\dagger}}{\otimes} id)X^*$, and assume that $[\delta(C)|\gamma\rangle_1 A] \subseteq [|\gamma\rangle_1 A]$ and $[\delta(C)|\beta\rangle_2] \subseteq [|\beta\rangle_2 C]$. Then δ is injective, a morphism from (K_{γ}, C) to $(K_{\gamma} \underset{\mathfrak{b}}{\otimes}_{\beta} H_{\alpha}, C_{\gamma} \underset{\mathfrak{b}}{\ast}^{\beta} A)$, and a coaction of (\mathcal{A}, Δ) on C_{K}^{γ} . If the inclusions above are equalities, then δ is left- or right-full, respectively.

Proof. Evidently, δ is injective. It is a morphism of C^* - \mathfrak{b} -algebras because $X|\xi\rangle_2 c = \delta(c)X|\xi\rangle_2$ for each $\xi \in \alpha$, $c \in C$ and because $[X|\alpha\rangle_2\gamma] = \gamma \triangleright \alpha$ and $[(X|\alpha\rangle_2)^*(\gamma \triangleright \alpha)] = [\langle \alpha|_2(\gamma \triangleleft \alpha)] = \gamma$. Finally, for each $c \in C$,

$$\begin{aligned} (\delta * \mathrm{id})(\delta(c)) &= X_{12}X_{13}c_1X_{13}^*X_{12}^* = X_{12}X_{13}V_{23}c_1V_{23}^*X_{13}^*X_{12}^* \\ &= V_{23}X_{12}c_1X_{12}^*V_{23}^* = (\mathrm{id}*\Delta)(\delta(c)), \end{aligned}$$

where c_1 denotes c acting on the first factor of an iterated relative tensor product.

8.1. THE REPRESENTATION OF *V* ASSOCIATED TO \mathcal{F} . Denote by $\mathcal{W} = \mathcal{W}(\Gamma_0(\mathcal{F}^0))$ the set of all $C_0(G^0)$ -weights on $\Gamma_0(\mathcal{F}^0)$ and let $\phi \in \mathcal{W}$.

LEMMA 8.2. Let $c, d \in \Gamma_c(\mathcal{F})$. Then the map $x \mapsto \phi_{s(x)}(c(x)^*d(x))$ lies in $C_c(G)$.

Proof. The function $G \to s^* \mathcal{F}^0$ given by $x \mapsto c(x)^* d(x)$ is continuous and has compact support, and the composition $h : x \mapsto \phi_{s(x)}(c(x)^* d(x))$ is continuous because the map $\mathcal{F}^0 \to \mathbb{C}$ given by $f \mapsto \phi_{p(f)}(f)$ is continuous.

Define Hilbert C^* - $C_0(G^0)$ -modules $\Gamma^2(\mathcal{F}, \lambda; \phi)$, $\Gamma^2(\mathcal{F}, \lambda^{-1}; \phi)$ and a Hilbert space $K_{\phi} = \Gamma^2(\mathcal{F}, \nu; \phi)$ as the respective completions of $\Gamma_c(\mathcal{F})$, where for all $c, d \in \mathcal{F}$

 $\Gamma_{c}(\mathcal{F}), f \in C_{0}(G^{0})$, the inner product $\langle c | d \rangle$ and the product cf are given by

$$\begin{split} u &\mapsto \int_{G^{u}} \phi_{s(x)}(c(x)^{*}d(x)) \, \mathrm{d}\lambda^{u}(x), \quad y \mapsto c(y)f(r(y)) \quad \text{in case of } \Gamma^{2}(\mathcal{F},\lambda;\phi), \\ u &\mapsto \int_{G_{u}} \phi_{s(x)}(c(x)^{*}d(x)) \, \mathrm{d}\lambda_{u}^{-1}(x), \quad y \mapsto c(y)f(s(y)) \quad \text{in case of } \Gamma^{2}(\mathcal{F},\lambda^{-1};\phi), \\ \text{and } \int_{G} \phi_{s(x)}(c(x)^{*}d(x)) \, \mathrm{d}\nu(x) & \text{in case of } \Gamma^{2}(\mathcal{F},\nu;\phi). \end{split}$$

LEMMA 8.3. $[\langle E|E\rangle] = [\phi(\Gamma_0(\mathcal{F}^0))]$ for $E \in \{\Gamma^2(\mathcal{F},\lambda,\phi),\Gamma^2(\mathcal{F},\lambda^{-1};\phi)\}.$

Proof. Assume that $(\phi(c^*c))(u) \neq 0$ for some $c \in \Gamma_c(\mathcal{F}^0)$, $u \in G^0$. Choose $d \in \Gamma_c(\mathcal{F})$ such that $d|_{G^0} = c$. Then the function on G given by $x \mapsto \phi_{s(x)}(d(x)^*d(x))$ is non-negative and nonzero at u, whence $\langle d|d\rangle_E(u) \neq 0$. Now, the assertion follows because $[\langle E|E\rangle]$ and $[\phi(\Gamma_0(\mathcal{F}^0))]$ are closed ideals in $C_0(G^0)$.

Let $K = \bigoplus_{\phi \in \mathcal{W}} K_{\phi}$ and identify each K_{ϕ} with a subspace of K. Given $c \in \Gamma_{c}(\mathcal{F})$ and $f \in C_{0}(G^{0})$, define $fc, cf, cD^{-1/2} \in \Gamma_{c}(\mathcal{F})$ by

$$fc: x \mapsto f(r(x))c(x), \quad cf: x \mapsto c(x)f(s(x)), \quad cD^{-1/2}: x \mapsto c(x)D^{-1/2}(x).$$

Let $\phi \in \mathcal{W}$. Straightforward calculations show that there exist maps

$$j_{\phi}: \Gamma^2(\mathcal{F}, \lambda; \phi) \to \mathcal{L}(\mathfrak{K}, K_{\phi}) \quad \text{and} \quad \hat{j}_{\phi}: \Gamma^2(\mathcal{F}, \lambda^{-1}; \phi) \to \mathcal{L}(\mathfrak{K}, K_{\phi})$$

such that $j_{\phi}(c)f = fc$ and $\hat{j}_{\phi}(c)f = (cD^{-1/2})f$ for all $c \in \Gamma_{c}(\mathcal{F}), f \in C_{c}(G^{0})$, and $j_{\phi}(c)^{*}j_{\phi}(d) = \langle c|d \rangle_{\Gamma^{2}(\mathcal{F},\lambda;\phi)}, \quad \hat{j}_{\phi}(c)^{*}\hat{j}_{\phi}(d) = \langle c|d \rangle_{\Gamma^{2}(\mathcal{F},\lambda^{-1};\phi)} \quad \text{for all } c, d \in \Gamma_{c}(\mathcal{F}).$

Denote by $\gamma \subseteq \mathcal{L}(\mathfrak{K}, K)$ and $\widehat{\delta} \subseteq \mathcal{L}(\mathfrak{K}, K)$ the closed linear span of all subspaces $j_{\phi}(\Gamma^2(\mathcal{F}, \lambda; \phi))$ and $\widehat{j}_{\phi}(\Gamma^2(\mathcal{F}, \lambda^{-1}; \phi))$, respectively, where $\phi \in \mathcal{W}$. Lemmas 6.4 and 8.3 imply:

LEMMA 8.4. $_{\gamma}K_{\hat{\delta}}$ is a C^* -($\mathfrak{b}, \mathfrak{b}^+$)-module, and for all $f \in C_0(G^0)$ and $(c_{\phi})_{\phi} \in \bigoplus_{\phi} \Gamma_c(\mathcal{F}) \subseteq K$, we have $\rho_{\gamma}(f)(c_{\phi})_{\phi} = (fc_{\phi})_{\phi}$ and $\rho_{\hat{\delta}}(f)(c_{\phi})_{\phi} = (c_{\phi}f)_{\phi}$.

For t = s, r, denote by $p_1^{t,r} : G_t \times_r G \to G$ the projection onto the first component, by $\mathcal{F}_{t,r}^2 = (p_1^{t,r})^* \mathcal{F}$ the corresponding pull-back of \mathcal{F} , and by $\Gamma^2(\mathcal{F}_{t,r}^2, \nu_{t,r}^2; \phi)$ the Hilbert space that is the completion of $\Gamma_c(\mathcal{F}_{t,r}^2)$ with respect to the inner product

$$\langle c|d\rangle = \int\limits_{G_t \times_{\mathbf{r}} G} \phi_{s(x)}(c(x,y)^* d(x,y)) \, \mathrm{d}\nu_{t,r}^2(x,y).$$

Straightforward calculations show that there exist unitaries

$$\Phi: K_{\widehat{\delta}} \underset{\mathfrak{b}^{\dagger}}{\otimes} H \to \bigoplus_{\phi \in \mathcal{W}} \Gamma^{2}(\mathcal{F}^{2}_{s,r}, v_{s,r}^{2}; \phi), \quad \Psi: K_{\gamma} \underset{\mathfrak{b}}{\otimes} H \to \bigoplus_{\phi \in \mathcal{W}} \Gamma^{2}(\mathcal{F}^{2}_{r,r}, v_{r,r}^{2}; \phi),$$

such that for all $\phi \in W$, $c \in \Gamma_c(\mathcal{F})$, $f \in C_c(G^0)$, $g \in C_c(G)$,

$$\begin{split} &\Phi(\widehat{j}_{\phi}(c) \otimes f \otimes j(g)) \in \Gamma^{2}(\mathcal{F}^{2}_{s,r}; \nu^{2}_{s,r}; \phi) \text{ is given by } (x,y) \mapsto ((cD^{-1/2})f)(x)g(y), \\ &\Psi(j_{\phi}(c) \otimes f \otimes j(g)) \in \Gamma^{2}(\mathcal{F}^{2}_{r,r}; \nu^{2}_{r,r}; \phi) \text{ is given by } (x,y) \mapsto (fc)(x)g(y). \end{split}$$

We shall use the isomorphisms above without further notice. If $(T_{\phi})_{\phi}$ is a normbounded family of operators between Hilbert spaces $(H^1_{\phi})_{\phi}$ and $(H^2_{\phi})_{\phi}$, we denote by $\bigoplus_{\phi} T_{\phi} \in \mathcal{L}(\bigoplus_{\phi} H^1_{\phi}, \bigoplus_{\phi} H^2_{\phi})$ the operator given by $(\xi_{\phi})_{\phi} \mapsto (T_{\phi}\xi_{\phi})_{\phi}$. Similar arguments as those used for the construction of *V* in Theorem 2.5 in [27] show:

PROPOSITION 8.5. If $\phi \in W$, then exists a unitary $X_{\phi} : \Gamma^2(\mathcal{F}^2_{s,r}, v^2_{s,r}; \phi) \rightarrow \Gamma^2(\mathcal{F}^2_{r,r}, v^2_{r,r}; \phi)$ such that $(X_{\phi}f)(x, y) = f(x, x^{-1}y)$ for all $f \in \Gamma_c(\mathcal{F}^2_{s,r}), (x, y) \in G_r \times_r G$, and the pair $({}_{\gamma}K_{\widehat{\delta}}, \bigoplus_{\phi} X_{\phi})$ is a representation of V.

8.2. THE COACTION OF $C_r^*(G)$ ON $C_r^*(\mathcal{F})$. We apply Lemma 8.1 to the representation $({}_{\gamma}K_{\widehat{\lambda}'}X)$ and obtain a coaction of $C_r^*(G)$ on $C_r^*(\mathcal{F})$ as follows.

LEMMA 8.6. Let $\phi \in W$. There exists a representation $\pi_{\phi} : C_{r}^{*}(\mathcal{F}) \to \mathcal{L}(K_{\phi})$ such that for all $c, d \in \Gamma_{c}(\mathcal{F}), x \in G$,

$$(\pi_{\phi}(c)d)(x) = \int_{G^{r(x)}} c(z)d(z^{-1}x)D^{-1/2}(z)\,\mathrm{d}\lambda^{r(x)}(z)$$

and $\pi_{\phi}(c)\widehat{j}_{\phi}(d) = \widehat{j}_{\phi}(cd)$ and $\pi_{\phi}(c)\rho_{\gamma}(f) = \pi_{\phi}(cf)$ for all $c, d \in \Gamma_{c}(\mathcal{F}), f \in C_{0}(G^{0}).$

Proof. Identify $\Gamma^2(\mathcal{F}, \lambda^{-1}) \otimes_{\phi} L^2(G^0, \mu)$ with K_{ϕ} via $c \otimes f \mapsto \hat{j}_{\phi}(c)f$ for all $c \in \Gamma_c(\mathcal{F}), f \in C_c(G^0)$, and define π_{ϕ} by $c \mapsto c \otimes_{\phi} id$.

Define
$$\pi : C_r^*(\mathcal{F}) \to \mathcal{L}(K)$$
 by $c \mapsto \bigoplus_{\phi} \pi_{\phi}(c)$. Lemmas 6.4 and 8.6 imply:

LEMMA 8.7. The representation π is faithful, $\pi(C_r^*(\mathcal{F}))_K^{\gamma}$ is a C*-b-algebra, and $[\pi(C_r^*(\mathcal{F}))\hat{\delta}] = \hat{\delta}$.

Define $\delta : \pi(C^*_{\mathbf{r}}(\mathcal{F})) \to \mathcal{L}(K_{\gamma \underset{\mathfrak{b}}{\otimes} \beta}H)$ by $\pi(c) \mapsto X(\pi(c) \underset{\mathfrak{b}^{\dagger}}{\otimes} \operatorname{id})X^*$. Let $c \in C^*_{\mathbf{r}}(\mathcal{F})$. Then $\delta(\pi(c)) = \bigoplus_{\phi} \delta(\pi(c))_{\phi}$ and each $\delta(\pi(c))_{\phi} \in \mathcal{L}(\Gamma^2(\mathcal{F}^2_{r,r}, v_{r,r}^2; \phi))$ acts as follows.

LEMMA 8.8. For all
$$c \in \Gamma_{c}(\mathcal{F})$$
, $\phi \in \mathcal{W}$, $d \in \Gamma_{c}(\mathcal{F}_{r,r}^{2})$, $(x,y) \in G_{r} \times_{r} G$,
 $(\delta(\pi(c))_{\phi}d)(x,y) = \int_{G^{r(x)}} c(z)d(z^{-1}x,z^{-1}y)D^{-1/2}(z) d\lambda^{r(x)}(z).$

Proof. The verification is straightforward and similar to the calculation of the comultiplication Δ on $C_r^*(G)$; see Section 3.4 in [27].

THEOREM 8.9. $(\pi(C^*_r(\mathcal{F}))^{\gamma}_K, \delta)$ is a very fine and left-full coaction of $C^*_r(G)$.

The proof involves the following two lemmas.

LEMMA 8.10. Let $\phi \in W$. Then there exist maps

 $T_{\phi}: \Gamma_{c}(\mathcal{F}^{2}_{r,r}) \to \mathcal{L}(K_{\phi}, \Gamma^{2}(\mathcal{F}^{2}_{r,r}, \nu^{2}_{r,r}; \phi)), \quad S_{\phi}: \Gamma_{c}(\mathcal{F}^{2}_{r,r}) \to \mathcal{L}(H, \Gamma^{2}(\mathcal{F}^{2}_{r,r}, \nu^{2}_{r,r}; \phi))$

that are continuous with respect to the inductive topology on $\Gamma_{c}(\mathcal{F}_{r,r}^{2})$ and the operator norm, respectively, such that for all $c \in \Gamma_{c}(\mathcal{F}_{r,r}^{2})$, $d \in \Gamma_{c}(\mathcal{F})$, $f \in C_{c}(G)$, $(x, y) \in G_{r} \times_{r} G$,

$$(T_{\phi}(c)d)(x,y) = \int_{G^{r(x)}} c(z,y)d(z^{-1}x)D^{-1/2}(z) \,d\lambda^{r(x)}(z),$$

$$(S_{\phi}(c)f)(x,y) = \int_{G^{r(y)}} c(x,z)f(z^{-1}y)D^{-1/2}(z) \,d\lambda^{r(y)}(z).$$

Proof. Let *c*, *d*, $T_{\phi}(c)d$ as above. Then

$$\begin{aligned} \|T_{\phi}(c)d\|^{2} &= \int_{G} \int_{G^{r(x)}} \int_{G^{r(x)}} \int_{G^{r(x)}} \phi_{s(x)}(d(z_{1}^{-1}x)^{*}c(z_{1},y)^{*}c(z_{2},y)d(z_{2}^{-1}x)) \\ &\cdot D^{-1/2}(z_{1})D^{-1/2}(z_{2}) \, \mathrm{d}\lambda^{r(x)}(y) \, \mathrm{d}\lambda^{r(x)}(z_{1}) \, \mathrm{d}\lambda^{r(x)}(z_{2}) \int \mathrm{d}\nu(x) \end{aligned}$$

We substitute $x' = z_1^{-1}x$, $z = z_1^{-1}z_2$, use the relations $D(z_2) = D(z_1)D(z)$ and

$$\begin{aligned} D^{-1}(z_1) \, \mathrm{d}\lambda^{r(x)}(z_1) \int \mathrm{d}\nu(x) &= D^{-1}(z_1) \, \mathrm{d}\lambda^{r(z_1)}(x) \int \mathrm{d}\nu(z_1) \\ &= \, \mathrm{d}\lambda^{s(z_1)}(x') \int \mathrm{d}\nu^{-1}(z_1) = \, \mathrm{d}\lambda^{-1}_{r(x')}(z_1) \int \mathrm{d}\nu(x'), \end{aligned}$$

and find

$$\begin{split} \|T_{\phi}(c)d &= \int_{G} \int_{G_{r(x')}} \int_{G^{r(x')}} \int_{G^{s(z_1)}} \phi_{s(x)}(d(x')^* c(z_1, y)^* c(z_1 z, y) d(z^{-1} x')) \\ &\quad \cdot D^{-1/2}(z) \, d\lambda^{s(z_1)}(y) \, d\lambda^{r(x')}(z) \, d\lambda^{-1}_{r(x')}(z_1) \int d\nu(x') \\ &= \int_{G} \int_{G^{r(x')}} \phi_{s(x')}(d(x') R_{c}(z) d(z^{-1} x')) \, d\lambda^{r(x')}(z) \int d\nu(x') = \langle d | \pi_{\phi}(R_{c}) d \rangle_{K_{\phi}}, \end{split}$$

where $R_c \in \Gamma_c(\mathcal{F})$ is given by

$$R_{c}(z) = \int_{G_{r(z)}} \int_{G^{s(z_{1})}} c(z_{1}, y)^{*} c(z_{1}z, y) \, \mathrm{d}\lambda^{s(z_{1})}(y) \, \mathrm{d}\lambda^{-1}_{r(z)}(z_{1}) \quad \text{for all } z \in G.$$

Hence, $T_{\phi}(c)$ extends to a bounded linear operator of norm $||T_{\phi}(c)||^2 \leq ||\pi_{\phi}(R_c)||$. If $(c_n)_n$ is a sequence in $\Gamma_c(\mathcal{F}^2_{r,r})$ converging to c in the inductive limit topology, then the functions $R_{(c-c_n)}$ defined similarly as R_c converge to 0 in the inductive limit topology and hence $||T_{\phi}(c-c_n)||^2 \leq ||\pi_{\phi}(R_{(c-c_n)})||$ converges to 0.

The proof of the assertion concerning S_{ϕ} is very similar.

Given
$$c, d \in \Gamma_{c}(\mathcal{F})$$
 and $f \in C_{c}(G)$, define $\omega_{c,d,f} \in \Gamma_{c}(\mathcal{F}_{r,r}^{2})$ by
 $(x,y) \mapsto \int_{G^{r(x)}} c(z)d(z^{-1}x)f(z^{-1}y) \, d\lambda^{r(x)}(z).$

LEMMA 8.11. The linear span of all elements $\omega_{c,d,f}$ as above is dense in $\Gamma_c(\mathcal{F}_{r,r}^2)$ with respect to the inductive limit topology.

Proof. Let $(x, y) \in G_r \times_r G$, $e \in \mathcal{F}_x$, let $C \subseteq G_r \times_r G$ be a compact neighbourhood of (x, y), and let $\varepsilon > 0$. Since $[\mathcal{F}_{r(x)}\mathcal{F}_x] = \mathcal{F}_x$, we can choose $c', d' \in \Gamma_c(\mathcal{F})$ such that $||c'(z)d'(z^{-1}x) - e|| < \varepsilon$ for all z in some neighbourhood of r(x) in $G^{r(x)}$. Next, we can choose $h_c, h_d, f \in C_c(G)$ such that the elements $c, d \in \Gamma_c(\mathcal{F})$ given by $c(z) = c'(z)h_c(z)$ and $d(z) = d'(z)h_d(z)$ for all $z \in G$ satisfy $||\omega_{c,d,f}(x,y) - e|| < \varepsilon$ and supp $\omega_{c,d,f} \subseteq C$. A standard partition of unity argument concludes the proof.

Proof of Theorem 8.9. We show that Lemma 8.1 applies. Let $\phi \in W$, $c, d \in \Gamma_c(\mathcal{F})$, $f, g \in C_c(G)$. Define $e_1, e_2, e_3, e_4 \in \Gamma^2(\mathcal{F}^2_{r,r}, \nu^2_{r,r}; \phi)$ and $\omega_1, \omega_2, \omega_3, \omega_4 \in \Gamma_c(\mathcal{F}^2_{r,r})$ by

$$\begin{aligned} e_1 &= \delta(\pi(c))_{\phi} |j(f)\rangle_2 d, & \omega_1(z,y) = c(z)f(z^{-1}y) \text{ for all } (z,y) \in G_r \times_r G, \\ e_2 &= |j(f)\rangle_2 \pi_{\phi}(c) d, & \omega_2(z,y) = c(z)f(y) \text{ for all } (z,y) \in G_r \times_r G, \\ e_3 &= |j_{\phi}(c)\rangle_1 L(f)g, & \omega_3(x,z) = c(x)f(z) \text{ for all } (x,z) \in G_r \times_r G, \\ e_4 &= \delta(\pi(c))_{\phi} |j_{\phi}(d)\rangle_1 L(f)g, & \omega_4 = \omega_{c,d,f}. \end{aligned}$$

Using Lemma 8.8, we find that for all $(x, y) \in G_r \times_r G$,

$$\begin{split} e_{1}(x,y) &= \int_{G^{r(x)}} c(z)D^{-1/2}(z)d(z^{-1}x)f(z^{-1}y) \,d\lambda^{r(x)}(z) = (T_{\phi}(\omega_{1})d)(x,y), \\ e_{2}(x,y) &= \int_{G^{r(x)}} c(z)d(z^{-1}x)D^{-1/2}(z) \,d\lambda^{r(x)}(z)f(y) = (T_{\phi}(\omega_{2})d)(x,y), \\ e_{3}(x,y) &= c(x)\int_{G^{r(y)}} f(z)D^{-1/2}(z)g(z^{-1}y) \,d\lambda^{r(y)}(z) = (S_{\phi}(\omega_{3})g)(x,y), \\ e_{4}(x,y) &= \int_{G^{r(x)}} c(z_{1})D^{-1/2}(z_{1})d(z_{1}^{-1}x)(L(f)g)(z_{1}^{-1}y) \,d\lambda^{r(x)}(z_{1}) \\ &= \int_{G^{r(x)}} \int_{G^{s(z_{1})}} c(z_{1})D^{-1/2}(z_{1})d(z_{1}^{-1}x)f(z_{2})D^{-1/2}(z_{2})g(z_{2}^{-1}z_{1}^{-1}y) d\lambda^{s(z_{1})}(z_{2}) d\lambda^{r(x)}(z_{1}) \end{split}$$

$$\begin{split} &= \int\limits_{G^{r(x)}} \int\limits_{G^{r(x)}} c(z_1) d(z_1^{-1}x) f(z_1^{-1}z_2') D^{-1/2}(z_2') g(z_2'^{-1}y) \, \mathrm{d}\lambda^{r(x)}(z_2') \, \mathrm{d}\lambda^{r(x)}(z_1) \\ &= (S_{\phi}(\omega_{c,d,f})g)(x,y). \end{split}$$

By Lemmas 7.2 and 8.11, sections of the form like $\omega_1, \omega_2, \omega_3$ or ω_4 , respectively, are linearly dense in $\Gamma_c(\mathcal{F}^2_{r,r})$. Therefore, $[\delta(\pi(C^*_r(\mathcal{F})))_{\phi}|\alpha\rangle_2] = [T_{\phi}(\Gamma_c(\mathcal{F}^2_{r,r}))] = [|\alpha\rangle_2 \pi_{\phi}(C^*_r(\mathcal{F}))]$ and similarly $[\delta(\pi(C^*_r(\mathcal{F})))|\gamma\rangle_1 C^*_r(G)] = \begin{bmatrix} \bigcup_{\phi \in \mathcal{W}} S_{\phi}(\Gamma_c(\mathcal{F}^2_{r,r})) \end{bmatrix} = [|\gamma\rangle_1 C^*_r(G)]$.

Given $g, g' \in C_c(G)$, define $h_{g,g'} \in C_c(G)$ by

(8.1)
$$h_{g,g'}(z) = \int_{G^{r(z)}} \overline{g(y)} g'(z^{-1}y) \, \mathrm{d}\lambda^{r(z)}(y) \quad \text{for all } z \in G$$

LEMMA 8.12. Let $c \in \Gamma_c(\mathcal{F})$, $g, g' \in C_c(G)$. Then $\langle j(g)|_2 \delta(\pi(c))_{\phi} | j(g') \rangle_2 = \pi_{\phi}(c')$, where $c'(x) = c(x)h_{g,g'}(x)$ for all $x \in G$.

Proof. The operators on both sides map each $d \in \Gamma_{c}(\mathcal{F})$ to the section

$$x \mapsto \int_{G^{r(x)}} \int_{G^{r(x)}} \int_{G^{r(x)}} \overline{g(y)} c(z) d(z^{-1}x) g'(z^{-1}y) D^{-1/2}(z) \, \mathrm{d}\lambda^{r(x)}(z) \, \mathrm{d}\lambda^{r(x)}(y). \quad \blacksquare$$

8.3. THE REDUCED CROSSED PRODUCT OF THE COACTION. The bundle $\mathcal{F}_{s,r}^2$ carries the structure of a Fell bundle, and the reduced crossed product $\pi(C_r^*(\mathcal{F})) \rtimes_r C_0(G)$ for the coaction δ constructed above can be identified with $C_r^*(\mathcal{F}_{s,r}^2)$ as follows.

Denote by $G \ltimes G$ the transformation groupoid for the action of G on itself given by right multiplication. Thus, $G \ltimes G = G_s \times_r G$ as a set, $(G \ltimes G)^0 = \bigcup_{u \in G^0} \{u\} \times G^u$ can be identified with G via $(r(y), y) \equiv y$, the range map \tilde{r} , the

source map \tilde{s} , and the multiplication are given by $(x, y) \xrightarrow{\tilde{r}} xy$, $(x, y) \xrightarrow{\tilde{s}} y$, and $((x, y), (x', y')) \mapsto (xx', y')$, respectively, and the topology on $G \ltimes G$ is the weakest topology that makes \tilde{r}, \tilde{s} and the map $(x, y) \mapsto x$ continuous. We equip $G \ltimes G$ with the right Haar system $\tilde{\lambda}^{-1}$ given by $\tilde{\lambda}_y^{-1}(C \times \{y\}) = \lambda_{r(y)}^{-1}(C)$ for all $C \subseteq G_{r(y)}, y \in G$.

The bundle $\mathcal{F}_{s,r}^2$ is a Fell bundle on $G \ltimes G$ with respect to the multiplication and involution given by $((f, y), (f', y')) \mapsto (ff', y')$ and $(f, y) \mapsto (f^*, p(f)y)$. The convolution product in $\Gamma_{c}(\mathcal{F}_{s,r}^2)$ is given by

(8.2)
$$(cd)(x,y) = \int_{G_{r(y)}} c(xz^{-1},zy)d(z,y) \, \mathrm{d}\lambda_{r(y)}^{-1}(z)$$

for all $c, d \in \Gamma_c(\mathcal{F}^2_{s,r}), (x, y) \in G_s \times_r G$, because $(G \ltimes G)_{\tilde{s}(x,y)} = G_{r(y)} \times \{y\}$ and $(x, y)(z, y)^{-1} = (xz^{-1}, zy)$ for all $z \in G_{r(y)}$.

PROPOSITION 8.13. There exists a unique isomorphism $\pi(C_r^*(\mathcal{F})) \rtimes_r C_0(G) \rightarrow C_r^*(\mathcal{F}_{s,r}^2)$ such that $\delta(\pi(c))(1 \bigotimes_{\mathfrak{b}} f) \mapsto L_{\mathcal{F}_{s,r}^2}(d)$ whenever $c \in \Gamma_c(\mathcal{F})$, $f \in C_c(G)$, and d(x,y) = c(x)f(y) for all $(x,y) \in G_s \times_r G$.

Let $\phi \in \mathcal{W}$. Then the map $r^*\phi: \Gamma_0((\mathcal{F}^2_{s,r})^0) \to C_0(G)$ given by $(r^*\phi(c))(y) = \phi_{r(y)}(c(r(y), y))$ for all $c \in \Gamma_0((\mathcal{F}^2_{s,r})^0)$ and $y \in G$ is a $C_0(G)$ -weight. One easily verifies that there exists a representation $L_{r^*\phi}: C^*_r(\mathcal{F}^2_{s,r}) \to \mathcal{L}(\Gamma^2(\mathcal{F}^2_{s,r}, \tilde{\lambda}^{-1}; r^*\phi))$ such that $L_{r^*\phi}(c)d = cd$ for all $c, d \in \Gamma_c(\mathcal{F}^2_{s,r})$.

LEMMA 8.14. (i) There exists a unique unitary $U_{\phi} : \Gamma^{2}(\mathcal{F}_{s,r}^{2}, \tilde{\lambda}^{-1}; r^{*}\phi) \otimes H \rightarrow \Gamma^{2}(\mathcal{F}_{s,r}^{2}, v_{s,r}^{2}; \phi) \subseteq K_{\hat{\delta}} \otimes_{\mathfrak{b}^{\dagger}} H$ such that $(U_{\phi}(e \otimes g))(x, y) = e(x, y)g(y)D^{-1/2}(x)$ for all $e \in \Gamma_{c}(\mathcal{F}_{s,r}^{2}), g \in C_{c}(G), (x, y) \in G_{s} \times_{r} G.$ (ii) $\delta(\pi(c))(1 \otimes f)X_{\phi}U_{\phi} = X_{\phi}U_{\phi}(L_{r^{*}\phi}(d) \otimes \mathrm{id})$ for all c, d, f as in Proposition 8.13.

Proof. (i) For all *e*, *g* as in above,

$$\|U_{\phi}(e \otimes g)\|^{2} = \int_{G} \int_{G_{r(y)}} \phi_{s(x)}(e(x,y)^{*}e(x,y))|g(y)|^{2} d\lambda_{r(y)}^{-1}(x) \int d\nu(y) = \|e \otimes g\|^{2}.$$

(ii) Let c, d, e, f, g, (x, y) as above and $\widehat{\Delta}(f)_{\phi} = X_{\phi}^*(1 \underset{\mathfrak{b}}{\otimes} f)X_{\phi}$. A short calculation shows that $(\widehat{\Delta}(f)_{\phi}U_{\phi}(e \otimes g))(x, y) = f(xy)e(x, y)g(y)D^{-1/2}(x)$. Using (8.2), we find that $((\pi_{\phi}(c) \underset{\mathfrak{b}}{\otimes} \operatorname{id})\widehat{\Delta}(f)_{\phi}U_{\phi}(e \otimes g))(x, y)$ is equal to

$$\begin{split} &\int\limits_{G^{r(x)}} c(z)f(z^{-1}xy)e(z^{-1}x,y)g(y)D^{-1/2}(z)D^{-1/2}(z^{-1}xy)\,\mathrm{d}\lambda^{r(x)}(z) \\ &= \int\limits_{G_{s(x)}} c(xz^{-1})f(zy)e(z,y)g(y)D^{-1/2}(xy)\,\mathrm{d}\lambda^{-1}_{s(x)}(z) \\ &= \int\limits_{G_{s(x)}} d(xz^{-1},zy)e(z,y)g(y)D^{-1/2}(xy)\,\mathrm{d}\lambda^{-1}_{s(x)}(z) = (U_{\phi}(de \otimes g))(x,y). \end{split}$$

So, $\delta(\pi(c))(1 \underset{\mathfrak{b}}{\otimes} f)X_{\phi}U_{\phi} = X_{\phi}(\pi_{\phi}(c) \underset{\mathfrak{b}^{\dagger}}{\otimes} \mathrm{id})\widehat{\Delta}(f)_{\phi}X_{\phi}U_{\phi} = X_{\phi}U_{\phi}(L_{r^{*}\phi}(d) \otimes \mathrm{id}).$

Proof of Proposition 8.13. Consider the *-homomorphism

$$\Phi: C^*_{\mathbf{r}}(\mathcal{F}^2_{s,r}) \to \mathcal{L}(K_{\gamma \bigotimes_{\mathfrak{b}} \beta} H), \quad L_{\mathcal{F}^2_{s,r}}(d) \mapsto \bigoplus_{\phi \in \mathcal{W}} X_{\phi} U_{\phi}(L_{r^*\phi}(d) \otimes \mathrm{id}) U^*_{\phi} X^*_{\phi}.$$

By part (ii) of the lemma above, $\Phi(C_r^*(\mathcal{F}_{s,r}^2))$ contains $[\delta(\pi(C_r^*(\mathcal{F})))(1 \bigotimes_{\mathfrak{b}} C_0(G))] = \pi(C_r^*(\mathcal{F})) \rtimes_r C_0(G)$. The same lemma implies that this inclusion is an equality because the map $a \to \bigoplus_{\phi} a \otimes$ id is continuous with respect to the inductive limit topology on $\Gamma_c(\mathcal{F}_{s,r}^2)$ and sections of the form $(x, y) \mapsto c(x)f(y)$, where $c \in \Gamma_c(\mathcal{F})$,

 $f \in C_{c}(G)$, are dense in $\Gamma_{c}(\mathcal{F}_{s,r}^{2})$ by Lemma 7.2. Finally, Lemma 6.4 implies that $\left[\bigcap_{\phi} \ker r^{*}\phi\right] = 0$, and therefore Φ is injective.

PROPOSITION 8.15. If \mathcal{F} is saturated, then $C^*_r(\mathcal{F}^2_{s,r}) \cong \mathcal{K}(\Gamma^2(\mathcal{F}, \lambda^{-1})).$

Proof. To simplify notation, let $\Gamma^2 = \Gamma^2(\mathcal{F}, \lambda^{-1})$, $\tilde{\Gamma}^2 = \Gamma^2(\mathcal{F}_{s,r}^2, \tilde{\lambda}^{-1})$, $\Gamma_0 = \Gamma_0(\mathcal{F}^0)$, $\tilde{\Gamma}_0 = \Gamma_0((\mathcal{F}_{s,r}^2)^0)$. There exists a unitary $\Psi : \Gamma^2 \otimes_{s^*} C_0(G) \to \tilde{\Gamma}^2$ such that $(\Psi(c \otimes f))(x, y) = c(xy)f(y)$ for all $c \in \Gamma_c(\mathcal{F})$, $f \in C_c(G)$, $(x, y) \in G_s \times_r G$, because

$$\begin{split} \langle \Psi(c \otimes f) | \Psi(c' \otimes f') \rangle ((r(y), y)) &= \int\limits_{G_{r(y)}} c(xy)^* c'(xy) \, \mathrm{d}\lambda_{r(y)}^{-1}(x) \overline{f(y)} f'(y) \\ &= \overline{f(y)} \langle c | c' \rangle_{\Gamma^2}(s(y)) f(y) = \langle c \otimes f | c' \otimes f' \rangle(y) \end{split}$$

for all $c, c' \in \Gamma_c(\mathcal{F})$, $f, f' \in C_c(G)$, $y \in G$ by right-invariance of λ^{-1} . The *-homomorphism $\Phi : \mathcal{K}(\Gamma^2) \to \mathcal{L}(\widetilde{\Gamma}^2)$ given by $k \mapsto \Psi(k \otimes_{s^*} id)\Psi^*$ is injective because $s^* : C_0(G^0) \to \mathcal{L}(C_0(G))$ is injective, and the claim follows once we have shown that $\Phi(\mathcal{K}(\Gamma^2)) = C^*_r(\mathcal{F}^2_{s,r})$. Let $d, d' \in \Gamma_c(\mathcal{F})$ and denote by $|d\rangle\langle d'| \in \mathcal{K}(\Gamma^2)$ the operator given by $e \mapsto d\langle d'|e\rangle$. Then for all c, f, (x, y) as above,

$$\begin{aligned} (\Psi(|d\rangle\langle d'|c \otimes f))(x,y) &= \int\limits_{G_{s(y)}} d(xy)d'(z)^*c(z)f(y) \, \mathrm{d}\lambda_{s(y)}^{-1}(z) \\ &= \int\limits_{G_{s(y)}} d(xy)d'(z)^*(\Psi(c \otimes f))(zy^{-1},y) \, \mathrm{d}\lambda_{s(y)}^{-1}(z) \\ &= \int\limits_{G_{r(y)}} d(xy)d'(z'y)^*(\Psi(c \otimes f))(z',y) \, \mathrm{d}\lambda_{r(y)}^{-1}(z'). \end{aligned}$$

Comparing with equation (8.2), we find that $\Psi(|d\rangle \langle d'| \otimes id) \Psi^* = L_{\mathcal{F}^2_{s,r}}(e)$, where $e \in \Gamma_c(\mathcal{F}^2_{s,r})$ is given by $e(xz^{-1}, zy) = d(xy)d'(zy)^*$, or equivalently, by $e(x', y') = d(x'y')d'(y')^*$ for all $(x', y') \in G_s \times_r G$. Since \mathcal{F} is saturated, Lemma 7.2 implies that sections of this form are dense in $\Gamma_c(\mathcal{F}^2_{s,r})$ with respect to the inductive limit topology, and since the map $e \mapsto L_{\mathcal{F}^2_{s,r}}(e)$ is continuous with respect to this topology, we can conclude that $\Phi(\mathcal{K}(\Gamma^2)) = \Psi(\mathcal{K}(\Gamma^2) \otimes id)\Psi^* = C^*_r(\mathcal{F}^2_{s,r})$.

COROLLARY 8.16. If \mathcal{F} is saturated, then $\pi(C_r^*(\mathcal{F})) \rtimes_r C_0(G)$ and $\Gamma_0(\mathcal{F}^0)$ are Morita equivalent.

Proof. One easily verifies that $\Gamma^2(\mathcal{F}, \lambda^{-1})$ is full.

EXAMPLE 8.17. Let σ be an action of G on an admissible $C_0(G^0)$ -algebra C and let δ_{σ} be the corresponding coaction of $C_0(G)$ on FC (Proposition 6.10). Then there exists an admissible Fell bundle C on G with fibre $C_x = C_{r(x)}$ for each

 $x \in G$, continuous sections $\Gamma_0(\mathcal{C}) = r^*C$, and multiplication and involution given by $cd = c\sigma_x(d)$, $c^* = \sigma_{x^{-1}}(c^*)$ for all $c \in \mathcal{C}_x$, $d \in \mathcal{C}_y$, $(x, y) \in G_s \times_r G$ [15], and the identity on $\Gamma_c(\mathcal{C}) = C_c(G)r^*C$ extends to an isomorphism $C_r^*(\mathcal{C}) \to C \rtimes_r G$. One easily verifies that with respect to the isomorphism $\pi(C_r^*(\mathcal{C})) \cong C_r^*(\mathcal{C}) \cong$ $C \rtimes_r G \cong FC \rtimes_r C_r^*(G)$ of Proposition 6.14, the coaction of Theorem 8.9 coincides with the dual coaction on $FC \rtimes_r C_r^*(G)$. Moreover, the Fell bundle \mathcal{C} is saturated and $C_r^*(\mathcal{C}) \rtimes_r C_0(G) \cong FC \rtimes_r C_r^*(G) \rtimes_r C_0(G)$ is Morita equivalent to $\Gamma_0(\mathcal{C}^0) \cong C$, as we already know by Theorem 4.11.

REMARK 8.18. The Fell bundle \mathcal{F} can be equipped with the structure of an $\mathcal{F}_{s,r}^2 - \mathcal{F}^0$ -equivalence in the sense of [19] in a straightforward way.

8.4. FUNCTORIALITY OF THE CONSTRUCTION. Let \mathcal{G} , \mathcal{F} be admissible Fell bundles on G with associated representations $((K_{\mathcal{G}}, \gamma_{\mathcal{G}}, \hat{\delta}_{\mathcal{G}}), X_{\mathcal{G}}), ((K_{\mathcal{F}}, \gamma_{\mathcal{F}}, \hat{\delta}_{\mathcal{F}}), X_{\mathcal{F}})$ and coactions $(\pi_{\mathcal{G}}(C_{r}^{*}(\mathcal{G}))_{K_{\mathcal{G}}}^{\gamma_{\mathcal{G}}}, \delta_{\mathcal{G}}), (\pi_{\mathcal{F}}(C_{r}^{*}(\mathcal{F}))_{K_{\mathcal{F}}}^{\gamma_{\mathcal{F}}}, \delta_{\mathcal{F}})$, and let T be a morphism from \mathcal{G} to \mathcal{F} .

PROPOSITION 8.19. There exists a unique morphism \widetilde{T}_* from $(\pi_{\mathcal{G}}(C^*_{\mathbf{r}}(\mathcal{G}))^{\gamma_{\mathcal{G}}}_{K_{\mathcal{G}}}, \delta_{\mathcal{G}})$ to $(\pi_{\mathcal{F}}(C^*_{\mathbf{r}}(\mathcal{F}))^{\gamma_{\mathcal{F}}}_{K_{\mathbf{r}}}, \delta_{\mathcal{F}})$ that satisfies $\widetilde{T}_*(\pi_{\mathcal{G}}(a)) = \pi_{\mathcal{F}}(T_*(a))$ for all $a \in \Gamma_{\mathbf{c}}(\mathcal{G})$.

The proof involves the following construction.

LEMMA 8.20. Let $\phi \in W(\Gamma_0(\mathcal{F}^0))$, $f \in \Gamma_0(\mathcal{F}^0)$ and define $\psi \in W(\Gamma_0(\mathcal{G}^0))$ by $g \mapsto \phi(f^*T^0_*(g)f)$.

(i) There exists an isometry $T_{\phi}^{f}: K_{\psi} \to K_{\phi}$ such that $T_{f}^{\phi}g = T_{*}(g)f$ for all $g \in \Gamma_{c}(\mathcal{G})$.

(ii) $T^f_{\phi}j_{\psi}(g) = j_{\phi}(T_*(g)f), T^f_{\phi}\hat{j}_{\psi}(g) = \hat{j}_{\psi}(T_*(g)f), \text{ and } T^f_{\phi}\pi_{\psi}(g) = \pi_{\phi}(T_*(g))T^f_{\phi}$ for all $g \in \Gamma_{c}(\mathcal{G})$.

Denote also the map $K_{\mathcal{G}} \to K_{\psi} \hookrightarrow K_{\mathcal{F}}$ given by $(\xi_{\psi'})_{\psi'} \mapsto T_{\phi}^f \xi_{\psi}$ by T_{ϕ}^f .

(iii) T_{ϕ}^{f} is a semi-morphism from $(K_{\mathcal{G}}, \widehat{\delta}_{\mathcal{G}}, \gamma_{\mathcal{G}})$ to $(K_{\mathcal{F}}, \widehat{\delta}_{\mathcal{F}}, \gamma_{\mathcal{F}})$ and $(T_{\phi}^{f} \bigotimes_{\mathfrak{b}} \mathrm{id})X_{\mathcal{G}} = X_{\mathcal{F}}(T_{\phi}^{f} \bigotimes_{\mathfrak{b}} \mathrm{id}).$

(iv) $\delta_{\mathcal{F}}(\pi_{\mathcal{F}}(h))(T^{f}_{\phi} \underset{\mathfrak{b}}{\otimes} \mathrm{id})\delta_{\mathcal{G}}(\pi_{\mathcal{G}}(g)) = \delta_{\mathcal{F}}(\pi_{\mathcal{F}}(hT_{*}(g)))(T^{f}_{\phi} \underset{\mathfrak{b}}{\otimes} \mathrm{id}) \text{ for all } h \in \Gamma_{c}(\mathcal{F}), g \in \Gamma_{c}(\mathcal{G}).$

Proof. (i) Uniqueness is clear. Existence follows from the fact that for all $g, g' \in \Gamma_{c}(\mathcal{G})$,

$$\begin{split} \langle T_*(g)f|T_*(g')f\rangle_{K_{\psi}} &= \int_G \phi_{s(x)}(f(s(x))^*T(g(x)^*g'(x))f(s(x))) \int \mathrm{d}\nu(x) \\ &= \int_G \psi_{s(x)}(g(x)^*g'(x)) \int \mathrm{d}\nu(x) = \langle g|g\rangle_{K_{\psi}}. \end{split}$$

(ii) Straightforward.

(iii) By (ii), $T^f_{\phi}\gamma_{\mathcal{G}} \subseteq \gamma_{\mathcal{F}}$ and $T^f_{\phi}\widehat{\delta}_{\mathcal{G}} \subseteq \widehat{\delta}_{\mathcal{F}}$. For all $\omega \in \Gamma_{c}(\mathcal{G}^{2}_{s,r})$ and $(x,y) \in G_r \times_{r} G$,

$$((T^{f}_{\phi} \underset{\mathfrak{b}}{\otimes} \mathrm{id})X_{\mathcal{G}}\omega)(x,y) = \omega(x,x^{-1}y)f(s(x)) = (X_{\mathcal{F}}(T^{f}_{\phi} \underset{\mathfrak{b}^{\dagger}}{\otimes} \mathrm{id})\omega)(x,y).$$

(iv) By parts (ii) and (iii),
$$X_{\mathcal{F}}(\pi_{\mathcal{F}}(h) \underset{\mathfrak{b}^{\dagger}}{\otimes} \mathrm{id}) X_{\mathcal{F}}^*(T_{\phi}^f \underset{\mathfrak{b}}{\otimes} \mathrm{id}) X_{\mathcal{G}}(\pi_{\mathcal{G}}(g) \underset{\mathfrak{b}^{\dagger}}{\otimes} \mathrm{id}) X_{\mathcal{G}}^* = X_{\mathcal{F}}(\pi_{\mathcal{F}}(hT_*(g)) \underset{\mathfrak{b}^{\dagger}}{\otimes} \mathrm{id}) X_{\mathcal{F}}^*(T_{\phi}^f \underset{\mathfrak{b}}{\otimes} \mathrm{id}) \text{ for all } g \in \Gamma_{\mathrm{c}}(\mathcal{G}) \text{ and } h \in \Gamma_{\mathrm{c}}(\mathcal{F}).$$

Proof of Proposition 8.19. Denote by $\mathcal{T} \subseteq \mathcal{L}(K_{\mathcal{G}}, K_{\mathcal{F}})$ the closed linear span of all operators T_{ϕ}^{f} , where $\phi \in \mathcal{W}(\Gamma_{0}(\mathcal{F}^{0}))$ and $f \in \Gamma_{0}(\mathcal{F}^{0})$. Then Lemma 8.20 and Proposition 7.9 imply that $S\pi_{\mathcal{G}}(g) = \pi_{\mathcal{F}}(T_{*}(g))S$ for all $S \in \mathcal{T}, g \in \Gamma_{c}(\mathcal{G})$ and that

$$[\mathcal{T}\gamma_{\mathcal{G}}] = \left[\bigcup_{\phi} j_{\phi}(T_*(\Gamma_{c}(\mathcal{G}))\Gamma_{0}(\mathcal{F}^{0}))\right] = \left[\bigcup_{\phi} j_{\phi}(\Gamma_{c}(\mathcal{F}))\right] = \gamma_{\mathcal{F}}$$

By Proposition 7.9, T_* extends to a nondegenerate *-homomorphism $C^*_r(\mathcal{G}) \to M(C^*_r(\mathcal{F}))$. Henceforth, there exists a semi-morphism \widetilde{T}_* from $\pi_{\mathcal{G}}(C^*_r(\mathcal{G}))_{K_{\mathcal{G}}}^{\gamma_{\mathcal{G}}}$ to $\pi_{\mathcal{F}}(C^*_r(\mathcal{F}))_{K_{\mathcal{F}}}^{\gamma_{\mathcal{F}}}$ such that $\widetilde{T}_*(\pi_{\mathcal{G}}(g)) = \pi_{\mathcal{F}}(T_*(g))$ for all $g \in \Gamma_c(\mathcal{G})$. For all $h \in \pi_{\mathcal{F}}(\Gamma_c(\mathcal{F})), g \in \pi_{\mathcal{G}}(\Gamma_c(\mathcal{G})), S \in \mathcal{T}$,

$$\delta_{\mathcal{F}}(h) \cdot (\widetilde{T}_* * \mathrm{id})(\delta_{\mathcal{G}}(g)) \cdot (S \underset{\mathfrak{b}}{\otimes} \mathrm{id}) = \delta_{\mathcal{F}}(h)(S \underset{\mathfrak{b}}{\otimes} \mathrm{id})\delta_{\mathcal{G}}(g) = \delta_{\mathcal{F}}(h\widetilde{T}_*(g))(S \underset{\mathfrak{b}}{\otimes} \mathrm{id})$$

by Lemma 8.20, and therefore $\delta_{\mathcal{F}}(h) \cdot (\widetilde{T}_* * \mathrm{id})(\delta_{\mathcal{G}}(g)) = \delta_{\mathcal{F}}(h\widetilde{T}_*(g))$.

Denote by \mathbf{Fell}_G^a the category of all admissible Fell bundles on *G*, and by $\mathbf{Coact}_{C^*(G)}^a$ the category of very fine left-full coactions of $C_r^*(G)$.

THEOREM 8.21. The assignments $\mathcal{F} \mapsto (\pi_{\mathcal{F}}(C^*_{\mathbf{r}}(\mathcal{F}))^{\gamma_{\mathcal{F}}}_{K_{\mathcal{F}}}, \delta_{\mathcal{F}})$ and $T \mapsto \widetilde{T}_*$ form a faithful functor $\check{\mathbf{F}} : \mathbf{Fell}^a_G \to \mathbf{Coact}^a_{C^*_*(G)}$.

Proof. Functoriality of the constructions is evident. Assume that $\mathbf{\check{F}}S = \mathbf{\check{F}}T$ for some morphisms S, T from \mathcal{F} to \mathcal{G} in \mathbf{Fell}_G^a . Then the maps $S_*, T_* : \Gamma_c(\mathcal{F}) \to \Gamma_c(\mathcal{M}(\mathcal{G}))$ coincide because $\pi_{\mathcal{G}}$ is injective. Since $\{a(x) : a \in \Gamma_c(\mathcal{F})\} = \mathcal{F}_x$ for each $x \in G$ and $S(a(x)) = (S_*a)(x) = (T_*a)(x) = T(a(x))$ for each $a \in \Gamma_c(\mathcal{F})$, $x \in G$, we can conclude that S = T.

9. FROM COACTIONS OF $C_r^*(G)$ TO FELL BUNDLES FOR ÉTALE G

We now assume that the groupoid *G* is étale in the sense that the set \mathfrak{G} of all open subsets $U \subseteq G$ for which the restrictions $r_U = r|_U : U \to r(U)$ and $s_U = s|_U : U \to s(U)$ are homeomorphisms is a cover of *G*; see [23]. Moreover, we assume that the Haar systems λ and λ^{-1} are the families of counting measures. Then the functor $\check{\mathbf{F}}$ has a right adjoint $\check{\mathbf{G}}$ and embeds the category of

admissible Fell bundles into a category of very fine coactions of $C_r^*(G)$ as a full and coreflective subcategory. The construction of the functor $\check{\mathbf{G}}$ uses the correspondence between Banach bundles and convex Banach modules developed in [8].

9.1. THE FELL BUNDLE OF A COACTION OF $C_r^*(G)$. Let δ be an injective coaction of $C_r^*(G)$ on a C^* -b-algebra $\mathcal{C} = C_K^{\gamma}$. Since G is étale, $\rho_{\beta}(\mathfrak{B}) \subseteq C_r^*(G)$ and $\delta(C)|\gamma\rangle_1 \subseteq [|\gamma\rangle_1 C_r^*(G)]$. For each $U \in \mathfrak{G}$, we define a closed subspace

$$C_{U} := \{ c \in [C\rho_{\gamma}(C_{0}(s(U)))] : dle | \delta(c)|\gamma\rangle_{1} \subseteq [|\gamma\rangle_{1}L(C_{0}(U))] \} \subseteq C,$$

denote by $s_{U*} : C_0(U) \to C_0(s(U))$ and $r_{U*} : C_0(U) \to C_0(r(U))$ the pushforward of functions along s_U and r_U , respectively, and consider C_U as a right Banach $C_0(U)$ -module via the formula $c \cdot f := c\rho_{\gamma}(s_{U*}(f))$. Denote by $\Gamma_f(\mathcal{F})$ the space of all sections of \mathcal{F} that can be written as finite sums of sections in $\Gamma_0(\mathcal{F}|_U)$, where $U \in \mathfrak{G}$. Then $\Gamma_f(\mathcal{F})$ is a *-algebra with respect to the operations defined in (7.1), and one has natural inclusions $\Gamma_c(\mathcal{F}) \subseteq \Gamma_f(\mathcal{F}) \subseteq C_r^*(\mathcal{F})$ of *-algebras.

PROPOSITION 9.1. There exist a continuous Fell bundle \mathcal{F} on G and a *-homomorphism $\iota : \Gamma_f(\mathcal{F}) \to C$ such that for each $U \in \mathfrak{G}$, the map ι restricts to an isometric isomorphism $\iota_U : \Gamma_0(\mathcal{F}|_U) \to C_U$ of Banach $C_0(U)$ -modules. If (\mathcal{F}', ι') is another such pair, then there exists an isomorphism $T : \mathcal{F} \to \mathcal{F}'$ such that $\iota' \circ T_* = \iota$.

The proof requires some preliminaries. First, for all $c \in C$, $f \in C_0(G^0)$,

$$\delta(c\rho_{\gamma}(f)) = \delta(c)\rho_{(\gamma \triangleright \alpha)}(f) = \delta(c)(1 \underset{\mathfrak{b}}{\otimes} \rho_{\alpha}(f)) = \delta(c)(1 \underset{\mathfrak{b}}{\otimes} r^{*}(f)).$$

LEMMA 9.2. Let $U, V \in \mathfrak{G}$.

(i) $c \cdot f = \rho_{\gamma}(r_{U*}(f))c$ for each $c \in C_U$ and $f \in C_0(U)$.

(ii) $C_V C_U \subseteq C_{VU}$, $(C_U)^* = C_{U^{-1}}$, and $C_U = [C_V C_0(U)] \subseteq C_V$ if $U \subseteq V$.

(iii) $C_{s(U)}$ is a continuous $C_0(s(U))$ -algebra.

(iv) C_U is a convex and continuous Banach $C_0(U)$ -module.

Proof. (i) Let *c*, *f* as above. Since $L(g)r^*(s_{U*}(f)) = r^*(r_{U*}(f))L(g)$ for all $g \in C_0(U)$, we have $\delta(c \cdot f) = \delta(c)(1 \bigotimes_{\mathfrak{b}} r^*(s_{U*}(f))) = (1 \bigotimes_{\mathfrak{b}} r^*(r_{U*}(f)))\delta(c) = \delta(\rho_{\gamma}(r_{U*}(f))c)$ and by injectivity of δ also $c \cdot f = \rho_{\gamma}(r_{U*}(f))c$. (ii) Clearly, $\delta(C_V C_U) |\gamma\rangle_1 \subseteq |\gamma\rangle_1 L(C_0(VU))$. Using (i) twice, we find

$$C_V C_U \subseteq [C_V \rho_{\gamma}(C_0(s(V))C_0(r(U)))C_U]$$

= $[C_V \rho_{\gamma}(C_0(s(V) \cap r(U))C_U)] \subseteq [C \rho_{\gamma}(C_0(s(VU)))].$

Consequently, $C_V C_U \subseteq C_{VU}$. By (i) again, we have $(C_U)^* = [\rho_{\gamma}(C_0(r(U)))C_U]^* \subseteq [C\rho_{\gamma}(C_0(s(U^{-1})))]$, and using the relation $\delta(C_U^*)|\gamma\rangle_1 \subseteq [|\gamma\rangle_1 C_r^*(G)]$, we obtain

$$\delta(C_{U}^{*})|\gamma\rangle_{1} \subseteq [|\gamma\rangle_{1}\langle\gamma|_{1}\delta(C_{U})^{*}|\gamma\rangle_{1}]$$
$$\subseteq [|\gamma\rangle_{1}L(C_{0}(U))^{*}\langle\gamma|_{1}|\gamma\rangle_{1}] = [|\gamma\rangle_{1}L(C_{0}(U^{-1}))].$$

If $U \subseteq V$, then $C_U \subseteq [C_V C_0(U)] \subseteq C_V$, and $C_V C_0(U) \subseteq C_U$ because

$$\delta(C_V C_0(U))|\gamma\rangle_1 = \delta(C_V)|\gamma\rangle_1 r^*(C_0(s(U)))$$

$$\subseteq [|\gamma\rangle_1 L(C_0(V))r^*(C_0(s(U)))] = [|\gamma\rangle_1 L(C_0(U))].$$

(iii) By (ii), $C_{s(U)}$ is a C^* -algebra. Consider $|\gamma\rangle_1$ as a Hilbert C^* -module over $r^*(C_0(G^0)) \cong C_0(G^0)$. Since $\delta(C_{G^0})|\gamma\rangle_1 \subseteq |\gamma\rangle_1$ and $\delta(c \cdot f)|\eta\rangle_1 = \delta(c)|\eta\rangle_1 r^*(f)$ for all $c \in C_{G^0}$, $f \in C_0(G^0)$, $\eta \in \gamma$, the formula $c \cdot |\eta\rangle_1 := \delta(c)|\eta\rangle_1$ defines a faithful field of representations $C_{G^0} \to \mathcal{L}(|\gamma\rangle_1)$ in the sense of Theorem 3.3 in [6]. Consequently, C_{G^0} is a continuous $C_0(G^0)$ -algebra and $C_{s(U)}$ a continuous $C_0(s(U))$ -algebra.

(iv) Let $c, c' \in C_U$ and $f, f' \in C_0(U)$ such that $0 \leq f, f'$ and $f + f' \leq 1$. Then $\|c \cdot f + c' \cdot f'\|^2 = \|c^* c \cdot g^2 + c^* c' \cdot gg' + c'^* c \cdot g'g + c'^* c' \cdot g'^2\|$, where $g = s_{U*}(f), g' = s_{U*}(f')$. Since $g^2 + gg'' + g'g + g'^2 \leq 1$ and $c^* c, c'^* c', c^* c', c'^* c'$ belong to the continuous $C_0(s(U))$ -algebra $C_{U^{-1}U}$, which is a convex Banach $C_0(s(U))$ -module, we get $\|cf + c'f'\|^2 \leq \max\{\|c\|, \|c'\|\}^2$. Finally, the norm $\|c_u\|^2 = \|(c^* c)_{u^{-1}u}\|$ depends continuously on $u \in U$ because $C_{U^{-1}U}$ is a continuous $C_0(s(U))$ -algebra.

Proof of Proposition 9.1. Using Lemma 9.2 and [8], one easily verifies that there exists a continuous Fell bundle \mathcal{F} on G with an isometric isomorphism $\iota_U : \Gamma_0(\mathcal{F}|_U) \to C_U$ of Banach $C_0(U)$ -modules for each $U \in \mathfrak{G}$ such that for all $U, V \in \mathfrak{G}$, the following properties hold. First, the map $\Gamma_0(\mathcal{F}|_U) \hookrightarrow \Gamma_0(\mathcal{F}|_V) \xrightarrow{\iota_V} C_V$ is equal to $\Gamma_0(\mathcal{F}|_U) \xrightarrow{\iota_U} C_U \hookrightarrow C_V$ if $U \subseteq V$, and second, $\iota_U(f)^* = \iota_{U^{-1}}(f^*)$, $\iota_{UV}(fg) = \iota_U(f)\iota_V(g)$ for all $f \in \Gamma_0(\mathcal{F}|_U), g \in \Gamma_0(\mathcal{F}|_V)$. Define $\iota : \Gamma_f(\mathcal{F}) \to C$ as follows. Given $a = \sum_i a_i \in \Gamma_f(\mathcal{F})$, where $a_i \in \Gamma_0(\mathcal{F}|_U)$ and $U_i \in \mathfrak{G}$, let $\iota(a) = \sum_i \iota_{U_i}(a_i)$. Using the preceding two properties of ι , one easily verifies that ι is well-defined and a *-homomorphism.

Denote by $p_0 : \Gamma_f(\mathcal{F}) \to \Gamma_0(\mathcal{F}^0)$ the restriction.

PROPOSITION 9.3. There exists a faithful conditional expectation p from the C^* -algebra $[\iota(\Gamma_f(\mathcal{F}))]$ to C_{G^0} satisfying $p \circ \iota = \iota_{G^0} \circ p_0$.

In the following lemma, $fh_{\xi,\xi'}$ denotes the pointwise product of functions $f, h_{\xi,\xi'} \in C_c(G)$, where $h_{\xi,\xi'}$ was defined in (8.1).

LEMMA 9.4. Let $\xi, \xi' \in C_c(G), c \in C, f \in C_c(G), \eta, \eta' \in \gamma$. Then:

(i) $\langle \eta |_1 \delta(\langle j(\xi) |_2 \delta(c) | j(\xi') \rangle_2) | \eta' \rangle_1 = \langle j(\xi) |_2 \Delta(\langle \eta |_1 \delta(c) | \eta' \rangle_1) | j(\xi') \rangle_2;$

(ii) $\langle j(\xi)|_2 \Delta(L(f)) | j(\xi') \rangle_2 = L(fh_{\xi,\xi'});$

(iii) $\langle j(\xi)|_2 \delta(c \cdot f) | j(\xi') \rangle_2 = c \cdot (fh_{\xi,\xi'})$ if $c \in C_U$ and $f \in C_0(U)$ for some $U \in \mathfrak{G}$.

Proof. (i) If $d = \langle j(\xi) |_2 \delta(c) | j(\xi') \rangle_2$, then $\delta(d) = \langle j(\xi) |_3 (\delta * id) (\delta(c)) | j(\xi') \rangle_3$ = $\langle j(\xi) |_3 (id *\Delta) (\delta(c)) | j(\xi') \rangle_3$ and $\langle \eta |_1 \delta(d) | \eta' \rangle_1 = \langle j(\xi) |_2 \Delta(\langle \eta |_1 \delta(c) | \eta' \rangle_1) | j(\xi') \rangle_2$. (ii) This is a special case of Lemma 8.12.

THOMAS TIMMERMANN

(iii) Let $\eta, \eta' \in \gamma$. Since $c \in C_U$, we have $\langle \eta |_1 \delta(c) | \eta' \rangle_1 = L(g)$ for some $g \in C_0(U)$. Let $\xi'' = r^*(s_{U*}(f))\xi$ and denote by $d_1, d_r \in C$ the left and the right hand side of the equation in (iii), respectively. Then $d_1 = \langle j(\xi) |_2 \delta(c) | j(\xi'') \rangle_2$, and by (i) and (ii),

$$\begin{aligned} \langle \eta |_1 \delta(d_1) | \eta' \rangle_1 &= \langle j(\xi) |_2 \Delta(\langle \eta |_1 \delta(c) | \eta' \rangle_1) | j(\xi'') \rangle_2 = \langle j(\xi) |_2 L(g) | j(\xi'') \rangle_2 = L(gh_{\xi,\xi''}), \\ \langle \eta |_1 \delta(d_r) | \eta' \rangle_1 &= \langle \eta |_1 \delta(c) | \eta' \rangle_1 r^* (s_{U*}(fh_{\xi,\xi'})) = L(g) L(s_{U*}(fh_{\xi,\xi'})). \end{aligned}$$

We can conclude that $\langle \eta |_1 \delta(d_1) | \eta' \rangle_1 = \langle \eta |_1 \delta(d_r) | \eta' \rangle_1$ because for all $x \in G$,

$$(gh_{\xi,\xi''})(x) = g(x) \int_{G^{r(x)}} \overline{\xi(y)} f(x)\xi'(x^{-1}y) \, \mathrm{d}\lambda^{r(x)}(y) = g(x)(s_{U*}(fh_{\xi,\xi'}))(s(x)).$$

Since η , $\eta' \in \gamma$ were arbitrary and δ is injective, we must have $d_1 = d_r$.

Proof of Proposition 9.3. Given a subset $U \subseteq G$, denote by χ_U its characteristic function. Using the same formulas as for elements of $C_c(G)$, we can define a map $j(\xi) : \mathfrak{K} \to H$ and the function $h_{\xi,\xi'}$ for the characteristic function $\xi = \xi' = \chi_{G^0}$ of $G^0 \subseteq G$, and then Lemma 9.4 still holds. Define $p : C \to C$ by $c \mapsto \langle j(\chi_{G^0})|_2 \delta(c)|j(\chi_{G^0})\rangle_2$. Then $\|p\| \leq \|j(\chi_{G^0})\|^2 = 1$, and the relation $h_{\chi_{G^0},\chi_{G^0}} = \chi_{G^0}$ and Lemma 9.4 imply that $p|_{C_{G^0}} = \text{id and } p|_{C_U} = 0$ whenever $U \in \mathfrak{G}$ and $U \cap G^0 = \emptyset$. Using a partition of unity argument and the fact that $G^0 \subseteq G$ is open and closed, we can conclude that $p \circ \iota = \iota_{G^0} \circ p_0$.

It remains to show that p is faithful. Using the right-regular representation of G, one easily verifies that $[C_r^*(G)'j(\chi_{G^0})\mathfrak{K}] = H$. Therefore, the map $q: C_r^*(G) \to \mathcal{L}(\mathfrak{K}), a \mapsto j(\chi_{G^0})^*aj(\chi_{G^0})$, is faithful in the sense that $q(a^*a) \neq 0$ if $a \neq 0$. If $c \in [\iota(\Gamma_f(\mathcal{F}))]$ and $p(c^*c) = 0$, then $\eta^*p(c^*c)\eta = q(\langle \eta^*|_1\delta(c^*c)|\eta\rangle_1) = 0$ and hence $\langle \eta^*|_1\delta(c^*c)|\eta\rangle_1 = 0$ and $\delta(c)|\eta\rangle_1 = 0$ for all $\eta \in \gamma$, whence $\delta(c) = 0$ and c = 0 by injectivity of δ .

Proposition 9.3 and Fact 3.11 of [15] imply:

COROLLARY 9.5. ι extends to an embedding $C^*_r(\mathcal{F}) \to C$.

We denote the extension above by ι again.

PROPOSITION 9.6. If δ is fine, then $\iota : C^*_r(\mathcal{F}) \to C$ is a *-isomorphism.

Proof. We only need to show that *C* is equal to the linear span of all C_U , where $U \in \mathfrak{G}$. Consider an element $d \in C$ of the form $d = \langle j(\xi)|_2 \delta(c)|j(\xi')\rangle_2$, where $c \in C, \xi \in C_c(V), \xi' \in C_c(V')$ for some $V, V' \in \mathfrak{G}$. Since *G* is étale and δ is fine, the closed linear span of all elements of the form like *d* is equal to $[\langle \alpha |_2 \delta(C) | \alpha \rangle_2] = [\langle \alpha |_2 | \alpha \rangle_2 C] = C$. We show that $d \in C_U$, where $U = VV'^{-1} \in \mathfrak{G}$, and then the claim follows. Let $\eta, \eta' \in \gamma$. By Lemma 9.4,

 $\langle \eta |_1 \delta(d) | \eta' \rangle_1 \in \langle j(\xi) |_2 \Delta(C_{\mathbf{r}}^*(G)) | j(\xi') \rangle_2 \subseteq [L(C_{\mathbf{c}}(G)h_{\xi,\xi'})] \subseteq L(C_0(U)).$

Using the relation $\delta(d)|\gamma\rangle_1 \subseteq [|\gamma\rangle_1 C_r^*(G)]$, we get $\delta(d)|\gamma\rangle_1 \subseteq [|\gamma\rangle_1 \langle \gamma|_1 \delta(d)|\gamma\rangle_1] \subseteq [|\gamma\rangle_1 L(C_0(U))]$. Moreover, since $h_{\xi,\xi'} \in C_c(U)$, we can choose $g \in C_0(U)$ with

 $h_{\xi,\xi'}g = h_{\xi,\xi'}$. Then $L(fh_{\xi,\xi'})r^*(s_{U^*}(g)) = L(fh_{\xi,\xi'})$ for each $f \in C_0(U)$, and hence $\langle \eta | {}_1 \delta(d\rho_{\gamma}(s_{U*}(g))) | \eta' \rangle_1 = \langle \eta | {}_1 \delta(d) | \eta' \rangle_1 r^*(s_{U*}(g)) = \langle \eta | {}_1 \delta(d) | \eta' \rangle_1$. Since δ is injective, we can conclude $d = d\rho_{\gamma}(s_{U*}(g)) \in C\rho_{\gamma}(C_0(s(U)))$ and finally $d \in C_{II}$.

PROPOSITION 9.7. If δ is fine, then \mathcal{F} is admissible.

Proof. The proof is similar to the proof of Lemma 6.2(i). By Lemma 9.2(iii), $\Gamma_0(\mathcal{F}^0) \cong C_{G^0}$ is a continuous $C_0(G^0)$ -algebra. Let $u \in G^0$, denote by $I_u \subset C_0(G^0)$ the ideal of all functions vanishing at u_i and assume $\mathcal{F}_u = 0$. Then $\Gamma_0(\mathcal{F}^0) =$ $[\Gamma_0(\mathcal{F}^0)I_u]$ and $[C_r^*(\mathcal{F})] = [C_r^*(\mathcal{F})\Gamma_0(\mathcal{F}^0)] = [C_r^*(\mathcal{F})I_u]$, whence $C = [C\rho_\gamma(I_u)]$. Define $i(\chi_{C^0})$ as in the proof of Proposition 9.3. Then $[\delta(C)|\gamma\rangle_1 C_r^*(G)] = [|\gamma\rangle_1 C_r^*(G)]$ and

$$\begin{aligned} [r^*(C_0(G^0))C_{\mathbf{r}}^*(G)] &= [\langle \gamma|_1 | \gamma \rangle_1 C_{\mathbf{r}}^*(G)] = [\langle \gamma|_1 \delta(CI_u) | \gamma \rangle_1 C_{\mathbf{r}}^*(G)] \\ &= [\langle \gamma|_1 | \gamma \rangle_1 r^*(I_u) C_{\mathbf{r}}^*(G)] = [r^*(I_u)C_{\mathbf{r}}^*(G)], \end{aligned}$$

whence $[i(\chi_{G^0})^*C^*_r(G)i(\chi_{G^0})] = I_\mu \neq C_0(G^0)$, a contradiction.

The construction of the Fell bundle is functorial with respect to the following class of morphisms.

DEFINITION 9.8. A morphism ρ of coactions (C_K^{γ}, δ_C) and $(D_1^{\varepsilon}, \delta_D)$ of $C_r^*(G)$ is strongly nondegenerate if $[\rho(\hat{C})D_{C^0}] = D$.

PROPOSITION 9.9. Let π be a strongly nondegenerate morphism of fine coactions $(C_{\kappa}^{\gamma}, \delta_{C}), (D_{1}^{\varepsilon}, \delta_{D})$ with associated Fell bundles \mathcal{F}, \mathcal{G} and *-homomorphisms $\iota_{\mathcal{F}}, \iota_{\mathcal{G}}$. Then there exists a unique morphism T from \mathcal{F} to \mathcal{G} such that $\iota_{\mathcal{G}} \circ T_* = \pi \circ \iota_{\mathcal{F}}$.

(-)-

Proof. Let
$$U, V \in \mathfrak{G}$$
. Then $\pi(C_U)D_V \subseteq D_{UV}$ because
 $\delta_D(\pi(C_U)D_V)|\varepsilon\rangle_1 = ((\pi * \mathrm{id})(\delta_C(C_U)))\delta_D(D_V)|\varepsilon\rangle_1$
 $\subseteq ((\pi * \mathrm{id})(\delta_C(C_U)))|\varepsilon\rangle_1 L(C_0(V))$
 $\subseteq |\varepsilon\rangle_1 L(C_0(U))L(C_0(V)) = |\varepsilon\rangle_1 L(C_0(UV))$
and $\pi(C_U)D_V \subseteq [\pi(C\rho_{\gamma}(C_0(s(U))))D_V] \subseteq [\pi(C)D\rho_{\varepsilon}(C_0(s(UV)))]$

where the last inclusion follows similarly as in the proof of Lemma 9.2(ii). Define a map $S_{U,V}: \Gamma_0(\mathcal{F}|_U) \times \Gamma_0(\mathcal{G}|_V) \to \Gamma_0(\mathcal{G}|_{UV})$ by $(f,g) \mapsto \iota_{\mathcal{G}}^{-1}(\pi(\iota_{\mathcal{F}}(f))\iota_{\mathcal{G}}(g)),$ let $(x, y) \in (U \times V) \cap G_s \times_r G$, and denote by $I_x \subseteq \Gamma_0(\mathcal{F}|_U), I_y \subseteq \Gamma_0(\mathcal{G}|_V)$, $I_{xy} \subseteq \Gamma_0(\mathcal{G}|_{UV})$ the subspaces of all sections vanishing at x, y, and xy, respectively. Using Lemma 9.2 (i), one easily verifies that $S_{U,V}$ maps $I_x \times \Gamma_0(\mathcal{G}|_V)$ and $\Gamma_0(\mathcal{F}|_U) \times I_v$ into I_{xy} . Hence, there exists a unique map $S_{x,y} : \mathcal{F}_x \times \mathcal{G}_y \to \mathcal{G}_{xy}$ such that $S_{x,y}(f(x), g(y)) = (S_{U,V}(f,g))(xy)$ for all $f \in \Gamma_0(\mathcal{F}|_U), g \in \Gamma_0(\mathcal{G}|_V)$, and this map depends on (x, y) but not on (U, V). For each $x \in G$ and $c \in \mathcal{F}_x$, define $T(c): \mathcal{G}|_{C^{s(x)}} \to \mathcal{G}|_{C^{r(x)}}$ by $T(c)d = S_{x,y}(c,d)$ for each $y \in G^{s(x)}$, $d \in \mathcal{G}_y$. One easily checks that then *T* is a continuous map from \mathcal{F} to $\mathcal{M}(\mathcal{G})$ which satisfies conditions (i) and (ii) of Definition 7.8, and that the representation $\widetilde{\pi}$:=

 $\iota_{\mathcal{G}}^{-1} \circ \pi \circ \iota_{\mathcal{F}} : C_{\mathbf{r}}^{*}(\mathcal{F}) \to M(C_{\mathbf{r}}^{*}(\mathcal{G}))$ satisfies $\widetilde{\pi}(f)g = (T \circ f)g$ for all $f \in \Gamma_{\mathbf{c}}(\mathcal{F})$, $g \in \Gamma_{\mathbf{c}}(\mathcal{G})$. We show that T also satisfies condition (iii) of Definition 7.8. Since π is strongly nondegenerate, $D = [\pi(C)D_{G^{0}}]$, that is, $C_{\mathbf{r}}^{*}(\mathcal{G}) = [\widetilde{\pi}(C_{\mathbf{r}}^{*}(\mathcal{F}))\Gamma_{0}(\mathcal{G}^{0})]$ and hence $\Gamma^{2}(\mathcal{G}, \lambda^{-1}) = [\widetilde{\pi}(C_{\mathbf{r}}^{*}(\mathcal{F}))\Gamma_{0}(\mathcal{G}^{0})]$. In particular, $\mathcal{G}_{x} = [T(\mathcal{F}_{x})\mathcal{G}_{s(x)}]$ for each $x \in G$ because $G_{s(x)}$ is discrete.

9.2. THE UNIT AND COUNIT OF THE ADJUNCTION. Denote by $\mathbf{Coact}_{C_r^*(G)}^{as}$ the category of very fine left-full coactions of $C_r^*(G)$ with all strongly nondegenerate morphisms. Then the functor $\check{\mathbf{F}} : \mathbf{Fell}_G^a \to \mathbf{Coact}_{C_r^*(G)}^a$ constructed in the preceding section actually takes values in $\mathbf{Coact}_{C_r^*(G)}^{as}$:

LEMMA 9.10. Let T be a morphism of admissible Fell bundles \mathcal{F}, \mathcal{G} on G. Then the morphism $\check{\mathbf{F}}T$ from $\check{\mathbf{F}}\mathcal{F}$ to $\check{\mathbf{F}}\mathcal{G}$ is strongly nondegenerate.

The proof is immediate from Proposition 7.9(ii).

The constructions in Proposition 9.1 and Proposition 9.9 yield a functor $\check{\mathbf{G}}$: $\mathbf{Coact}^{as}_{C^*_{\mathbf{r}}(G)} \rightarrow \mathbf{Fell}^a_G$. We now obtain an embedding as a full and coreflective subcategory $(\check{\mathbf{F}}, \check{\mathbf{G}}, \check{\eta}, \check{\varepsilon})$ of \mathbf{Fell}^a_G into $\mathbf{Coact}^{as}_{C^*_{\mathbf{r}}(G)}$.

PROPOSITION 9.11. Let \mathcal{F} be an admissible Fell bundle, $(\pi_{\mathcal{F}}(C_{r}^{*}(\mathcal{F}))_{K_{\mathcal{F}}}^{\gamma_{\mathcal{F}}}, \delta_{\mathcal{F}}) = \mathbf{\check{F}}\mathcal{F}$ the associated fine coaction, and $\mathcal{G} = \mathbf{\check{G}}\mathbf{\check{F}}\mathcal{F}$ and $\iota_{\mathcal{G}} : C_{r}^{*}(\mathcal{G}) \to \pi_{\mathcal{F}}(C_{r}^{*}(\mathcal{F}))$ the Fell bundle and the *-homomorphism associated to this coaction as above. Then there exists a unique isomorphism $\check{\eta}_{\mathcal{F}} : \mathcal{F} \to \mathcal{G}$ such that $\iota_{\mathcal{G}} \circ (\check{\eta}_{\mathcal{F}})_{*} = \pi_{\mathcal{F}}$.

Proof. Let $(C_K^{\gamma}, \delta) = (\pi_F(C_r^*(\mathcal{F}))_{K_F}^{\gamma_F}, \delta_F)$ and $U \in \mathfrak{G}$. We show that $C_U = \pi_F(\Gamma_0(\mathcal{F}|_U))$. Note that $[\{h_{\xi,\xi'}: \xi \in C_c(r(U)), \xi' \in C_c(U)\}] = C_0(U)$, where the functions $h_{\xi,\xi'}$ were defined in (8.1). Using Lemma 9.4, we can conclude

$$C_{U} = [\langle j(C_{c}(r(U))) |_{2} \delta(\pi_{\mathcal{F}}(\Gamma_{c}(\mathcal{F}))) | j(C_{c}(U)) \rangle_{2}]$$

By Lemma 8.12, we have for all $\xi \in C_c(r(U))$, $f \in \Gamma_c(\mathcal{F})$, $\xi' \in C_c(U)$,

$$\langle j(\xi)|_2 \delta(\pi_{\mathcal{F}}(f))|j(\xi')\rangle_2 = \pi_{\mathcal{F}}(fh_{\xi,\xi'}) \in \pi_{\mathcal{F}}(\Gamma_0(\mathcal{F}|_U)),$$

where $fh_{\xi,\xi'}$ denotes the pointwise product. Consequently, $C_U = \pi_{\mathcal{F}}(\Gamma_0(\mathcal{F}|_U))$. Since $U \in \mathfrak{G}$ was arbitrary, we can conclude that there exists an isomorphism $\check{\eta}_{\mathcal{F}} : \mathcal{F} \to \mathcal{G}$ of Banach bundles such that $\iota_{\mathcal{G}} \circ (\check{\eta}_{\mathcal{F}})_* = \pi_{\mathcal{F}} : \Gamma_c(\mathcal{F}) \to C$. Using the fact that $(\check{\eta}_{\mathcal{F}})_*$ is a *-homomorphism and that *G* is étale, one easily concludes that $\check{\eta}_{\mathcal{F}}$ is an isomorphism of Fell bundles.

PROPOSITION 9.12. Let (\mathcal{C}, δ) be a very fine coaction of $C_r^*(G)$, where $\mathcal{C} = C_{K'}^{\gamma}$ and let $\mathcal{F}, \iota : C_r^*(\mathcal{F}) \to C$ be the associated Fell bundle and *-isomorphism. Then there exists a unique strongly nondegenerate morphism $\check{\epsilon}_{(\mathcal{C},\delta)}$ from $(\pi_{\mathcal{F}}(C_r^*(\mathcal{F}))_{K_{\mathcal{F}}}^{\gamma_{\mathcal{F}}}, \delta_{\mathcal{F}})$ to (\mathcal{C}, δ) such that $\check{\epsilon}_{(\mathcal{C},\delta)} \circ \pi_{\mathcal{F}} = \iota$.

LEMMA 9.13. Let $U \in \mathfrak{G}$, $\xi \in C_{c}(U)$, $\eta \in \gamma$, and $\omega = |\eta\rangle_{1}j(\xi) \in \gamma \triangleright \alpha \subseteq \mathcal{L}(\mathfrak{K}, K_{\gamma \bigotimes_{h} \beta} H)$.

(i) There exists a $C_0(G^0)$ -weight $\phi : \Gamma_0(\mathcal{F}^0) \to C_0(G^0) \subseteq \mathcal{L}(\mathfrak{K}), f \mapsto \omega^* \delta(\iota(f)) \omega$. (ii) There exists a unique isometry $S_\omega : K_\phi = \Gamma^2(\mathcal{F}, \nu; \phi) \to K_\gamma \bigotimes_{\mathfrak{b}} H$ such that $S_\omega \widehat{j}_\phi(f) = \delta(\iota(f)) \omega$ for all $f \in \Gamma_c(\mathcal{F})$, and $S_\omega \pi_\phi(f) = \delta(\iota(f)) S_\omega$ for all $f \in \Gamma_c(\mathcal{F})$. (iii) $S_\omega j_\phi(\Gamma_c(\mathcal{F})) \subseteq \gamma \triangleright \alpha$.

Proof. (i) First, $\omega^* \delta(C_{G^0}) \omega \subseteq [\alpha^* \langle \gamma |_1 | \gamma \rangle_1 L(C_0(G^0)) \alpha] = [\alpha^* \langle \gamma |_1 | \gamma \rangle_1 \alpha] = C_0(G^0) \subseteq \mathcal{L}(\mathfrak{K})$. Second, observe that for all $c \in C_{G^0}$, $f \in C_0(G^0)$,

$$\begin{split} \phi(cf) &= j(\xi)^* \langle \eta |_1 \delta(cf) | \eta \rangle_1 j(\xi) \\ &= j(\xi)^* \langle \eta |_1 \delta(c) | \eta \rangle_1 r^*(f) j(\xi) = j(\xi)^* \langle \eta |_1 \delta(c) | \eta \rangle_1 j(\xi) f = \phi(c) f. \end{split}$$

(ii) As before, denote by $p_0 : \Gamma_f(\mathcal{F}) \to \Gamma_0(\mathcal{F}^0)$ the restriction. Let $U \in \mathfrak{G}$, $f, f' \in \Gamma_c(\mathcal{F})$, and $g = f^*f'$. Using the relation supp $h_{\xi,\xi} \subseteq G^0$ and Lemma 9.4, we find

$$\begin{split} \omega^* \delta(\iota(f))^* \delta(\iota(f')) \omega &= \eta^* \langle j(\xi) |_2 \delta(\iota(g)) | j(\xi) \rangle_2 \eta^* = \eta^* \iota(g \cdot h_{\xi,\xi}) \eta \\ &= \omega^* \delta(\iota(p_0(g))) \omega^* = \phi(p_0(g)) = \langle f | f' \rangle_{\Gamma^2(\mathcal{F}, \lambda^{-1}, \phi)}. \end{split}$$

The existence of S_{ω} follows. Finally, $S_{\omega}\pi_{\phi}(f) = \delta(\iota(f))S_{\omega}$ since $S_{\omega}\pi_{\phi}(f)\hat{j}_{\phi}(g) = S_{\omega}\hat{j}_{\phi}(fg) = \delta(\iota(fg))\omega = \delta(\iota(f))S_{\omega}\hat{j}_{\phi}(g)$ for all $f, g \in \Gamma_{c}(\mathcal{F})$.

(iii) Let $V \in \mathfrak{G}$, $f \in \Gamma_{c}(\mathcal{F}|_{V})$, $\zeta \in C_{c}(G^{0})$, and define $\zeta' \in L^{2}(G^{0}, \mu)$ by $\zeta'(s(x)) = \zeta(r(x))D^{1/2}(x)$ for all $x \in V$ and $\zeta'(y) = 0$ for all $y \in G^{0} \setminus s(V)$. Then $(j_{\phi}(f)\zeta)(x) = f(x)\zeta(r(x)) = (\hat{j}_{\phi}(f)\zeta')(x)$ for all $x \in G$ and therefore

$$S_{\omega}j_{\phi}(f)\zeta = S_{\omega}\hat{j}_{\phi}(f)\zeta' = \delta(\iota(f))\omega\zeta' = \delta(\iota(f))|\eta\rangle_{1}j(\xi)\zeta'.$$

Since $f \in \Gamma_{c}(\mathcal{F}|_{V})$, there exist $f' \in L(C_{0}(V))$, $\eta' \in \gamma$ such that $\delta(\iota(f))|\eta\rangle_{1} = |\eta'\rangle_{1}L(f')$. Now,

$$S_{\omega}j_{\phi}(f)\zeta = \delta(\iota(f))|\eta\rangle_1 j(\xi)\zeta' = |\eta'\rangle_1 L(f')j(\xi)\zeta' = |\eta'\rangle j(L(f')\xi)\zeta$$

because $(L(f')j(\xi)\zeta')(z) = 0$ for $z \notin VU$ and

$$(L(f')j(\xi)\zeta')(xy) = D^{-1/2}(x)f'(x)\xi(y)\zeta'(r(y)) = f'(x)\zeta(r(x))\xi(y) = j(L(f')\xi)\zeta(xy)$$

for all $(x, y) \in (V \times U) \cap G_s \times_r G$. Thus, $S_{\omega} j_{\phi}(f) \zeta \in \gamma \triangleright \alpha$. The claim follows.

Proof of Proposition 9.12. Since $\pi_{\mathcal{F}}$ is injective, we can define $\check{\epsilon} = \check{\epsilon}_{(\mathcal{C},\delta)} := \iota \circ \pi_{\mathcal{F}}^{-1}$. We show that $\delta \circ \check{\epsilon}$ is a morphism from $\pi_{\mathcal{F}}(C_{\mathbf{r}}^*(\mathcal{F}))_{K_{\mathcal{F}}}^{\gamma_{\mathcal{F}}}$ to $\delta(C)_{K_{\gamma} \otimes_{\beta}}^{\gamma \otimes_{\beta}} H$. For each $C_0(G^0)$ -weight ϕ on $\Gamma_0(\mathcal{F}^0)$, denote by $p_{\phi} : K_{\mathcal{F}} \to K_{\phi}$ the canonical projection. Let $\mathcal{S} \subseteq \mathcal{L}(K_{\mathcal{F}}, K_{\gamma} \otimes_{\beta} H)$ be the closed linear span of all operators of the form $S_{\omega} p_{\phi}$, where $U, \xi, \eta, \omega, \phi$ are as in the lemma above. Then $Sa = \delta(\check{\epsilon}(a))$ for each $S \in \mathcal{S}, a \in \pi_{\mathcal{F}}(C_{\mathbf{r}}^*(\mathcal{F}))$, and $[\mathcal{S}\gamma_{\mathcal{F}}] = [\delta(\iota(\Gamma_{\mathbf{c}}(\mathcal{F})))(\gamma \triangleright \alpha)] = \gamma \triangleright \alpha$. The claim follows. Since δ is an isomorphism from \mathcal{C} to $\delta(C)_{K_{\gamma} \otimes_{\beta} H}^{\gamma \otimes_{\beta} H}$, we can conclude that $\check{\epsilon}$

is a morphism from $\pi_{\mathcal{F}}(C^*_{\mathbf{r}}(\mathcal{F}))_{K_{\mathcal{F}}}^{\gamma_{\mathcal{F}}}$ to \mathcal{C} . The relation $(\check{\epsilon} * \mathrm{id}) \circ \delta = \delta \circ \check{\epsilon}$ follows from the fact that

$$\begin{aligned} \langle j(\xi)|_2 \delta(\check{\varepsilon}(g) \cdot f)|j(\xi')\rangle_2 &= \check{\varepsilon}(g) \cdot (fh_{\xi,\xi'}) \\ &= \check{\varepsilon}(g \cdot (fh_{\xi,\xi'})) = \check{\varepsilon}(\langle j(\xi)|_2 \delta(\pi_{\mathcal{F}}(g))|j(\xi')\rangle_2) \end{aligned}$$

for all $U \in \mathfrak{G}$, $g \in \Gamma_{c}(\mathcal{F}|_{U})$, $f \in C_{c}(U)$, $\xi, \xi' \in C_{c}(G)$ by Lemma 9.4.

COROLLARY 9.14. Every very fine coaction of $C^*_r(G)$ is left-full.

Proof. Let (\mathcal{C}, δ) be a very fine coaction of $C_r^*(G)$, let $(\pi_{\mathcal{F}}(C_r^*(\mathcal{F}))_{K_{\mathcal{F}}}^{\gamma_{\mathcal{F}}}, \delta_{\mathcal{F}})$ and $\check{\epsilon}_{(\mathcal{C},\delta)}$ be as above, and let $I := \{T \in \mathcal{L}_s((K_{\mathcal{F}}, \gamma_{\mathcal{F}}), (K, \gamma)) : Tx = \check{\epsilon}_{(\mathcal{C},\delta)}(x)T$ for all $x \in \pi_{\mathcal{F}}(C_r^*(\mathcal{F}))\}$. Then $\gamma = [I\gamma_{\mathcal{F}}]$ because $\check{\epsilon}_{(\mathcal{C},\delta)}$ is a morphism, and since $\delta_{\mathcal{F}}$ is left-full,

$$\begin{split} [\delta(C)|\gamma\rangle_1 C_r^*(G)] &= [(\check{\epsilon}_{(\mathcal{C},\delta)} * \mathrm{id})(\delta_{\mathcal{F}}(\pi_{\mathcal{F}}(C_r^*(\mathcal{F}))))(I \otimes_{\mathfrak{b}} \mathrm{id})|\gamma_{\mathcal{F}}\rangle_1 C_r^*(G)] \\ &= [(I \otimes_{\mathfrak{b}} \mathrm{id})\delta_{\mathcal{F}}(\pi_{\mathcal{F}}(C_r^*(\mathcal{F})))|\gamma_{\mathcal{F}}\rangle_1 C_r^*(G)] \\ &= [(I \otimes_{\mathfrak{b}} \mathrm{id})|\gamma_{\mathcal{F}}\rangle_1 C_r^*(G)] = [|\gamma\rangle_1 C_r^*(G)]. \quad \blacksquare$$

THEOREM 9.15. $(\check{\mathbf{F}}, \check{\mathbf{G}}, \check{\eta}, \check{\epsilon})$ is an embedding of \mathbf{Fell}_G^a into $\mathbf{Coact}_{C_r^*(G)}^{as}$ as a full and coreflective subcategory.

Proof. One easily verifies that $\check{\mathbf{G}}$ is faithful and that the families $(\check{\eta}_{\mathcal{F}})_{\mathcal{F}}$ and $(\check{\epsilon}_{(\mathcal{C},\delta)})_{(\mathcal{C},\delta)}$ are natural transformations as desired. Since $\check{\eta}$ is a natural isomorphism, $\check{\mathbf{F}}$ is full and faithful; see Theorem IV.3.1 in [18].

REFERENCES

- S. BAAJ, Représentation régulière du groupe quantique des déplacements de Woronowicz, Astérisque 232(1995), 11–48.
- [2] S. BAAJ, G. SKANDALIS, C*-algèbres de Hopf et théorie de Kasparov équivariante, K-Theory 2(1989), 683–721.
- [3] S. BAAJ, G. SKANDALIS, Unitaires multiplicatifs et dualité pour les produits croisés de C*-algèbres, Ann. Sci. École Norm. Sup. (4) 26(1993), 425–488.
- [4] É. BLANCHARD, Représentations de champs de C*-algèbres, C. R. Acad. Sci. Paris Sér. I Math. 314(1992), 911–914.
- [5] É. BLANCHARD, Tensor products of C(X)-algebras over C(X), Astérisque 232(1995), 81–92.
- [6] É. BLANCHARD, Déformations de C*-algèbres de Hopf, Bull. Soc. Math. France 124(1996), 141–215.
- [7] G. BÖHM, Galois theory for Hopf algebroids, Ann. Univ. Ferrara Sez. VII (N.S.) 51(2005), 233–262.

- [8] M.J. DUPRÉ, R.M. GILLETTE, Banach Bundles, Banach Modules and Automorphisms of C*-algebras, Res. Notes Math., vol. 92, Pitman (Adv. Publ. Program), Boston, MA 1983.
- [9] S. ECHTERHOFF, S. KALISZEWSKI, J. QUIGG, I. RAEBURN, A categorical approach to imprimitivity theorems for C*-dynamical systems, *Mem. Amer. Math. Soc.* 850(2006), 180.
- [10] M. ENOCK, Measured quantum groupoids in action, Mém. Soc. Math. France (N.S.) 114(2008).
- [11] M. ENOCK, J.-M. VALLIN, Inclusions of von Neumann algebras and quantum groupoids. II, J. Funct. Anal. 1(2000), 156–225.
- [12] J.M.G. FELL, R.S. DORAN, Representations of *-Algebras, Locally Compact Groups, and Banach *-Algebraic Bundles. Vol. 2. Banach *-Algebraic Bundles, Induced Representations, and the Generalized Mackey Analysis, Pure Appl. Math., vol. 126, Academic Press, Inc., Boston, MA 1988.
- [13] L. KADISON, Hopf algebroids and H-separable extensions, Proc. Amer. Math. Soc. 131(2003), 2993–3002.
- [14] G.G. KASPAROV, Equivariant KK-theory and the Novikov conjecture, *Invent. Math.* 91(1988), 147–201.
- [15] A. KUMJIAN, Fell bundles over groupoids, Proc. Amer. Math. Soc. 126(1998), 1115– 1125.
- [16] E.C. LANCE, Hilbert C*-Modules. A Toolkit for Operator Algebraists, London Math. Soc. Lecture Note Ser., vol. 210, Cambridge Univ. Press, Cambridge 1995.
- [17] P.-Y. LE GALL, Théorie de Kasparov équivariante et groupoïdes. I, K-Theory 16(1999), 361-390.
- [18] S. MACLANE, Categories for the Working Mathematician, second ed., Grad. Texts in Math., vol. 5, Springer-Verlag, New York 1998.
- [19] P.S. MUHLY, D.P. WILLIAMS, Equivalence and disintegration theorems for Fell bundles and their C*-algebras, *Dissertationes Math. (Rozprawy Mat.)* 456(2008), 1–57.
- [20] A.L.T. PATERSON, *Groupoids, Inverse Semigroups, and their Operator Algebras,* Progress in Math., vol. 170, Birkhäuser Boston, Inc., Boston, MA 1999.
- [21] V. PAULSEN, *Completely Bounded Maps and Operator Algebras*, Cambridge Stud. Adv. Math., vol. 78, Cambridge Univ. Press, Cambridge 2002.
- [22] J.C. QUIGG, Discrete C*-coactions and C*-algebraic bundles, J. Austral. Math. Soc. Ser. A 60(1996), 204–221.
- [23] J. RENAULT, A groupoid approach to C*-algebras, Lecture Notes in Math., vol. 793, Springer, Berlin 1980.
- [24] K. SZLACHÁNYI, Weak Hopf algebras, in Operator Algebras and Quantum Field Theory (Rome, 1996), Cambridge Univ. Press, Cambridge, MA 1997, pp. 621–632.
- [25] K. SZLACHÁNYI, Galois actions by finite quantum groupoids, in *Locally Compact Quantum Groups and Groupoids (Strasbourg, 2002)*, IRMA Lect. Math. Theor. Phys., vol. 2, de Gruyter, Berlin 2003, pp. 105–125.

- [26] T. TIMMERMANN, A definition of compact C*-quantum groupoids, in Operator Structures and Dynamical Systems (Leiden, 2007), Amer. Math. Soc., Providence, RI 2009, pp. 267–289.
- [27] T. TIMMERMANN, C*-pseudo-multiplicative unitaries, Hopf C*-bimodules, and their Fourier algebras, *J. Inst. Math. Jussieu* **11**(2010), 189–220.
- [28] T. TIMMERMANN, The relative tensor product and a fiber product in the setting of *C**-algebras, *J. Operator Theory*, to appear; arXiv:0907.4846.
- [29] T. TIMMERMANN, *C**-pseudo-multiplicative unitaries and Hopf *C**-bimodules, preprint, arXiv:0908.1850.
- [30] J.-M. VALLIN, Actions and coactions of finite quantum groupoids on von Neumann algebras, extensions of the matched pair procedure, *J. Algebra* **314**(2007), 789–816.

THOMAS TIMMERMANN, FB 10 MATHEMATISCHES INSTITUT, WESTFÄLISCHE WILHELMS-UNIVERSITÄT, EINSTEINSTRASSE 62, 48149 MÜNSTER, GERMANY *E-mail address*: timmermt@uni-muenster.de

Received November 3, 2009.